

The Child's Right to Representation of Article 12 UNCRC in Family Law Proceedings:

A comparison and evaluation of the legal frameworks in Australia, France, the Netherlands and South Africa



Charlotte Mol – 5505496 Legal Research Master – Master Thesis First supervisor: Dr. M. van den Brink Second supervisor: Prof. dr. W. Schrama 19 March 2017

Table of Contents

Table of	Contents	1
List of F	igures and Tables	5
List of A	bbreviations	6
1. Inti	roduction	7
1.1.	Research Questions	11
1.2.	Methodology	15
1.2.	1. Choice of Jurisdictions	
1.2.	2. Sources	17
1.3.	Terminology	19
2. Chi	ldren's right to representation in the UN Convention on the Rights of the Chi	ild 22
2.1.	General remarks on Article 12 UNCRC	
2.2.	Understanding Article 12 UNCRC	
2.2.	1. Article 12 paragraph 1 UNCRC	
2.2.		
2.2.	3. How to implement Article 12 UNCRC	
2.3.	Article 12 and its relationship to other UNCRC articles	
2.3.	1. Article 2 and Article 12 UNCRC	
2.3.	2. Article 3 and Article 12 UNCRC	
2.3.	3. Article 5 and Article 12 UNCRC	
2.3.	4. Articles 13 and 17 and Article 12 UNCRC	
2.3.	5. Other Articles and Article 12 UNCRC	
2.4.	Summary	
3. Cor	nparison and evaluation of the legal framework of child representation	39
3.1.	Forms of representation	40
3.1.	1. Historical development	
3.1.	2. Sources of law	44
3.1.	3. Relation to other manners of hearing the child's views	

3.3.2.	Conflict related requirements	56
3.3.3.	Combined requirements for representation	58
3.4. App	ointment of a child representative	60
3.4.1.	Who appoints the child representative	60
3.4.2.	Duration of child representation	61
3.5. The	function, work and payment of the child's representative	63
3.5.1.	Function requirements	
3.5.2.	The work of the child representatives	65
3.5.3.	Payment of the child representatives	69
3.6. Sum	mary and combined evaluation	

4.	Conclusion	7:	5
----	------------	----	---

bliography	
Primary sources	
Case Law	
Australia	
France	
The Netherlands	
South Africa	
Literature	
Other sources	

5.	Annexes.	
5.	1. Aust	ralia (New South Wales)
	5.1.1.	What forms of representation are available for children in family law proceedings in
	Australia	New South Wales) and how are they regulated?
	5.1.1.1.	When were the forms of representation introduced or amended?100
	5.1.1.2.	In which types of cases can they be represented?
	5.1.1.3.	When can children be represented in family law proceedings in Australia (New South
	Wales)?	
	5.1.1.4.	What requirements are set for the children, e.g. age, level of maturity?103
	5.1.1.5.	What other requirements are applied, e.g. conflict of views between child and parents? .105
	5.1.1.6.	Who decides whether the child will be represented?
	5.1.1.7.	How is the child's representative financed?107

5.	.1.2.	What is the task of the child's representative in family law proceedings in	Australia
()	New Sou	th Wales)?	108
	5.1.2.1.	What are the function requirements for the child's representative?	
	5.1.2.2.	How should the child's representative complete their task?	
	5.1.2.3.	Is the child representative the only option to be heard, are other options availab	le and can
	they be	complementary?	
5.2.	Fra	nce	114
5.	.2.1.	What forms of representation are available for children in family law proce	edings in
F	rance an	d how are they regulated?	114
	5.2.1.1.	When were the forms of representation introduced or amended?	114
	5.2.1.2.	In which types of cases can they be represented?	115
	5.2.1.3.	When can children be represented in family law proceedings in France?	116
	5.2.1.4.	What requirements are set for the children, e.g. age, level of maturity?	117
	5.2.1.5.	What other requirements are applied, e.g. conflict of views between child and par	ents? .117
	5.2.1.6.	Who decides whether the child will be represented?	
	5.2.1.7.	How is the child's representative financed?	
5.	.2.2.	What is the task of the child's representative in family law proceedings in Frank	ance?119
	5.2.2.1.	What are the function requirements for the child's representative?	119
	5.2.2.2.	How should the child's representative complete their task?	
	5.2.2.3.	Is the child representative the only option to be heard, are other options availab	le and can
	they be	complementary?	
5.3.	The	Netherlands	123
5.	.3.1.	What forms of representation are available for children in family law proce	edings in
th	e Nethe	rlands and how are they regulated?	123
	5.3.1.1.	When were the forms of representation introduced or amended?	
	5.3.1.2.	In which types of cases can they be represented?	
	5.3.1.3.	When can children be represented in family law proceedings in the Netherlands?.	
	5.3.1.4.	What requirements are set for the children, e.g. age, level of maturity?	
	5.3.1.5.	What other requirements are applied, e.g. conflict of views between child and par	ents? .127
	5.3.1.6.	Who decides whether the child will be represented?	
	5.3.1.7.	How is the child's representative financed?	
5.	.3.2.	What is the task of the child's representative in family law proceeding	gs in the
Ν	etherlan	ds?	130
	5.3.2.1.	What are the function requirements for the child's representative?	
	5.3.2.2.	How should the child's representative complete their task?	
	5.3.2.3.	Is the child representative the only option to be heard, are other options availab	le and can
	they be	complementary?	
5.4.	Sou	th Africa	135

5.4.1.	What forms of representation are available for children in family law pro-	ceedings in
South Afri	ica and how are they regulated?	
5.4.1.1.	When were the forms of representation introduced or amended?	135
5.4.1.2.	In which types of cases can they be represented?	136
5.4.1.3.	When can children be represented in family law proceedings in South Africa? .	138
5.4.1.4.	What requirements are set for the children, e.g. age, level of maturity?	138
5.4.1.5.	What other requirements are applied, e.g. conflict of views between child and p	arents? .139
5.4.1.6.	Who decides whether the child will be represented?	140
5.4.1.7.	How is the child's representative financed?	141
5.4.2.	What is the task of the child's representative in family law proceeding	gs in South
Africa?		
5.4.2.1.	What are the function requirements for the child's representative?	143
5.4.2.2.	How should the child's representative complete their task?	144
5.4.2.3.	Is the child representative the only option to be heard, are other options avail	able and can
they be o	complementary?	144

List of Figures and Tables

Figure 1: Timeline of the introduction of representation forms in the jurisdictions43
Figure 2: Overall compliance with the UNCRC of representation frameworks in the four
jurisdictions75
Table 1: Final selection of jurisdictions for comparison 17
Table 2: The forms of representation and their general task per jurisdiction
Table 3: Overview of complementarity of forms by which children can be heard in each of the
jurisdictions
Table 4: Forms of representation available in different categories of proceedings in each jurisdiction
per type of representation
Table 5: The role of child related requirements for the representation forms in the different
jurisdictions*
Table 6: Requirements for the forms of representation per jurisdiction
Table 7: Decision maker for each form of representation per jurisdiction 60
Table 8: Function requirements of which type of practitioners and their registration in relation to the
forms of representation
Table 9: Compliance of the legal representation frameworks in four jurisdictions with Article 12
UNCRC

List of Abbreviations

AFLA	Australian Family Law Act 1975
AHA	Ad hoc administrator [France; administrateur ad hoc]
Art.	Article
AU	Australia
CAL	Curator ad litem [South Africa]
CCPA (NSW)	Children and Young Persons (Care and Protection) Act 1998 (NSW)
CL	Children's lawyer [France; avocat d'enfant]
DCC	Dutch Civil Code [Burgerlijk Wetboek]
DCCP	Dutch Code of Civil Procedure [Wetboek van Burgerlijke Rechtsvordering]
DLR	Direct legal representative [New South Wales, Australia]
FA	Family advocate
FamR	Family report
FCC	French Civil Code [Code Civil]
FCCP	French Code of Civil Procedure [Code de procédure civile]
FCPC	French Code of Criminal Procedure [Code de procédure pénale]
FR	France
FGAL	Filiation guardian ad litem [The Netherlands; bijzondere curator ex. Art. 1:212
	DCC]
GGAL	General guardian ad litem [The Netherlands; bijzondere curator ex. Art. 1:250
	DCC]
GAL	Guardian ad litem [New South Wales, Australia]
ICL	Independent children's lawyer [Australia]
ILR	Independent legal representative [New South Wales, Australia]
JM	Judicial Meetings
LOVF	Landelijk Overleg Vakinhoud Familie- en Jeugdrecht (the National Consultations
	on Family and Child Law matters)
LR	Legal representative [South Africa]
NL	The Netherlands
NSW	New South Wales, Australia
S.	Section
SA	South Africa
SACA	South African Children's Act 2005
UK	United Kingdom
UNCRC	United Nations Convention on the Rights of the Child
WR	Welfare report

1. Introduction

'Children should be seen and not heard' – Victorian proverb

In Victorian times, when children were seen as property, the maxim that children should be silent was promoted.¹ In modern times this maxim has been turned on its head, by ending children's silence and stimulating – if not requiring – adults to hear children. While children have gradually been recognized as legal subjects, not objects, Article 12 of the United Nations Convention on the Rights of the Child (UNCRC) has served as the real catalyst for change regarding children's right to be heard.² Article 12 determines that children should be provided the opportunity to express their views in all matters concerning them and that these views should be given due weight. More specifically, in paragraph 2 of the article, the child's right to be heard, whether directly or through a representative, in any judicial and administrative proceedings affecting the child is safeguarded.

Family law proceedings are the most common judicial proceedings concerning children. Family law proceedings concern, for example, custody following the separation of parents, matters of filiation, or child protection cases. They are not only the most common judicial proceedings concerning children, but they also have a profound impact on a child's life. As emphasized by Baroness Hale, Justice of the UK Supreme Court, in many family law proceedings it is vital that courts do not lose sight of the fact that it is ultimately the child's future that is being decided upon.³ When a child and its future is at the center of proceedings, it is necessary to take into account his or her views. Especially because existing psychological research has revealed the developmental and practical benefits of letting children participate in family law proceedings.⁴

The question is, how can or should children participate in family law proceedings? The UNCRC leaves a great deal of discretion as to how the right to participation should be given shape nationally. However, the Committee on the Rights of the Child has given some guidance. Amongst others, the child should be able to choose how he or she wants to be heard, directly or through a representative, if at all.⁵ With regards to representatives, the Committee notes that it can be the parent(s), a lawyer or another person. However, there is often the risk that there is a conflict between the child and the most obvious representative, the parent(s), and thus another form of representation might be required.⁶ According to the Committee the representative should have sufficient knowledge and experience in working with children in legal proceedings and should exclusively represent the interests of the child and not those of other persons.⁷ Besides these remarks, it remains open to national discretion how representation for children in judicial proceedings is set up. While children

¹ Fortin 2005, p. 11.

² Parkes 2013, p. 91.

³ Hale 2006 as cited by Parkes 2013, p. 90.

⁴ See, for example, Ballard et al. 2014; and Parkinson & Cashmore 2008.

⁵ UNCRC General Comment No. 12 2009, para. 35.

⁶ UNCRC General Comment No. 12 2009, para. 36.

⁷ UNCRC General Comment No. 12 2009, para. 36 and 37.

have the right to be heard, they remain dependent on adults, as parents, judges and lawmakers, as to whether and how they can be heard or represented.⁸

Children's right to representation

The topic of this research is children's right to representation in family law proceedings. Representation is one of the forms of child participation. In the following, the choice for the topic of child representation will be explained.

Children's right to be heard and to have a representative has become widely accepted, in part due to the UNCRC, but also on the regional level the child's right to participate has been gaining traction.9 Therefore, the discussion on how children should be able to participate in family law proceedings is thriving around the world. Because 'there is a world of difference' between having the child speak to the judge individually, having the child's views represented by a third party, or having the child represented by a lawyer directly, a core question concerns the forms of participation.¹⁰ In legal proceedings children can participate in a variety of ways, for example by speaking with the judge directly in a judicial meeting. In general, participation in family law proceedings can very roughly be separated into four forms: children's litigation on their own behalf, judicial meetings, bestinterests representation, and separate legal representation.¹¹

This research focuses on the latter two representation forms of participation. There are four main reasons why the representation forms are the most interesting forms of participation to study. The first two reasons concern the UNCRC. As discussed above, Article 12 explicitly mentions representation as a means for the child to express his or her views. Representation can be a very suitable means because it can mitigate the fears of placing children in a burdensome position in family law proceedings.¹² Secondly, it also plays into the UNCRC's aim of having 'an equilibrium between respecting children as human persons and not abandoning them to their autonomy at too early a stage'.¹³ Children are not yet adults, they are still gradually acquiring autonomy, capabilities and competences and must therefore to some extent be protected. The third reason to focus on representation is because existing research in social sciences has shown the beneficial effects of providing children with a representative.¹⁴ In proceedings where a child's life is profoundly affected,

⁸ Daly 2016, p. 2; and Lundy 2007, p. 920.

⁹ On the African regional level: African Charter on the Rights and Welfare of the Child (ACRWC), adopted July 1990 (entered into force 29 November 1999) OAU Doc.CAB/LEG/24.9/49 (1990). On the European level: European Convention on the Exercise of Children's Rights 1996, ETS No. 160 and the Guidelines of the Committee of Ministers of the Council of Europe on child friendly justice (Adopted by the Committee of Ministers on 17 November 2010 at the 1098th meeting of the Ministers' Deputies), CM/Del/Dec(2010)1098/10.2abc-app6. On the EU level: Art. 24 of the Charter of Fundamental Rights of the European Union. ¹⁰ Sutherland 2012, p. 28.

¹¹ These are the generally sketched four levels of participation as can be found in literature. See, amongst others, Daly 2016; Fernando 2013; Koh Peters 2005; Parkes 2013; and Parkinson & Cashmore 2008.

¹² Sutherland 2012, p. 28. ¹³ Lücker-Babel 1995, p. 404.

¹⁴ See, for example, Ballard et al. 2014; Birnbaum & Bala 2009; Cashmore 2011; Tisdall et al. 2004; and Tisdall & Morrison 2012

a child's representative can function as their friend or their champion.¹⁵ The representative can actively ensure that the parents and judge are aware of and understand the child's view. Finally, the choice to focus on representation is made because consensus on the topic is still far-sought. The discussion regarding whether or not to have children's representatives and if so, what the form, function, tasks and other specifics should be, is taking place in the Netherlands,¹⁶ but is also the subject of lively debate in many other jurisdictions.¹⁷

Aim of this research

This exploratory research aims to contribute to the current academic and political debate through a comparative legal overview and evaluation in light of Article 12 UNCRC. As the evaluative framework, Article 12 of the UNCRC forms a core aspect of this research. The UNCRC is one the many international instruments which does not aim for the harmonization of law, but promotes and imposes common human rights standards.¹⁸ These human rights standards ought to be complied with in all areas of national law. As Krause remarked, because UN treaties recognize individual rights in abstract terms, they have a 'significant impact in encouraging commonalities in family law principles across national boundaries'.¹⁹ More specifically, the ratification of the UNCRC starting in the 1990s stimulated the creation and implementation of formal mechanisms for children's representation in many jurisdictions.²⁰ Therefore, this research departs from the hypothesis that the child's right to be heard, according to Article 12 UNCRC also through a representative, has led to a convergence in the national family and procedural laws of jurisdictions in providing children with a representative in family law proceedings.²¹

However, the research also departs from the understanding that Article 12 UNCRC leaves some leeway as to how States Parties implement the international human rights standard.²² The content of the rights contained in the UNCRC are minimum human rights standards. Minimum in the sense that States may provide more advantageous rights to children. States are given a margin of discretion in how they implement these rights, but the minimum standard must be provided. Therefore, this research aims to see to what extent the legal framework in four jurisdictions, all similarities and differences included, comply with the minimum standard provided by Article 12 UNCRC and to study how the jurisdictions have colored in the (relatively) open framework.

¹⁵ Hale 2011, p. 3.

¹⁶ See, for example, De Graaf & Limbeek 2011; Kentie & Hendriks 2013; Pieters 2012; and van Teeffelen 2008.

¹⁷ See, for example, Atwood 2011; Bala, Birnbaum & Bertrand 2013; Elrod 2007; and Tisdall et al. 2004.

¹⁸ Sutherland 2012, p. 10.

 ¹⁹ Krause 2006, p. 1099.
 ²⁰ Bilson & White 2005, p. 222.

²¹ This hypothesis is in line with the trend that Nicola describes: 'With the rise of international human rights conventions addressing the family [...], lawyers are increasingly addressing the convergence of family law regimes through the language of universal and individual rights.' See, Nicola 2010, p. 805. ²² Parkes 2013, p. 255.

This research also aims to conduct novel research to add to the existing body of knowledge on this topic. There is a clear 'Anglocentric focus in much of the international literature'²³ as most comparative articles focus on common law jurisdictions, especially the United Kingdom, Australia, New Zealand, Canada and the United States. This research will include civil law and mixed jurisdictions, to see what similarities and differences exist and whether common law, civil law and mixed jurisdictions can learn from one another.

Besides expanding the geographical scope, this research also expands the scope with regards to the areas of family law proceedings. Existing comparative work is more limited, focusing only on the representation of children in specific types of family law proceedings, for example in custody cases or child protection cases.²⁴ This research has a broader scope, including all family law proceedings. While 'what should be considered as 'family law' is highly debated',²⁵ in this research 'family law proceedings' are understood to include proceedings on horizontal and vertical family relations and both proceedings of a 'private' as well as 'public' nature, as the distinction between private and public law varies per jurisdiction. More specifically, as the topic concerns contentious family law proceedings concerning children, these are proceedings on the matters of: custody, parental authority, contact and access, maintenance, parentage, adoption, child protection, (temporary or alternative) care, and international child abduction. Succession law and property law relating to children are left out of the picture. An all-round understanding of how children are represented in all sorts of family law proceedings allows for a combined evaluation as to whether the standard in Article 12 UNCRC is complied with.

²³ Bilson & White 2005, p. 223.

 ²⁴ For example, an international comparison only for public law cases: Bilson & White 2005; or the comparison only for custody cases following divorce: Rešetar & Emery 2008.
 ²⁵ Scherpe 2016, p. ix.

1.1. Research Questions

The main research question to be answered is: To what extent is the minimum international human rights law standard for child representation provided by Article 12 UNCRC complied with by the legal framework for family law proceedings in four jurisdictions?

To answer the main research question the following five sub-questions have been formulated:

1. What is the protected minimum standard of the child's right to representation in family law proceedings in the UN Convention on the Rights of the Child?

The first sub-question addresses the international human rights framework in the UN Convention on the Rights of the Child. The sub-question outlines the evaluative criteria for the main research question. The thesis will commence with a chapter devoted to this sub-question, addressing what Article 12 UNCRC provides for, how it should be understood and what its relationship is to other rights protected in the UNCRC.

This sub-question will primarily be answered using primary sources, such as the Convention text, the General Comments of the Committee on the Rights of the Child, and the Reports of the Working Group when drafting the Convention. The primary sources will be supplemented by secondary sources, e.g. legal explanatory texts on the UNCRC.

- 2. What forms of representation are available for children in family law proceedings in the different jurisdictions and how are they regulated?
- 3. What is the task of the child's representative in family law proceedings in the different jurisdictions?

The second and third sub-questions divide the subject of child representation into smaller parts to allow for a uniform comparison, this is also known as the Cartesian method.²⁶ The extended answers to these sub-questions, which form the basis for the comparative chapter, will be included per jurisdiction in the annexes to allow for a more compact comparison.

The two sub-questions concern the manner in which states regulate the representation of children in family law proceedings. The questions focus on the "five Ws" regarding child representation: who, what, when, where and why? Both sub-questions cover the main points of discussion in academic and political debates.²⁷

²⁶ Boele-Woelki 2008, p. 7.

²⁷ See for example, Australian Law Reform Commission 1997; Bilson & White 2005; Daly 2016, p. 5-6; and Pathways for Children 2004.

2.	Wh	at forms of representation are available for children in family law proceedings in the		
 	diff	erent jurisdictions and how are they regulated?		
 	a. When were the forms of representation introduced or amended?			
	b.	In which types of cases can children be represented in family law proceedings in the		
		different jurisdictions?		
ļ	c.	When can children be represented in family law proceedings?		
- 	d.	What requirements are set for the children, e.g. age, level of maturity?		
 	e.	What other requirements are applied, e.g. conflict of views between child and parents?		
 	f.	Who decides whether the child will be represented?		
	g.	How is the child's representative financed?		
		·		

In sub-question two the available forms of representation and the manner in which they are regulated are studied. Question 2(a) looks at the historical development of the forms of representation. This subquestion informs the comparison and analysis, to allow for an explanation of differences and similarities and to determine the impact of the UNCRC in domestic laws. Question 2(b) is the elaboration of the decision not to limit the scope of the research to a specific type of family law proceeding. As previously discussed, this decision was made because comparative research on how the forms of representation are available in different types of family law proceedings is lacking. When discussing and evaluating the ways in which children can be represented in family law proceedings it is important to research the complete picture, to be able to determine whether differences are made between types of family law proceedings, and if so, why these differences have been made and whether it is reasonable to do so in relation to Article 12 UNCRC. Questions 2(c), (d), (e), (f), and (g) are questions which are often raised in the academic debates.²⁸

3. What is the task of the child's representative in family law proceedings in the different jurisdictions?

- a. What are the function requirements for the child's representative?
- b. How should the child's representative complete their task?
- c. Is the child representative the only option to be heard, are other options available and can they be complementary?

_ _ _ _ _ _ _ _ _ _ _

Sub-question three goes into the task of the different children's representatives, also looking at the function requirements of the child representative and how they are to execute their task. What the task of the child representative is or ought to be is one of the main points of discussion in the academic

²⁸ See, for example, Bilson & White 2005; Daly 2016, p. 5-6; Ross 2013b, p. 333; and van Teeffelen 2008.

debate and therefore requires a sub-question.²⁹ Question 3(c) is included to mitigate the decision to limit the scope of the research to representation of children, as described above, for four main reasons. The additional, more practical, reason is that it would be too extensive to map out all the different ways in which children can be heard in four jurisdictions. As this limitation creates the risk of overlooking the grey area of 'almost' representation, question 3(c) not only addresses whether the forms of representation can be complementary, but also briefly what the other options for children to be heard are in the four jurisdictions.

Both sub-questions will be answered through the use of primary sources of law, including legislation and – for the common law and mixed jurisdictions – leading case-law, primary sources of legislative history for sub-question 2(a), as well as primary sources such as guidelines on legal aid or the functioning of the representative. This will be combined with secondary sources to ensure a correct understanding of the foreign law.

4. What are the similarities and differences between the jurisdictions with regards to subquestions 2 and 3 and how can they be explained?

Sub-question four is the comparison, bringing together all the information collected through subquestions two and three. The comparison will form the third chapter of this research. The comparison will be presented simultaneously per theme and the similarities and differences between jurisdictions will be discussed. At the same time an effort will be made to provide an explanation for these similarities and differences, by also looking at the historical developments in the jurisdictions and the effects of the UNCRC.

5. To what extent do the legal frameworks of the forms of representation in the four jurisdictions comply with the minimum international human rights law standard provided in Article 12 UNCRC?

The fifth sub-question analyses the comparison made in sub-question four and evaluates to what extent the differences and similarities in regulating child representation comply with the minimum human rights standard in Article 12 UNCRC and how the leeway afforded by the standard is filled in. This evaluation will be included within the third chapter, combining the comparison and the evaluation to avoid unnecessary repetition. The third chapter will be followed by the fourth chapter consisting of a summary and conclusion. The main sources for this evaluation will be what has previously been covered in the discussion on Article 12 UNCRC and in the comparison. In addition,

²⁹ See, for example, Atwood 2011; Bala, Birnbaum & Bertrand 2013; Bilson & White 2005; Elrod 2007; Strutz & Verhagen 2015; and Taylor et al. 2012.

some other sources will also be referred to, such as the concluding observations of the Committee on the Rights of the Child to the periodic reports of the included jurisdictions and secondary sources.

1.2. Methodology

The goal of this research, to compare the ways in which a children's representation in family law proceedings can be given form and to evaluate the conformity thereof with the norms of Article 12 UNCRC, denotes the use of the comparative legal method. According to Glenn, comparative law can have four aims: (i) the general scientific goals of increasing knowledge and better understanding law, (ii) the evolutionary aim of establishing a taxonomy, and finally the pragmatic and utilitarian aims of (iii) contributing to better knowledge of the national law and thus to law reform and (iv) the regional or international harmonization of law.³⁰ The aim of this research falls within three of these categories. Firstly, this research aims to increase knowledge in a field which has been dominated by international literature on the common law jurisdictions by taking into account a wide range of jurisdictions. Secondly, this research can contribute to law reform in various jurisdictions as it looks at how other jurisdictions have addressed child representation. According to Sutherland, this aim of comparative research 'provides an opportunity to be critically selective' as to the methods used in foreign jurisdictions.³¹ This research aims to provide suggestions for national law reform in light of the leeway provided by the human rights standard of Article 12 UNCRC. However, it cannot provide empirically tested suggestions for law reform, as that would require further research. Finally, while this research does not formally aim to harmonize the law, it does aim to examine the effect of international human rights law in how jurisdictions have given form to the child's right to be heard through a representative in family law proceedings and whether this has led to convergence amongst these jurisdictions. The research will also look at whether this convergence can further help clarify the human rights standard in Article 12 UNCRC or if the differences require further guidance by the Committee on the Rights of the Child.

Within the comparative law methodology, a plurality of approaches exists.³² The so-called functional-institutional approach will be used in this research. The functional-institutional approach, introduced by Örücü, answers the question 'Which institution in system B performs an equivalent function to the one under survey in system A?³³ In this approach the legal institution forms the starting point of the comparison, instead of, for example, a specific problem in the problem-solving approach.³⁴ The functional-institutional approach fits best with the goal of this research as the child's representative, in the broad sense, is the institution that will be compared in all the chosen jurisdictions.

Due to the choice for the functional-institutional method, the tertium comparationis, the object to be compared, is the institution with the similar function.³⁵ Therefore, the *tertium comparationis* will be the legal institutions which function as a form of child representation in family

³⁰ Glenn 2012.

³¹ Sutherland 2012, p. 2. ³² Oderkerk 2015, p. 590-591.

 ³³ Örücü 2007, p. 33.
 ³⁴ Oderkerk 2015, p. 612.

³⁵ Oderkerk 2015, p. 610.

law proceedings. With this *tertium comparationis*, the research will be a micro-level comparison as it focuses on one form of legal institution within the legal systems.³⁶

The object to be compared, the 'child representatives', need not be named the same in the different jurisdictions. Firstly, due to the issue of translation³⁷ and secondly, because there is no universally accepted definition of what constitutes a child representative or guardian ad litem³⁸ and there are many different terms used to refer to similar institutions, e.g. independent children's lawyer.³⁹

1.2.1. Choice of Jurisdictions

Perhaps the most important decision made within comparative research is the choice of jurisdictions to be compared. In principle, as a starting point all jurisdictions in the world are eligible for selection.⁴⁰ A selection must be based on the aim and topic of the research and must be in line with the methodological choices with regards to the approach and the *tertium comparationis*.⁴¹ The aim of evaluating the conformity of child representation forms with Article 12 UNCRC already excludes one country, the United States, as it is the only country in the world that has not ratified the UNCRC.⁴² The functional-institutional approach and the *tertium comparationis* in this research require the jurisdictions selected to have a form of child representatives in family law proceedings. Thus, for the selection only jurisdictions with at least one form of child representative have been chosen.⁴³ The form of child representative and the family law proceedings in which they operate have been taken in the broadest sense of the words.

The second selection guideline was to include jurisdictions from different legal-technical traditions,⁴⁴ to add to the existing body of research by shifting away from a complete 'Anglocentric focus'. At first the selection of jurisdictions was more extensive: Canada, Chile, England & Wales and Scotland were also included. However, due to time constraints and the (in part, unforeseen) complexity of the topic a more restricted selection had to be made. The choice for Australia as the common law jurisdiction was made because it is known for its well-developed child participation regime.⁴⁵ The final selection includes one common law jurisdiction, two civil law jurisdictions and one jurisdiction with a mixed legal system. The selection was also made to include jurisdictions from different parts of the world, including two European jurisdictions, one African jurisdiction, and one

³⁶ Örücü 2007, p. 31.

³⁷ Oderkerk 2015, p. 616.

³⁸ Bilson & White 2005, p. 223; and Parkes 2013, p. 99.

³⁹ Parkinson & Cashmore 2008, p. 49.

⁴⁰ Oderkerk 2001, p. 295-298.

⁴¹ Oderkerk 2015, p. 603-605.

⁴² The United States of America is the only nation in the world that has not ratified the UNCRC, both Somalia and South-Sudan, have ratified the Convention in 2015, making the UNCRC an almost universal treaty.

⁴³ As a starting point the international research of Koh Peters (2005) was used to determine which jurisdictions had representation for children in family law proceedings.

⁴⁴ De Boer 1992, p. 43.

⁴⁵ Barrie 2013; and Parkinson & Cashmore 2008.

Oceanian jurisdiction. Due to the personal language skills of the researcher and the available time and sources, the selection does not include any Asian, South American, North-American or Middle-Eastern countries.

The final selection thus includes four jurisdictions with forms of child representatives – see Table 1 below. With regards to one of these jurisdictions, Australia, a specific state has been selected. This selection is required as Australia is a federation, with a division of power between the federal legislature and the state and territory legislatures. In Australia, family law is federal law, so for a large part this research concerns all the states and territories.⁴⁶ However, each state and territory has its own child protection and adoption legislation, although they all provide for the child's view to be heard.⁴⁷ Therefore, the state jurisdiction of New South Wales, the most populous state, will be taken as the example. South Africa, the Netherlands, and France are unitary states. This means that although each of these countries have local governments in their respective provinces, departments and/or municipalities, they are not federations. Thus, in all three jurisdictions family law is a matter for the national legislature.

TT 1 1 T 1 T 1	1	C · · 1 ·	· c	
Table 1: Final	selection	of iurisdich	ons for	comparison
1 4010 1. 1 11141	serection	of ful isaicii	ions joi	companison

Jurisdiction	Type of legal system	Form of Child Representation
Australia	Common law	Federal: Independent children's lawyer
(New South Wales)		NSW: Independent legal representative, direct legal representative & Guardian ad Litem
France	Civil law	Ad hoc administrator (<i>Administrateur ad hoc</i>) & Children's lawyer (<i>avocat d'enfant</i>)
The Netherlands	Civil law	General and Filiation guardian ad litem (<i>Bijzondere curator</i>) & separate legal representation
South Africa	Mixed legal system	Legal representative & Curator ad litem

1.2.2. Sources

A wide array of sources will be used in this comparative research, as discussed above per subquestion. With regards to the methodology, a potential validity threat regarding the sources is the issue of language. I can read the sources in the original language easily for most jurisdictions, for some other jurisdictions I am also sufficiently capable of reading the language (French) to make use of primary sources. In addition, enough source material is available to me in other languages allowing me to be able to check the correctness of the claims made in the sources.⁴⁸ Therefore, the falsifiability of the claims is protected.

This research is exploratory and limited to the law in the books in order to be able to describe how child representation is regulated. Although these sources may suffice to discuss how it is regulated,⁴⁹ no conclusions can be drawn with regards to which regulation works best. The child

⁴⁶ See Section 51(xxi) and (xxii) of the Constitution of Australia.
⁴⁷ Shackel 2016, p. 47.
⁴⁸ De Boer 1992, p. 46.

⁴⁹ Oderkerk 2015, p. 616.

representative may look very promising on paper, but not function at all in practice. I hope to be able to further empirically investigate one or two of the forms of child representation abroad in follow-up doctoral research. In light thereof, this research aims to make the first step, by evaluating how child representation is regulated in jurisdictions across the world in light of Article 12 UNCRC, and not to determine a 'better law', because an evaluation of the law in action will not be made.

1.3. Terminology

The core subject of this research is representation for the child, which falls within the ambit of child participation and the child's right to be heard. The right to be heard which is referred to as concerning the child's 'views' in the text of Article 12 UNCRC, but is also referred as concerning the child's 'voice', 'wishes' or 'opinions', terms which had been used in previous iterations.⁵⁰ Various terms are used, often interchangeably, without taking into account the nuanced differences and without clarifying what is understood under the term, e.g. 'the right to be heard' and 'the right to participate^{1,51} In this section, a few brief remarks will be made with regards to the terminology in this research.

The use of various shorter and interchangeable terms to convey the content of Article 12 UNCRC has been strongly criticized by Lundy. Although she recognizes that it is more convenient and catchy to use other terms to refer to the child's right to express his or her own views freely and for those views to be given due weight,⁵² she highlights the potential dangers of doing so. Firstly, that substitutes potentially weaken the impact of the right by incorrectly summarizing what the content of the right is.⁵³ Secondly, the shortened versions, e.g. 'the right to be heard' or 'the voice of the child', weaken the impact of the right because they only concern specific individual aspects of Article 12 UNCRC and not the right as a whole. Both of these dangers lead to an even greater danger, that the popularity of the shorter terms can overshadow the – perhaps less popular – changes which need to be achieved in order to correctly implement the right of Article 12 UNCRC.⁵⁴ To address this problem, Lundy does not argue for the consistent use of the phrase 'the child's right to express his or her own views freely' or to get rid of all the shorter popular terms and phrases in circulation, but instead proposes a model to conceptualize Article 12 UNCRC. While this model is compelling, the four interconnected factors of which it consists (Space, Voice, Audience and Influence) do not provide new short, catchy, and convenient terms or phrases which are immediately understandable in texts. As the Committee on the Rights of the Child itself, as well as a multitude of other organizations use the shorter phrases in all variations to refer to Article 12 UNCRC, it appears a necessary risk to take until a suitable user-friendly replacement is found.

The term 'participation' is frequently used in the debate on how to understand and implement Article 12 UNCRC.⁵⁵ This is curious, because the term itself does not appear in the text of Article 12.⁵⁶ The Committee also drew attention to this fact in the General Comment on Article 12, noting that the implementation of Article 12 'has been broadly conceptualized as participation' and that the

⁵⁰ Sutherland 2013, p. 342.

⁵¹ Cashmore 2011, p. 515.

 ⁵² Lundy 2007, p. 930 and 933.
 ⁵³ Lundy 2007, p. 930.
 ⁵⁴ Lundy 2007, p. 931-933.

⁵⁵ See, for example, Cleophas & Assim 2015; Hale 2006; Kaufman & Flekkoy 1998; Lowe & Murch 2001; Morag, Rivkin & Sorek 2012; Pérez Manrique 2007; Smits 2015; Taylor, Tapp & Henaghan 2007; Taylor et al. 2012; Tisdall & Morrison 2012; and Tisdall et al. 2004.

⁵⁶ Krappmann 2010, p. 501.

term 'participation' is 'now widely used'.⁵⁷ The term 'participation' is, according to Milne, a 'buzzword' which is vague as it captures 'both everything and nothing'.⁵⁸ On a more positive note, various authors have noted that 'participation' is a 'multi-faceted concept'⁵⁹ or a 'container concept'.⁶⁰ and is potentially much broader than the right provided for in Article 12 UNCRC.⁶¹ In any case, one can conclude that it is a comprehensive and complex term, as Herbots and Put pose it is 'not easily encapsulated by a single definition^{2,62} However, they themselves are one of the various academics who have critically examined the term 'participation' to form a ladder,⁶³ disc,⁶⁴ or other model of participation.65

While these models will not be discussed in this research, it is interesting to note that most focus on participation in the sense of democratic participation in communities, not on participation of children individually in legal proceedings. Hart's ladder was based on a citizenship model and applies best to (community) projects,⁶⁶ Treseder's and Shier's models were both geared to organizations aiming to work with or already working with children,⁶⁷ Thomas' theory only concerns collective decision-making,⁶⁸ and Lundy's model was developed in the context of education.⁶⁹ This shows how the term 'participation' has, to a certain extent, been appropriated by the discussions on children's citizenship and their role in public life. However, participation not only applies to community decision-making, but also to individual decision-making on the level of the child,⁷⁰ e.g. in family law proceedings. This is also acknowledged and included in the participation disc proposed by Herbots and Put. One of the four components of their disc is 'the context of participation'.⁷¹ The disc differentiates between the social level at which the participation is to take place, e.g. whether at the micro system level of the family or the macro system level of the community, as well as between private or public topics concerning the child.⁷² The meaning designated to 'participation' by the Committee also does not focus on one topic or level of participation. Instead, 'participation' is understood to: 'describe ongoing processes, which include information-sharing and dialogue between

⁵⁷ UNCRC General Comment No. 12 2009, para. 3.

⁵⁸ Milne 2015, p. 88.

⁵⁹ Sutherland 2013, p. 354.

⁶⁰ Herbots & Put 2015, p. 166.

⁶¹ Parkes 2013, p. 15.

⁶² Herbots & Put 2015, p. 156.

⁶³ Based on Arnstein's ladder of citizen participation, Hart's ladder of child participation is perhaps the most well-known model of participation (Hart 1992, see p. 8 for a figure of the ladder). This ladder has been further adopted and redesigned by, amongst others, Treseder (1997). See also, Parkes 2013, p. 16-20.

⁶⁴ Herbots & Put 2015, see p. 167 for a figure of their participation disc.

⁶⁵ Other models include Shier's model (Shier 2001, see p. 111 for a figure), Thomas' theory (Thomas 2007) or Lundy's model of child participation (Lundy 2007, see p. 932 for a figure of the model).

⁶⁶ Hart 1992. See also Black 1994, p. 29 in which this was seen as a limit to Hart's ladder.
⁶⁷ Treseder 1997; and Shier 2001.
⁶⁸ Thomas 2007, p. 199.

⁶⁹ Lundy 2007.

⁷⁰ Thomas 2007, p. 199.

⁷¹ Herbots & Put 2015, p. 156.

⁷² Herbots & Put 2015, p. 158-159.

children and adults based on mutual respect, and in which children can learn how their views and those of adults are taken into account and shape the outcome of such processes'.⁷³

⁷³ UNCRC General Comment No. 12 2009, para. 3.

2. Children's right to representation in the UN Convention on the Rights of the Child

The most-widely ratified international human rights treaty, the UN Convention on the Rights of the Child safeguards children's rights to provision, protection and participation. The right to participation is safeguarded in Article 12 which determines that children should be provided the opportunity to express their views in all matters concerning them and for these views to be given due weight. More specifically, in 'any judicial and administrative proceedings affecting the child', the child must be provided the opportunity to be heard, whether directly or through a representative.

In this chapter, some general remarks with regards to Article 12 of the UNCRC will first be made in section 2.1. This is followed by section 2.2, which further dissects Article 12 and section 2.3 on the relationship between Article 12 and the other rights contained in the UNCRC. Finally, a summary of this chapter is included in section 2.4. The conclusions made in this chapter will be used to analyze the legal frameworks of child representation in Australia, France, the Netherlands and South Africa.

2.1. General remarks on Article 12 UNCRC

The Committee on the Rights of the Child has emphasized that the Convention has four general principles.⁷⁴ These general principles are those rights in the Convention which are vital for the effective implementation of all the rights guaranteed by the Convention. The right to be heard (Art. 12) is one of these four general principles, together with the principle of non-discrimination (Art. 2), the principle that the best interests of the child should be a primary consideration (Art. 3) and the right to life, survival and development (Art. 6). As one of the general principles, the right to be heard, often referred to as the right to participation, provides a vehicle for children to exercise all the other rights contained in the UNCRC.⁷⁵ This means that Article 12 not only provides a substantive right as a means in itself, but also that it is a procedural right in relation to the other articles of the UNCRC.⁷⁶ Therefore, the correct implementation of Article 12 UNCRC at a national level assists the functioning of many other rights contained in the UNCRC.

Besides being a general principle, Article 12 UNCRC – and the right of participation that it provides for – is considered to be a core right of the Convention.⁷⁷ A principal goal of the Convention is to clarify that children also have human rights. Previously, the child's welfare perspective was predominant, because children were seen as 'incomplete' human beings: incompetent subjects who required protection.⁷⁸ The Convention departs from that long-standing view, in recognizing that, just like adults, children are individuals and subjects who hold rights.⁷⁹ In this regard, a lot of emphasis has been placed on the right of participation. Acknowledging that children have their own feelings and views and specifically listening to them, represents 'the recognition of, and respect for, the child's separate identity'.⁸⁰ As the Committee highlights, Article 12 UNCRC is a right granted to children based on their autonomous status as individuals, unlike most other rights which are based on either children's vulnerability - the protection rights - or children's dependency on adults - the provision rights.⁸¹ Therefore, according to the Committee the right 'should establish a new social contract' and it can be seen as the symbol for the recognition of children as right holders.⁸²

Article 12 UNCRC grants a right to participation in line with the understanding that children have competences, but also recognizes the limits of these competences and therefore does not grant a right to self-determination.⁸³ Children are recognized as full members of society, but they are not yet

⁷⁴ UNCRC General Comment No. 5 2003, p. 3-4.

⁷⁵ Parkes 2013, p. 5.

⁷⁶ Hodgkin & Newell 2007, p. 150.

⁷⁷ See Parkes 2013, p. 6.

 ⁷⁸ Krappmann 2010, p. 502.
 ⁷⁹ Hodgkin & Newell 2007, p. 149; and Lansdown 2016, p. 31.

⁸⁰ Sutherland 2014, p. 155.

⁸¹ UNCRC General Comment No. 12 2009, para. 18.

⁸² UNCRC Recommendations: Day of General Discussion 2006, Preamble.

⁸³ Hodgkin & Newell 2007, p. 150; and Herbots & Put 2015, p. 160.

given the full responsibilities.⁸⁴ This is because the view remains that children need to be protected and provided for and that this must be simultaneously provided together with the autonomy.⁸⁵ Although the UNCRC and Article 12 thereof do grant children more rights than before, a difference remains with adults. Children still have to grow, so they gradually acquire autonomy, competences and the right to self-determination throughout childhood.⁸⁶ Article 12 is one way in which children are granted the right to have a chance to exert their autonomy, through participation, in a safe environment, while they are not responsible for the final decision.

The movement started by the UNCRC towards recognizing children as rights holders is generally lauded. But, according to Milne, the movement is unfinished, as the focus on protection and provision remains controlling.⁸⁷ The critique is that most UNCRC rights objectify children because instead of recognizing the same rights as granted to adults for children, the UNCRC gives a comparable but not identical protection to children.⁸⁸ According to Milne, Article 12 only grants children a certain independence from adults and their opinion of what is best for children.⁸⁹ This issue is further discussed in section 2.3.2 below, in light of the relationship between Article 12 and Article 3 UNCRC. At this point it is important to note that while Article 12 gives an impetus for legal and social change, in family law proceedings, adults as lawmakers, judges and representatives, often still determine how, when and if a child can participate. In the following section, the legal framework of Article 12 UNCRC within which these adults must offer children the chance to participate and be represented will be discussed.

⁸⁴ Herbots & Put 2015, p. 181.

⁸⁵ Herbots & Put 2015, p. 160.

⁸⁶ Lansdown 2016, p. 31.

⁸⁷ Milne 2015, p. 15.

⁸⁸ Milne 2015, p. 16.

⁸⁹ Milne 2015, p. 18.

2.2. Understanding Article 12 UNCRC

In this section Article 12 UNCRC will be dissected and analyzed. To come to a complete understanding, amongst others, the Committee Comment no. 12 will be considered and, most importantly, the text of Article 12, which states:

Article 12

1. States Parties shall assure to the child who is capable of forming his or her own views the right to express those views freely in all matters affecting the child, the views of the child being given due weight in accordance with the age and maturity of the child.

2. For this purpose, the child shall in particular be provided the opportunity to be heard in any judicial and administrative proceedings affecting the child, either directly, or through a representative or an appropriate body, in a manner consistent with the procedural rules of national law.

2.2.1. Article 12 paragraph 1 UNCRC

The first paragraph of Article 12 UNCRC contains the general right of the child to express his or her views and for those views to be taken seriously in all matters concerning the child. By starting with the phrase 'shall assure', Article 12 imposes a strict legal obligation on the State parties to take proactive steps.⁹⁰ As no State parties have made an express reservation regarding Article 12, this obligation must be implemented no matter potential cultural or traditional reasons not to.⁹¹ The cultural and traditional attitudes towards children are a serious obstacle towards implementing Article 12 successfully universally, as remarked by the Committee.⁹² It is questionable whether the attitudes are actually determined culturally, as the traditional attitude in all State parties has been not to hear children. The aim of the UNCRC and in particular of Article 12 is to shift these attitudes towards the acceptance of providing children an opportunity to express their views.

The next part of the first paragraph of Article 12, namely that this right concerns the child 'capable of forming his or her own views', should not be seen as a limitation, according to the Committee.⁹³ It is not, and should not, lead to an age limit – in law or in practice – on the right of children to express their views.⁹⁴ Governments, courts and others should always start with the presumption that the child is capable of forming his or her own views,⁹⁵ instead of assuming that the

⁹⁰ UNCRC General Comment No. 12 2009, para. 19; Lundy 2007, p. 933-934; and Sutherland 2013, p. 341.

⁹¹ See the United Nations Treaty Collection website for all the reservations and declarations made regarding the UNCRC, at: https://treaties.un.org/Pages/ViewDetails.aspx?src=IND&mtdsg_no=IV-11&chapter=4&clang=_en (last visited 19 March ⁹² UNCRC Recommendations: Day of General Discussion 2006, para. 9.

⁹³ UNCRC General Comment No. 12 2009, para. 20.

⁹⁴ UNCRC General Comment No. 12 2009, para. 21.

⁹⁵ Lansdown 2011, p. 20.

child is incapable.⁹⁶ This means that it is the State's duty to determine whether or not the child is capable of forming a view, the child does not have the duty to prove his or her capability.⁹⁷ As the UNCRC aims to remove the longstanding binary view of capacity and capability, that children do not have the capacity to form and express views at all, it should not be the case that Article 12 creates a new binary in the form of a lower age limit. The Committee finds that 'age should not be a barrier to the child's right to participate fully in the justice process'.⁹⁸ An age limit to determine capability or capacity would also contradict the findings developmental psychologists have made about the gradual development of children's capacities over time.⁹⁹ The Committee also refers to developmental findings, emphasizing that all children – including very young and disabled children – are capable of forming views and thus, that non-verbal or other expression forms must be taken into account.¹⁰⁰

So, when is a child capable of forming their own views? According to the Committee, it is 'not necessary that the child has comprehensive knowledge of all aspects of the matter affecting him or her'.¹⁰¹ This means that a child does not need to be able to understand all the aspects of the matter or be able to foresee the consequences of his or her views, instead it is sufficient if the child has 'sufficient understanding' to make his or her own views,¹⁰² which also include their own feelings, insights and concerns.¹⁰³ Furthermore, States should not make use of the protectionist argument that it is contrary to the child's best interest to withhold children the opportunity to express their views.¹⁰⁴ The mainly negative explanation of Committee as to when a child is capable of forming his or her own views in effect leaves it undefined. Does the Committee require a standard for capability relative to the matter and decision to be made? Or is it not of great importance, because the Committee actually aims at a very low standard of capability to ensure the inclusion of many children in the scope of the right? It remains unclear. Of importance, and strongly worded by Archard and Skivenes, is that 'a child should not be judged against a standard of competence by which even most adults would fail'.¹⁰⁵

This inclusive approach to granting all children, no matter their age, the right to express their views does not mean that these views should always be determinative in the matter concerned. Firstly, because, as previously discussed, Article 12 UNCRC is a right to participation not a right to self-determination. More importantly, because the first paragraph of the article determines that the due weight to be given to these views should be **'in accordance with the age and maturity of the child'**. On the one hand this limits the weight given to the child's expressed views on grounds of age and

⁹⁶ UNCRC General Comment No. 12 2009, para. 20.

⁹⁷ Krappmann 2010, p. 507.

⁹⁸ UNCRC Recommendations: Day of General Discussion 2006, para. 51.

⁹⁹ Parkinson & Cashmore 2008, p. 4.

¹⁰⁰ UNCRC General Comment No. 12 2009, para. 21.

¹⁰¹ UNCRC General Comment No. 12 2009, para. 21.

¹⁰² UNCRC General Comment No. 12 2009, para. 21.

¹⁰³ Lansdown 2011, p. 20.

¹⁰⁴ Hodgkin & Newell 2007, p. 153.

¹⁰⁵ Archard & Skivenes 2009, p. 10.

maturity. On the other hand, this phrase is an obligation for States to give serious consideration to the child's views when making a decision.¹⁰⁶ As the Committee notes, 'simply listening to the child is insufficient'.107

The phrase refers to the 'age and maturity' of the child instead of repeating the term 'capable'. Instead of referring to capability, the phrase 'age and maturity' refers to the capacity of the child according to the Committee.¹⁰⁸ The phrase consists of two factors, because age, by itself, is insufficient to determine the child's capacity or the significance that should be attributed to his or her views.¹⁰⁹ A child's capacity does not only evolve with age, as is the case with capabilities, but also depends on the child's individual situation (e.g. environment and culture) and experiences.¹¹⁰ The capacity of a child can also be limited due to disabilities, although the Committee urges States to ensure that children with disabilities are provided with the necessary means to be able to express their views.¹¹¹ The second mentioned factor to determine the child's capacity is maturity. What is maturity? Even the Committee recognizes that it 'is difficult to define', but states that in this regard it concerns the child's ability 'to understand and assess the implications' of the matter and whether the child can express his or her views 'in a reasonable and independent manner'.¹¹²

The combination of age and maturity requires that the capacity of the child and his or her views expressed have to be assessed case-by-case.¹¹³ It is important to do so more thoroughly the greater the impact the final decision has on the child's life.¹¹⁴ It must especially be done carefully, because determining the child's capacity is generally dependent on adults' perceptions and adults may underestimate the capacity of a child.¹¹⁵ That is why Archard and Skivenes recommend that the assessment of the child's maturity 'be made independently of an evaluation of the child's opinion', because otherwise immaturity may be concluded simply because the adult disagrees with the child's view.116

There are two final important phrases in the first paragraph of Article 12 UNCRC. The phrase that children have the right to 'express those views freely' express two vital points. Firstly, that the child has the choice whether or not to exercise his or her right to express his or her views.¹¹⁷ It is a right for the child to participate, not an obligation. Secondly, it expresses the fact that the views should be expressed without the child being placed under any pressure or influence.¹¹⁸ The child should be able

¹⁰⁶ Lansdown 2011, p. 23.

¹⁰⁷ UNCRC General Comment No. 12 2009, para. 28.

¹⁰⁸ UNCRC General Comment No. 12 2009, para. 28.

¹⁰⁹ UNCRC General Comment No. 12 2009, para. 29.

¹¹⁰ UNCRC General Comment No. 12 2009, para. 29.

¹¹¹ UNCRC General Comment No. 12 2009, para. 78.

¹¹² UNCRC General Comment No. 12 2009, para. 30.

¹¹³ UNCRC General Comment No. 12 2009, para. 29. ¹¹⁴ Lansdown 2011, p. 23.

¹¹⁵ Lundy 2007, p. 937-938.

¹¹⁶ Archard & Skivenes 2009, p. 10.

¹¹⁷ UNCRC General Comment No. 12 2009, para. 16 and 22; and Sutherland 2013, p. 344.

¹¹⁸ UNCRC General Comment No. 12 2009, para. 22.

to express their views in a safe space, where they are encouraged and supported instead of fearing criticism or punishment.¹¹⁹

The second and final important phrase is that the child has the right to express their views in 'all matters affecting the child'. This phrase gives the right to participation protected in Article 12 UNCRC a very broad application.¹²⁰ Considering the history of this phrase makes it even more valuable. Article 7 of the revised Polish draft of the Convention of 1979 was the conception of what is now in Article 12 UNCRC. In that first version, the right was limited to expressing views 'in matters concerning his own person, and in particular, marriage, choice of occupation, medical treatment, education and recreation.¹²¹ During the 1981 Working Group discussions, the United States suggested a revised version of Article 7, expanding the list of matters by adding: 'religion, political and social beliefs, matters of conscience, cultural and artistic matters, ...,¹²² However, most delegations¹²³ were of the opinion that the right should not 'be subject to the limits of a list' and thus called for the deletion of the list.¹²⁴ The Working Group thus opted to remove the list and instead adopt the text with the phrase 'in all matters'.¹²⁵ Later, in the second reading and the Working Group discussion of 1989, Finland suggested adding to the phrase 'affecting the child'.¹²⁶ It was discussed whether or not it should be 'affecting the child' or 'affecting the rights of the child', but finally, the Working Group adopted the phrase as: 'in all matters affecting the child'.¹²⁷ According to the Finnish observer, that phrase could be interpreted as including matters affecting the rights of the child.¹²⁸ The phrase remained unchanged in the final text and thus went from a limited list to ensuring the broad application of the first paragraph of Article 12 UNCRC.¹²⁹

2.2.2. Article 12 paragraph 2 UNCRC

The second paragraph of Article 12 UNCRC contains the child's right to be heard in judicial and administrative proceedings. The paragraph was first introduced in 1981 by the United States as a paragraph in Article 3.¹³⁰ While some, at that time, were of the opinion that the paragraph added nothing new as the child's right to express views was already incorporated in another article, most were in agreement that the specific reference to legal proceedings had added value and that it was logical to include it in Article 3 as the views of the child would be a way to ascertain their best

¹¹⁹ Lansdown 2011, p. 22.

¹²⁰ Sutherland 2013, p. 343-344.

¹²¹ UNCHR Question of a Convention on the Rights of the Child 1980.

¹²² UNCHR Report of the Working Group 1981, para. 76.

¹²³ It is not stated which delegations these were, but representatives of twenty-one States attended the Working Group meetings in 1981, see: UNCHR Report of the Working Group 1981, para. 5. ¹²⁴ UNCHR Report of the Working Group 1981, para. 76.

¹²⁵ UNCHR Report of the Working Group 1981, para. 79 and 80.

¹²⁶ UNCHR Report of the Working Group 1989, para. 235.

¹²⁷ UNCHR Report of the Working Group 1989, para. 253.

¹²⁸ UNCHR Report of the Working Group 1989, para. 242 and 243.

¹²⁹ Krappmann 2010, p. 504.

¹³⁰ UNCHR Report of the Working Group 1981, para. 20.

interests.¹³¹ However, in 1989 the Working Group did decide to move the paragraph to, what is now, Article 12 UNCRC, because the scope overlapped and it was more logical to include it in the general participation principle.¹³² The fact that this paragraph fits in Article 12 UNCRC is emphasized by the starting phrase 'for this purpose' and the phrase 'shall in particular'. This shows that the second paragraph contains the specific right for children to express their views and have them taken into account in legal proceedings.

What does the second paragraph provide for specifically as a right? First of all, 'the opportunity to be heard'. The term 'opportunity' signals that children can make use of their right to be heard if they so wish. The right to be heard in legal proceedings must therefore be made accessible and childappropriate, barriers, such as costs and lack of legal counsel, must be eliminated.¹³³ The phrase 'to be heard' is more judicial then the phrases expressing and giving due weight to the expressed views as used in the first paragraph, but it encompasses the same even if in most jurisdictions conditions are attached to the right to be heard and not to the right to express views.¹³⁴

Secondly, the right is provided for 'in any judicial and administrative proceedings affecting the child'. The Committee explicitly emphasizes that this means 'all relevant judicial proceedings affecting the child, without limitation', referring as an example to, amongst others, family law proceedings such as the separation of parents, custody, care and adoption, but also to criminal law, health care and refugee cases.¹³⁵ The Committee also clarified that no distinction should be made between proceedings initiated by the child him- or herself or those initiated by others.¹³⁶

The next phrase – perhaps the most important in light of this research – is that the opportunity to be heard should be provided 'either directly, or through a representative or an appropriate **body**'. Before discussing what this means, it is interesting to briefly look at the *travaux préparatoires* concerning this phrase. In the first version of the paragraph, the opportunity was to be granted to the child 'as an independent party to the proceedings'.¹³⁷ The representative of the Netherlands then suggested that the phrase, 'directly or indirectly through a representative' be added, with other representatives agreeing with the addition that 'independent' be removed.¹³⁸ The phrase after the first reading thus read that children should be provided the opportunity to be heard, 'either directly or indirectly through a representative, as a party to the proceedings'.¹³⁹ The final transformation to the phrase as it stands today occurred when the Finnish representative (on behalf of a drafting group)

¹³¹ UNCHR Report of the Working Group 1981, para. 27 and 28.

¹³² UNCHR Report of the Working Group 1989, para. 128 and 236. See also, Hodgkin & Newell 2007, p. 155.

¹³³ UNCRC General Comment No. 12 2009, para. 34; and Lansdown 2011, p. 52.

 ¹³⁴ Krappmann 2010, p. 506.
 ¹³⁵ UNCRC General Comment No. 12 2009, para. 32.

¹³⁶ UNCRC General Comment No. 12 2009, para. 33.

¹³⁷ UNCHR Report of the Working Group 1981, para. 20.

¹³⁸ UNCHR Report of the Working Group 1981, para. 30.

¹³⁹ UNCHR Draft Convention on the Rights of the Child 1987, p. 6.

proposed to move the paragraph from Article 3 to – what is now – Article 12 UNCRC.¹⁴⁰ In doing so, the phrase 'as a party to the proceedings' was removed and 'or an appropriate body' was added. It is unclear what motivated these changes, just as it is unclear what the Committee means with 'an appropriate body'.¹⁴¹ In any case this phrase now provides a child with another choice. Once he or she has opted to be heard in a proceeding, he or she should be provided the opportunity to decide how to be heard.¹⁴² According to Parkes, the paragraph implies that having a representative must be available as an option to children in proceedings.¹⁴³ Newell and Hodgkin interpret it differently, stating that States have the discretion to determine how the child's views should be heard.¹⁴⁴ Although neither the Article itself, nor the Committee, goes so far as to put an obligation on States to provide for a form of child representation in legal proceedings, it does not seem to be the case that States can simply decide to provide children with only one manner in which to be heard. The remark of Newell and Hodgkin would mean that the child does not have a choice in how to be heard, while the Committee specifically has stated that the child should and that the alternative mechanism, e.g. representation, should be accessible and effective.¹⁴⁵

Concerning the representative, the Committee makes a few remarks. The Committee notes that various persons can be the representative of a child in proceedings: the parent(s), a lawyer or another person.¹⁴⁶ However, one must be cautious with having the parent(s) as a representative. Although they are the most obvious representatives, there is often the risk in proceedings that there is a conflict between the child and the parent(s), or between the two parents who ought to represent the child's interests jointly.¹⁴⁷ In those cases another representative is required. With regard to the duties of such a representative, the Committee notes that they should have sufficient knowledge and understanding of the proceeding and have experience in working with children.¹⁴⁸ The representative should ensure that the views of the child are 'transmitted correctly to the decision maker' and thus should also exclusively represent the interests of the child and not those of other persons.¹⁴⁹ With regards to the latter, it is important that the representative does not confuse his or her role with the obligation contained in Article 3 UNCRC (see also section 2.3.2).¹⁵⁰ The representative should represent the child's views and not merely his or her own views as to what is in the best interests of said child. Finally, the Committee notes that codes of conduct should be developed for these representatives.¹⁵¹

¹⁴⁶ UNCRC General Comment No. 12 2009, para. 36.

¹⁴⁰ UNCHR Report of the Working Group 1989, para. 235.

¹⁴¹ Parkes 2013, p. 255.

¹⁴² UNCRC General Comment No. 12 2009, para. 35.

¹⁴³ Parkes 2013, p. 98.

¹⁴⁴ Hodgkin & Newell 2007, p. 157.

¹⁴⁵ Sutherland 2013, p. 345.

¹⁴⁷ UNCRC General Comment No. 12 2009, para. 36.

¹⁴⁸ UNCRC General Comment No. 12 2009, para. 36.

¹⁴⁹ UNCRC General Comment No. 12 2009, para. 36 and 37.

¹⁵⁰ Lansdown 2011, p. 25.

¹⁵¹ UNCRC General Comment No. 12 2009, para. 37.

The final phrase of Article 12 paragraph 2 UNCRC is that all the above should be provided 'in a manner consistent with the procedural rules of national law'. As stressed by the Committee, this should not be read as permitting procedural rules of national law to restrict or prevent the enjoyment of the child's right to be heard.¹⁵² Instead, the States are encouraged to ensure that the child's right to be heard complies with 'the basic rules of fair proceedings'.¹⁵³

2.2.3. How to implement Article 12 UNCRC

In General Comment No. 12, the Committee on the Rights of the Child not only gave its interpretation of Article 12 UNCRC, but also outlined five steps that should be undertaken to 'effectively realize' the child's right to be heard in proceedings or other contexts. These five steps are: (1) sufficient preparation of the child through informing him or her of the available options and the impact it will have,¹⁵⁴ (2) the hearing of the child in an enabling and encouraging context,¹⁵⁵ (3) the case-by-case assessment of the capacity of the child and due weight given thereto, ¹⁵⁶ (4) the information about the weight given to the views of the child (the feedback) given to the child to guarantee that his or her views have been taken seriously,¹⁵⁷ and (5) the provision of complaints, remedies and redress procedures to children when their right to be heard has been breached or violated.¹⁵⁸ This fifth step has also been emphasized by the Committee with regards to the general measures of implementation of the UNCRC as a whole. State parties have to provide child-friendly access to complaints, remedies or redress procedures for when a child's right is breached, especially due to children's 'special and dependent status'.159

Of recurring importance in the implementation of Article 12 is the child's right to information. Information is essential for the child to be able to decide whether or not to freely express his or her views¹⁶⁰ as well as how he or she wishes to do so.¹⁶¹ Prior to the proceedings, the child should be informed in a child-friendly manner, inter alia, on how the proceeding will go, what the (potential) role of his or her representative is, what degree of confidentiality will be afforded to their views, how the decision will be made and what role his or her views will play in the reaching of that decision.¹⁶² Once the decision has been made, the right to information takes the form of the right to receive feedback, either by the decision maker or the child's representative, on how weight has been given to

¹⁵² UNCRC General Comment No. 12 2009, para. 38.

¹⁵³ UNCRC General Comment No. 12 2009, para. 38; and Lansdown 2011, p. 26.

¹⁵⁴ UNCRC General Comment No. 12 2009, para. 41.

¹⁵⁵ UNCRC General Comment No. 12 2009, para. 42.

¹⁵⁶ UNCRC General Comment No. 12 2009, para. 44.

 ¹⁵⁷ UNCRC General Comment No. 12 2009, para. 45.
 ¹⁵⁸ UNCRC General Comment No. 12 2009, para. 46 and 47.

¹⁵⁹ UNCRC General Comment No. 5 2003, para. 24.

¹⁶⁰ UNCRC General Comment No. 12 2009, para. 25.

¹⁶¹ Hodgkin & Newell 2007, p. 159.

¹⁶² Lansdown 2011, p. 51-52 and p. 64.

the child's views and how the child may be able to take further measures.¹⁶³ The importance of the child's right to information is also emphasized by the fact that the core obligations of State parties is, according to the General Comment, 'to introduce mechanisms providing children with access to appropriate information, adequate support, if necessary, feedback on the weight given to their views, and procedures for complaints, remedies or redress.¹⁶⁴

 ¹⁶³ Lansdown 2011, p. 57.
 ¹⁶⁴ UNCRC General Comment No. 12 2009, para. 48.

2.3. Article 12 and its relationship to other UNCRC articles

All the rights provided in the Convention on the Rights of the Child are interrelated. Especially the four general principles of the UNCRC, including Article 12 UNCRC, play an important role in relation to all other rights guaranteed. Therefore, Article 12 cannot be read in isolation and as it's relation to some of the other rights are of extra significance,¹⁶⁵ these will briefly be discussed in this section.

2.3.1. Article 2 and Article 12 UNCRC

The right to non-discrimination in Article 2 UNCRC is also applicable in the context of Article 12 UNCRC. As the Committee stresses, the right provided in Article 12 must be provided to children without discrimination on grounds of race, color, sex, language, religion, political or other opinion, national, ethnic or social origin, property, disability, birth or other status.¹⁶⁶ Therefore, States must also pay extra attention that vulnerable or marginalized groups of children are afforded the opportunity to be heard.¹⁶⁷

2.3.2. Article 3 and Article 12 UNCRC

The principle of Article 3 UNCRC, that the best interests of the child shall be a primary consideration in all actions concerning children, is interdependent with Article 12 UNCRC according to the Committee.¹⁶⁸ Both the General Comment on Article 12 as well as the General Comment concerning Article 3 UNCRC stress the importance of the relationship between the two articles. In the General Comment on the best interests of the child, the Committee emphasizes that the child should be provided a representative, either legal or as a guardian ad litem, in judicial proceedings concerning their interests.¹⁶⁹ In the General Comment on Article 12 UNCRC the importance of the hearing child's views as a method in determining the best interests of the child is also emphasized: 'there is no tension between articles 3 and 12, only a complementary role'.¹⁷⁰ The Committee stressed the symbiotic relationship between the two articles, in which Article 3 reinforces Article 12 by placing the child at the center of decisions concerning them and the right to be heard of Article 12 is indispensable for the correct application of Article 3.¹⁷¹ Accepting the complementary role of the two articles is crucial to effect the right to be heard, as the best interests principle should not be used to 'trump' the child's right to be heard.¹⁷²

¹⁶⁵ Lücker-Babel 1995, p. 393.

¹⁶⁶ UNCRC General Comment No. 12 2009, para. 75.

 ¹⁶⁷ Lansdown 2011, p. 32; and UNCRC General Comment No. 12 2009, para. 77 and 78.
 ¹⁶⁸ UNCRC General Comment No. 12 2009, para. 68; and UNCRC General Comment No. 14 2013, para. 43.

¹⁶⁹ UNCRC General Comment No. 14 2013, para. 90 and 96.

¹⁷⁰ UNCRC General Comment No. 12 2009, para. 74.

¹⁷¹ UNCRC General Comment No. 12 2009, para. 74.

¹⁷² Lansdown 2011, p. 33.

However, in practice the relationship might not be as harmonious as presented by the Committee.¹⁷³ For example, one can say there is a tension between Article 3 and Article 12 because the determination of the best interests of the child in a legal proceeding is up to the discretion of the judges and can completely diverge from the views expressed by the child.¹⁷⁴ But, this is in fact not a tension. As discussed above Article 12 is not a right to self-determination, children are not granted the right to make a choice in the decision, instead Article 12 is a procedural right. As long as the child has been granted the opportunity to express his or her views and these views have been given due weight in the decision-making process, it is acceptable that the judge's decision in the best interests of the child diverges from the child's view on the matter.

Another practical tension between Article 3 and Article 12 UNCRC is that certain assumptions about the best interests of the child can limit the meaningful participation of the child in proceedings.¹⁷⁵ This tension arises because the UNCRC requires adults to paternalistically determine the best interests of the child while at the same time considering the child's – potentially contradictory – views.¹⁷⁶ An example of this practical tension is that when the child's competence has to be determined, adults often have the tacit assumption that when 'children use different criteria for making choices, those are necessarily defective or at least inferior to adult criteria.¹⁷⁷ Therefore, the adults' assumption of what is in the best interests of the child can cut-off the child's opportunity to express his or her views. Although it is very likely and understandable that this 'tension' occurs in practice, the Articles themselves and the understanding thereof as held by the Committee require adults, especially decision-makers, to provide children with child-friendly, enabling and encouraging opportunities to express their views and require these adults to also take these views seriously. One can therefore conclude, that the view of the Committee that there is 'no tension' between Article 3 and Article 12, is perhaps a tad idealistic but should be strived for.

2.3.3. Article 5 and Article 12 UNCRC

Article 5 UNCRC concerns parental guidance and the child's evolving capacities, providing that the State parties shall respect the parents' responsibilities, rights and duties to give direction and guidance to the child in the exercise of his or her rights. The continually shifting equilibrium between children and their parents due to the child's evolving capacities as protected in Article 5 is closely linked to Article 12 UNCRC.¹⁷⁸ According to the Committee, the latter stimulates the requirement in Article 5 for parents to listen to the child and to, finally, exchange views with them on an equal footing.¹⁷⁹ The

¹⁷³ Lücker-Babel 1995, p. 394; Lundy 2007, p. 938; and Sutherland 2013, p. 347.

¹⁷⁴ Raitt 2004, p. 153.

¹⁷⁵ Raitt 2004, p. 153.

¹⁷⁶ Archard & Skivenes 2009, p. 2.

¹⁷⁷ Thomas & O'Kane 1998, p. 151.

¹⁷⁸ Lansdown 2011, p. 36-37.

¹⁷⁹ UNCRC General Comment No. 12 2009, para. 84 and 85.
relationship between Articles 5 and 12 UNCRC does not have a specific relation to the right to representation in family law proceedings, but it does emphasize the emancipation of children.¹⁸⁰

2.3.4. Articles 13 and 17 and Article 12 UNCRC

Two of the other civil rights and freedoms articles in the UNCRC are closely linked to Article 12, Articles 13 and 17 UNCRC. Both these articles are 'crucial prerequisites' for the child to effectively exercise his or her right to be heard, according to the Committee.¹⁸¹ Article 13 contains the right to freedom of expression. The difference between Articles 12 and 13 is that the latter is more broad because it concerns the child's right to hold and express any and all opinions without any restriction by the State, while Article 12 concerns the right to express views in specific matters concerning the child. Although both articles are very different, the combined effort of the Articles to afford children the (safe) space to express their views is important.¹⁸²

Article 17 provides for the child's right to have access to appropriate information. The article mostly focuses on (mass) media, but also provides for the right to information relating to their rights, proceedings affecting them, national legislation, available services, etc.¹⁸³ Especially the latter is, as discussed above, of importance in the implementation of Article 12. If the child is not informed, he or she cannot freely exercise the right to be heard.

While both Article 13 and Article 17 UNCRC are linked to the child effectively exercising the right to express his or her views, they are not of great relevance for the right of the child to representation in family law proceedings. As both articles concern the child's civil rights and freedoms they are more related to the child's right to participate in more public situation, for example in school and at the community level. With regards to the right of the child to representation in family law proceedings, Article 13 does signal the emancipation of children and Article 17 does emphasize the right to information, but both of those aspects are also covered by Article 12 specifically.

2.3.5. Other Articles and Article 12 UNCRC

In various other articles of the UNCRC relating specific situations, the child's right to be heard, as generally protected in Article 12 UNCRC, is reaffirmed. In proceedings where the child is to be separated from his or her parents and placed in alternative care, Article 9(2) UNCRC requires that 'all interested parties shall be given an opportunity to participate in the proceedings and make their views known'. The child who is the subject of the proceedings is one of these interested parties and thus must be given the opportunity to be heard.¹⁸⁴ Article 21(a) UNCRC concerning adoption (or *kafalah* in Islamic law) proceedings also states that the 'persons concerned' must have given their informed

¹⁸⁰ Lansdown 2011, p. 37.

¹⁸¹ UNCRC General Comment No. 12 2009, para. 80.

¹⁸² UNCRC General Comment No. 12 2009, para. 81.

¹⁸³ Lansdown 2011, p. 36.

¹⁸⁴ Lansdown 2011, p. 66.

consent to the adoption. This article does not specifically mention the child, but 'persons concerned' can be read to include the child¹⁸⁵ and the Committee has underscored the vital importance of hearing the child in adoption proceedings with regards to Article 3 and Article 12 UNCRC.¹⁸⁶ Lastly, in penal judicial proceedings the rights afforded by Article 12 and Article 40 are of importance.¹⁸⁷ Child offenders have the right to be heard and the correspondent right to remain silent throughout the – entire – penal judicial process.¹⁸⁸ They also, following Article 40(2)(b)(ii) and (iii) UNCRC, have the right to legal assistance in the preparation and presentation of their defence and during the hearing.

¹⁸⁵ Ferreira 2013, p. 377.

¹⁸⁶ UNCRC General Comment No. 12 2009, para. 55 and 56.

¹⁸⁷ UNCRC General Comment No. 10 2007, para. 43-45.

¹⁸⁸ UNCRC General Comment No. 12 2009, para. 58.

2.4. Summary

This section will summarize the most important conclusions made in this chapter for the continuation of the thesis. The child's right to representation in judicial proceedings is predominantly safeguarded by Article 12 UNCRC. This article contains one of the four general principles, therefore playing a crucial role in the understanding and implementation of all other UNCRC rights. It is also a core right of the UNCRC in the sense that it stresses the general aim of the UNCRC to recognize children as individuals with rights, competences, and capabilities, albeit to some extent limited by the other key aim of the UNCRC, namely protecting children.

In understanding the text of Article 12 multiple key aspects were analyzed in both paragraphs of the article. In the first paragraph of Article 12, five aspects are important in understanding the scope of the right. Firstly, that Article 12 contains a strict legal obligation for all State parties. That the requirement of the child being capable of forming his or her views should not be seen as a limitation. There should be no age limit or assumption of incapability, as all children are to some extent capable of forming views. However, it remains unclear where the Committee places the dividing line between a child being capable or not, as well as how the Committee expects the capability to be determined. What is emphasized is that it is important to determine the capability of the child within the context of the proceeding on a case-by-case basis. Thirdly, an important aspect concerns a similar threshold, capacity, as due weight should be given to the views of the child in accordance with their age and maturity. This capacity must be determined by both factors on a case-by-case basis as reaching a specific age by itself is insufficient and maturity is 'difficult to define', but concerns the child's ability to understand and assess independently and reasonably the issue at hand. The last two important aspects are that children should be able to express the views 'freely' and that they must be able to do so in 'all matters affecting the child'.

The second paragraph of Article 12 UNCRC contains three aspects of importance, the first being that it is a more specified right for the child to be heard, namely in any judicial and administrative proceedings affecting the child. This means no limitations are placed on the types of proceedings, thus including all sorts of family law proceedings. Secondly, the right contained in Article 12(2) UNCRC must be provided in a manner consistent with the basic rules of fair proceedings in national law. The last aspect, that the child should be provided the opportunity to be heard either directly or through a representative, is of greatest importance in this research. Article 12 thus provides that the child should have a choice how to be heard, if they want to be heard. Therefore, it appears that State parties ought to provide for a form of child representation in legal proceedings. According to the Committee, if a form is provided then the representative must potentially be someone other than the child's parent(s), must have sufficient knowledge, understanding and experience in working with children in legal proceedings, must represent the child's views not merely

his or her own views of what is in the best interests of the child. The analysis in the following chapters will consider whether the various jurisdictions comply with these Committee guidelines.

Finally, in this chapter it has become clear that Article 12 should be applied without discrimination (ex. Art. 2 UNCRC), considering the child's evolving capacities (ex. Art. 5 UNCRC) and in combination with the child's right to information and freedom of expression (ex. Art. 17 and 13 UNCRC). With respect to the relationship between having the best interests of the child be a primary consideration in all aspects concerning them (Art. 3 UNCRC) and Art. 12 UNCRC, it can be concluded that although the Committee is convinced of their harmonious symbiotic relationship, practical tensions do arise on the one hand because the paternalistic welfare view of children remains more common than the view of children as capable right-holders and because it is not possible to determine in absolute terms what is in the best interests of the child.

3. Comparison and evaluation of the legal framework of child representation

In this chapter the legal framework of child representation in Australia (New South Wales), France, the Netherlands and South Africa will be compared and evaluated in light of the international human rights standard of Article 12 UNCRC. For each separate jurisdiction a report has been written answering the second and third sub-questions separately. These reports can be found as annexes attached to this research. Based on the information contained in these reports, a simultaneous comparison per theme will discuss the similarities and differences between jurisdictions. Each comparison will be combined with the direct evaluation of the legal frameworks. This evaluation is based on the conclusions drawn in the previous chapter regarding the minimum international standard of child representation afforded by the UNCRC. A summarized and combined overall evaluation will be included in the final section of this chapter (section 3.6).

The twelve questions of sub-questions 2 and 3 which have shaped the reports per jurisdiction, have been combined into five themes for the comparison. The researcher has determined these themes as the overarching issues included in the sub-questions. The first theme being the forms of representation (section 3.1), combining the answers to sub-questions 2, 2a, 3, and 3c on the forms of representation, their historical development, their general task and their complementarity to other manners in which children can be heard. The second theme, the types of proceedings in which children can be represented (section 3.2), compares the answers to sub-question 2b. The requirements applied to child representation is the third theme (section 3.3). It compares the answers to the two sub-questions 2d and 2e. The fourth theme concerns the decisions regarding child representation (section 3.4). This combines the sub-question 2f on who decides whether the child may be represented and sub-question 2c on when they can make this decision and until when the representative must act. Finally, the fifth theme covers three practical issues: what are the function requirements for child representatives, how should they execute their task, and who pays them (section 3.5). Thereby combining and comparing the answers to sub-questions 3a, 3b and 2g.

3.1. Forms of representation

All four jurisdictions have at least two forms of child representation alongside other options for the child's views to be expressed in family law proceedings. The forms of child representation in each of the jurisdictions can best be introduced and recognized by their general task, whether they are a best interests representative, representing and advancing the child's best interests, or a separate legal representative, representing the child's views by acting on their instructions. Distinguishing the forms of representation by these two general tasks occurs both in academic literature,¹⁸⁹ as mentioned in the introduction, as well as in the national jurisdictions themselves.¹⁹⁰ Thus, Table 2 below includes an overview of all the forms of representation per jurisdiction and their general task.¹⁹¹

			Gener	al task
			Best interests	Separate legal
		Forms of representation	representative	representative
	Federal	Independent children's lawyer (ICL)	Х	
Australia (AU)	New	Independent legal representative (ILR)	Х	
	South	Direct legal representative (DLR)		Х
	Wales	Guardian ad litem (GAL)	Х	
France (FR)	Ad hoc adn	ninistrator (AHA)	Х	
	Children's	lawyer (CL)		Х
The Netherlands	General gua	ardian ad litem (GGAL)	Х	
(NL)	Filiation gu	ardian ad litem (FGAL)	Х	
	Separate leg	gal representative (SLR)		Х
South Africa	South Africa Legal representative (LR)		X*	Х
(SA)	Curator ad	litem (CAL)	Х	

Table 2: The forms of representation and their general task per jurisdiction

*Only in exceptional circumstances.

Due to the division of powers in Australia, there is one form of representation provided in federal law and three other forms in the state of New South Wales. The federal independent children's lawyer and the independent legal representative in New South Wales are relatively similar in their task of representing the child's best interests in family law proceedings. The guardian ad litem in New South Wales also has a similar general task, but the execution of that task is different from that of the independent children's lawyer and the independent legal representative, as will be discussed later. In France and in the Netherlands there is a clear division of the general task between the two forms of child representation. Both have a best interests representative who represents the child's interests, respectively the *administrateur ad hoc* (hereafter: ad hoc administrator) and the *bijzondere curator* (hereafter: guardian ad litem¹⁹²) in both the general (Art. 1:250 Dutch Civil Code (DCC)) and filiation

¹⁸⁹ See, amongst others, Daly 2016; Fernando 2013; Koh Peters 2005; Parkes 2013; and Parkinson & Cashmore 2008.

¹⁹⁰ See, for example, the difference between the independent legal representative and the direct legal representative in New South Wales, Australia in section 5.1.2 of the annexes.

¹⁹¹ For extended information see sections 5.1.1 and 5.1.2 (AU); 5.2.1 and 5.2.2 (FR); 5.3.1 and 5.3.2 (NL); and 5.4.1 and 5.4.2 (SA) of the annexes.

¹⁹² The translation of *bijzondere curator* as guardian ad litem has been chosen, because it is the translation adopted by the Dutch government. See, for example, UNCRC Third Periodic Report of the Netherlands 2008, para. 63.

form (Art. 1:212 DCC), and a separate legal representative acting on the instructions of the child, respectively the *avocat d'enfant* (hereafter: children's lawyer) and the separate legal representative. The South African curator ad litem represents the child's best interests, the legal representative will also do so if the child is very young and cannot give instructions, but generally the legal representative is a client-directed advocate representing the child's views.

Although it is interesting to note these different general tasks of the child representatives, it is potentially troublesome, considering Article 12 UNCRC and the analysis in the previous chapter, that most forms represent the child's best interests instead of being client-directed forms of representation sticking solely to the child's views. According to the Committee on the Rights of the Child the representative should exclusively represent the child.¹⁹³ This has been interpreted by Lansdown as meaning that the representative should not confuse their role of transmitting the views of the child with the obligation contained in Article 3 UNCRC that the child's best interests be a primary consideration.¹⁹⁴ However, it can also be interpreted as meaning that the child's representative may not also represent a parent at the same time. In that case there would be no issue with best interests representative, as long as they correctly transmit the child's views to the court and solely represent the child's interests.¹⁹⁵ It remains uncertain, however.¹⁹⁶

On a more positive note, the fact that each jurisdiction has multiple forms of representation available to children complies with Article 12(2) UNCRC, which requires that children should be able to decide how they wish to be heard according to the Committee.¹⁹⁷ As discussed in section 2.2.2, neither the Article nor the Committee explicitly require representation forms to be available to children. It is required however that the child has a choice between various mechanisms of being heard. Having two or more forms of representation available in each jurisdiction is the first step to achieving this. It remains the question, however, whether the alternative mechanisms are accessible and effective and which other mechanisms aside from representation are available. The latter will be answered in section 3.1.3 below. The accessibility and effectiveness depends in part on whether these forms of representation the same proceedings, which will be examined in section 3.2, and in part on whether the children themselves have the choice between alternative mechanisms, which will be discussed in section **Error! Reference source not found.**.

3.1.1. Historical development

To fully understand the forms of representation and the evaluation thereof with the minimum standard of Article 12 UNCRC, the historical development of the representations forms must be considered.

¹⁹³ UNCRC General Comment No. 12 2009, para. 37.

¹⁹⁴ Lansdown 2011, p. 25.

¹⁹⁵ UNCRC General Comment No. 12 2009, para. 36; see section 2.2.2 of the annexes.

¹⁹⁶ Parkes 2013, p. 253-254.

¹⁹⁷ UNCRC General Comment No. 12 2009, para. 35.

The historical development also informs the explanation of the similarities and differences between the jurisdictions in the comparison.

In each jurisdiction, there is one of form of representation with a longstanding history. Both the Dutch guardians ad litem and South African curator ad litem have Roman law origins. The *curator* in the Roman-Dutch law of the 17th and 18th centuries had the duty to assist the minor in litigation.¹⁹⁸ This form of participation stayed throughout Old Dutch law and can be found in the Dutch codes of the 19th and 20th century before the introduction of the modern guardians ad litem.¹⁹⁹ With the transplant of Roman-Dutch law to South Africa, the curator ad litem continued its existence in the common law of South Africa.²⁰⁰ It was the English common law, which introduced its historic guardian ad litem to Australia prior to the codified Australian Family Law Act of 1975 (hereafter AFLA 1975).²⁰¹ France has the youngest oldest form of representation, with the ad hoc administrator being introduced as a representative to mitigate paternal power over children's property in 1901.²⁰²

Since the 1970s, child representation has gained more traction in the four jurisdictions studied. This has led to the introduction of new forms of representation as can be seen in the timeline below (Figure 1).²⁰³ The French children's lawyer and South African legal representative were both introduced prior to the adoption of the UNCRC, albeit for specific types of proceedings, namely divorce proceedings and child protection proceedings. All the other forms of representation first saw the light of day after the ratification of the UNCRC in each of the jurisdictions. In most cases, it can be concluded that the UNCRC has prompted the introduction of explicit legislation on child participation. That would be fitting with the strict legal obligation of Article 12 UNCRC that States should take proactive steps to 'assure' the child's right to be heard.²⁰⁴

Take Australia, for example, although the AFLA 1975 very generally introduced representation for children in family law disputes, it was not until after the ratification of the UNCRC and an explicit recommendation by the Committee on the Rights of the Child that the independent children's lawyer was introduced.²⁰⁵ The independent legal representative and direct legal representative were introduced eight years following the ratification of the UNCRC.²⁰⁶ The French legislature also aimed to conform to the obligations contained in Article 12 UNCRC when introducing the ad hoc

 ¹⁹⁸ De Bruin 2010, p. 65; and Helmholz 1978, p. 229-230 and 232. See sections 5.3.1.1 and 5.4.1.1 of the annexes.
 ¹⁹⁹ Asser 1957, p. 472-473 and 597; Helmholz 1978, p. 229-230 and 232; Veegens 1923, p. 223 and 256. See section 5.3.1.1 of the annexes.

²⁰⁰ De Bruin 2010, p. 50 and 371. See section 5.4.1.1 of the annexes.

²⁰¹ Australian Law Reform Commission 1997, para. 13.10; and Bates 2013, p. 48. See section 5.1.1.1 of the annexes.

²⁰² Loi du 6 avril 1910 pour la bonne administration des biens des mineurs. See Grevot 2010, p. 12; Fédération nationale des administrateurs ad hoc 2009, p. 12; and section 5.2.1.1 of the annexes. ²⁰³ For the more extended historical background of each of the forms of representation see sections 5.1.1.1, 5.2.1.1, 5.3.1.1,

and 5.4.1.1 of the annexes. ²⁰⁴ UNCRC General Comment No. 12 2009, para. 19. See also Lundy 2007, p. 933-934; Sutherland 2013, p. 341; and

section 2.2.1. ²⁰⁵ UNCRC Concluding Observations Australia 2005, para. 30. See also the Family Law Amendment (Shared Parental

Responsibility) Act 2006; Bell 2015, p. 3; and section 5.1.1.1 of the annexes.

²⁰⁶ See Ross 2013b, p. 334; and section 5.1.1.1 of the annexes.

administrator in 1993.²⁰⁷ More specifically, the amendments of 2007 which introduced the children's lawyer for family law proceedings were inspired, according to the French government, by the terms of Article 12 UNCRC.²⁰⁸

AU: Family Law Act introduces	1975 1979SA: Divorce Act introduces LR d 1981
AU: Ratification UNCRC FR: Ratification UNCRC	——————————————————————————————————————
FR: AHA introduced ——	 1993 NL: Ratification UNCRC & GGAL 1995 introduced SA: Ratification UNCRC
	SA: Constitution introduces right to separate legal representation & LR introduced
NSW: ILR & DLR introduced	1998 — NL: FGAL introduced
AU: ICL introduced	2006
FR: CL introduced in famil law proceedings	 ^y — 2007 2008 — NL: mandatory SLR in secure youth care 2009 — NL: role of GGAL extended
FR: AHA for child protection proceedi	2014 — NL: mandatory SLR for execution of care and supervision order proceedings ngs — 2016
CAL: Curator ad litemGAL: GuardiCL: Children's lawyerGGAL: Gene	ion guardian ad litemILR: Independent legal representativean ad litemLR: Legal representativeral guardian ad litemSLR: Separate legal representativedent children's lawyerSLR: Separate legal representative

Figure 1: Timeline of the introduction of representation forms in the jurisdictions

Of all four jurisdictions, the UNCRC had the most impact in South Africa where it was the first international human rights treaty to be ratified by its first universally elected democratic government.²⁰⁹ In 1996, the South African Constitution embedded a multitude of children's rights,

²⁰⁷ See Attias 2012; Bazin 2014; Fédération nationale des administrateurs ad hoc 2009, p. 12; Grevot 2010, p. 12; Le Mintier 1994, p. 1; and section 5.2.1.1 of the annexes. ²⁰⁸ UNCRC Third and Fourth Periodic Report of France 2008, para. 196. See also Loi n°2007-308 du 5 mars 2007 portant

réforme de la protection juridique des majeurs; and section 5.2.1.1 of the annexes. ²⁰⁹ Sloth-Nielsen 1996, p. 323.

including the right to representation, in s. 28.²¹⁰ This was an unequivocal response by the South African government to the international human rights obligations contained in the UNCRC.²¹¹ The Constitutional protection advanced further specific provisions, such as the introduction of the legal representative into the Child Care Act (which later became the Children's Act of 2005).²¹²

In contrast, although the Dutch forms of representation were introduced in the decades following the ratification of the UNCRC, it was not necessarily because of it as the Dutch government did not consider reforms necessary to conform to the UNCRC.²¹³

3.1.2. Sources of law

The historical background of each of the representation forms combined with the legal tradition of the jurisdictions, has influenced in which sources of law the regulation of the forms of representation are contained. Discussing the sources of law shows to what extent each of these regulations are crystallized by law and to what extent legal doctrinal writing is important in understanding the forms of representation.

In Australia, specifically New South Wales, the independent children's lawyer, independent legal representative and dependent legal representative are regulated extensively in statutes. For example, the section on the independent children's lawyer in the AFLA 1975 is three pages long and five sections are dedicated to the independent legal representative and the dependent legal representative in the Children and Young Persons (Care and Protection) Act 1998 (NSW) (hereafter CCPA (NSW)). In addition, for these three forms of representation clear guidelines exist: the federal Family Court's Guidelines for Independent Children's Lawyers and the New South Wales' Representation Principles for Children's Lawyers and the Care and Protection Practice Standards.²¹⁴ Thus, with the detailed statutes and guidelines available few gaps remain. For the New South Wales' guardian ad litem, the situation is a bit different. It's common law background explains why the statutes have remained brief regarding the guardian ad litem. In addition, there are no guidelines on the task of the guardian ad litem, requiring case law and doctrine for clarification.²¹⁵

In South Africa the situation is similar. On the one hand, the legal representative has been regulated in detail in the Children's Act of 2005 (hereafter SACA 2005), albeit less extensively compared to Australia. On the other hand, the curator ad litem is grounded in common law, which means that case law is crucial as statutes or written guidelines are lacking.²¹⁶

²¹⁰ UNCRC Initial Periodic Report of South Africa 1998, para. 1. See section 5.4.1.1 of the annexes.

²¹¹ UNCRC Initial Periodic Report of South Africa 1998, para. 1. See also the Constitutional Court in S v M (Centre for Child Law as Amicus Curiae) (CCT 53/06) [2007] ZACC 18 (26 September 2007), para. 16. Referred to in: Liefaard & Doek 2015.

²¹² Zaal & Skelton 1998, p. 541. See section 5.4.1.1 of the annexes.
²¹³ Limbeek & Bruning 2015, p. 89; and Liefaard & Vonk 2016, p. 314. See section 5.3.1.1 of the annexes.

²¹⁴ Guidelines for Independent Children's Lawyers 2013; Representation Principles for Children's Lawyers 2014; and Care and Protection Practice Standards 2015. ²¹⁵ See Bao-Er 2006, p. 4; and section 5.1.2.2 of the annexes.

²¹⁶ See sections 5.4.1.1 and 5.4.1.2 of the annexes.

The two civil law jurisdictions, France and the Netherlands have less extensive and detailed statutes. Instead, in both jurisdictions, there is simply one article in the civil code per form of representation, e.g. Article 1:250 DCC and Article 388-2 French Civil Code (Code Civil; hereafter FCC). In addition, some more general articles concerning legal aid may be applicable and in the Netherlands, there are two guidelines of the Landelijk Overleg Vakinhoud Familie- en Jeugdrecht (the National Consultations on Family and Child Law matters, hereafter: LOVF) concerning the guardians ad litem.²¹⁷ This means that legal doctrinal literature is crucial for the full understanding of the French and Dutch child representatives.

3.1.3. Relation to other manners of hearing the child's views

In addition to representation, all four jurisdictions also offer other possibilities for the child's views to be heard. Children can directly express their views to the judge in judicial meetings in Australia, France, and the Netherlands.²¹⁸ The form and execution of these judicial meetings varies in the three jurisdictions. The other general option is a welfare report including the child's views. In Australia family consultants can be requested to write a family report which should include the child's views on the matter.²¹⁹ In the Netherlands the Child Care and Protection Board (Raad voor de Kinderbescherming) can advise the court on the welfare of the child and should speak to the child, although this is not legally required.²²⁰ In South Africa the family advocate can make enquiries and collect information to report to the court about the child's welfare.²²¹ In doing so, the family advocate will generally interview the child, although they are not obliged to.

Complementarity of participation forms

Aside from knowing which forms of representation and other means of hearing the child's views in family law disputes exist in the four jurisdictions, it is also relevant to see whether these forms can simultaneously play a role. In Table 3 below, the complementarity of the manners in which the child can be represented or heard is displayed, with the non-representation forms shaded in dark.²²² In general, the table shows that the representation forms do not function in a complementary fashion, but that it is possible for the non-representation forms to supplement the representatives. The latter is especially the case in the Netherlands and South Africa. South Africa being the only jurisdiction where all the forms of children being heard can function together. The forms of representation in Australia and France are not often supplemented by other forms of hearing the child, because of the

²¹⁷ See Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014; and Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014. ²¹⁸ S. 60CD(2)(c) AFLA 1975, Art. 388-1 FCC, and Art. 809 DCCP. See sections 5.1.2.3, 5.2.2.3, 5.3.2.3, and 5.4.2.3 of the

annexes. ²¹⁹ S. 62G(2) AFLA 1975. See section 5.1.2.3 of the annexes.

²²⁰ Art. 810 DCCP. See section 5.3.2.3 of the annexes.

²²¹ S. 4 of the Mediation in Certain Divorce Matters Act 1987. See section 5.4.2.3 of the annexes.

²²² For the extended information see sections 5.1.2.3 (AU), 5.2.2.3 (FR), 5.3.2.3 (NL), and 5.4.2.3 (SA) of the annexes.

different types of proceedings in which representation, welfare reports or judicial meetings are called for.

Australia (NSW)										
	ICL	IL	R	DLR		GA	Ĺ	JM		FamR
ICL	-	I		-		I		Х		Х
ILR	-	I		-		Х		-		-
DLR	-	-		-		Х		-		-
GAL	-	Х		Х		-		-		-
JM	Х	-		-		-		-		-
FamR	Х	-		-		-		-		-
The Net	herland	ls								
	GGA	Ľ	FC	GAL	S	LR	JN	Λ	V	VR
GGAL	Х		1		Х	Κ	Х		Х	K
FGAL	-		1		I		Х		Х	K
SLR	Х		-		-		Х		Χ	K
JM	Х		Х		X	Κ	-		Χ	K
WR	Х		Х		X	Κ	Х		-	

France						
	AHA	CL	JM			
AHA	-	-	-			
CL	-	-	Х			
JM	-	Х	-			
South A	Africa					
	LR	CAL	FA			
LR	-	Х	Х			
LR CAL	- X	X -	X X			

Abbreviations:

AHA: Ad hoc administrator CAL: Curator ad litem CL: Children's lawyer DLR: Direct legal representative FA: Family advocate FamR: Family report FGAL: Filiation guardian ad litem GAL: Guardian ad litem GGAL: General guardian ad litem ICL: Independent children's lawyer

ILR: Independent legal representative JM: Judicial meetings LR: Legal representative SLR: Separate legal representative WR: Welfare report from CCPB

The complementarity of forms by which children can be heard is relevant as it clarifies whether children have an actual choice in how their views are heard. As discussed in section 3.1 above, the Committee's understanding of Article 12 UNCRC requires that alternative mechanisms are available for the child to choose between. For this, the existence of multiple forms of representation is important, but also whether these forms are available complementarily together with other mechanisms by which the child's views can be heard. Having complementary mechanisms available increases the child's options. For example, in the Netherlands children can be heard by judges themselves in a judicial meeting, but may at the same time have a general guardian ad litem who represents their best interests in court. Therefore, the manner in which children can opt to express their views in the Netherlands and South Africa in combination with representation, is positive. The options for children in Australia and France are in that sense, much more limited.

3.2. Types of proceedings

All the forms of representation mentioned in the previous section function in family law proceedings, but in which types of proceedings? The types of family proceedings included in the scope of this research and in which children can be represented can be divided into five categories.

The first category is proceedings concerning parental authority. All jurisdictions have at least one form of representation available in this type of proceeding. In Australia, the independent children's lawyer can be appointed for proceedings concerning parental authority, parental plans, parenting orders and child maintenance orders.²²³ In France a children's lawyer can assist the child and an ad hoc administrator may be appointed in cases concerning parental authority, e.g. the litigation of custody and access after parental separation.²²⁴ In the Netherlands it is the general guardian ad litem who may be appointed in proceedings concerning custody, residence and access.²²⁵ In South Africa both the curator ad litem and the legal representative can be appointed in proceedings concerning parental authority, for the legal representative this is specifically proceedings on the termination, extension, suspension or restriction of parental responsibilities and rights, the assignment of care and contact, guardianship, maintenance, custody and access.²²⁶

The second category of proceedings are those concerning parentage. In Australia the independent children's lawyer and in the Netherlands the filiation guardian ad litem can function in all forms of parentage proceedings, ranging from recognition proceedings to denial of parentage.²²⁷ In France and South Africa the role of representative is more limited in this category. The ad hoc administrator can represent the child in establishing or contesting parentage and the South African legal representative only plays a role in confirmation of paternity proceedings.²²⁸

The third category is child protection proceedings. These proceedings are different from other family law disputes in the sense that the state plays an active role in these types of proceedings through interfering in the family to protect the child. In Australia, France and South Africa the available forms of representation are involved in a wide range of these proceedings. The independent legal representative, direct legal representative and guardian ad litem in New South Wales can all function in proceedings ranging from emergency care orders, the provision of counselling or treatment services, and the development and review of care orders.²²⁹ The French ad hoc administrator and children's lawyer can both function in child protection proceedings, which of the two represents the child in these proceedings depends on the child's discernment (see section 3.3.1).²³⁰ In South

²²³ S. 68L AFLA 1975. See section 5.1.1.2 of the annexes.

²²⁴ Art. 388-1 FCC. See also, UNCRC Third and Fourth Periodic Report of France 2008, para. 197; Gouttenoire 2006, p. 62; Grevot 2010, p. 12; and section 5.2.1.2 of the annexes. ²²⁵ Art. 1:250 DCC. See Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 5; *Kamerstukken II*

^{2004/05, 301245, 3,} p. 7; Ter Haar 2015; and section 5.3.1.2 of the annexes. ²²⁶ S. 29(6)(a) SACA 2005 and S. 6(4) Divorce Act of 1979. See section 5.4.1.2 of the annexes.

²²⁷ S. 69P to 69ZD AFLA 1975 and Art. 1:212 DCC. See sections 5.1.1.2 and 5.3.1.2 of the annexes.

²²⁸ Art. 325 and 327 jo. Art. 328 or 334 FCC and S. 26(b) SACA 2005. See sections 5.2.1.2 and 5.4.1.2 of the annexes.

 $^{^{229}}$ S. 10 CCPA (NSW). See section 5.1.1.2 of the annexes.

²³⁰ Art. 375 jo. 388-2 FCC and Art. 375 FCC jo. 1186 FCCP. See section 5.2.1.2 of the annexes.

Africa the legal representative can represent the child in protection and well-being, support, provision of development or intervention services, temporary safe care or alternative care proceedings and in civil proceedings on child maltreatment, neglect and abuse.²³¹ In the Netherlands both the general guardian ad litem and the separate legal representative can play a role in child protection proceedings, but both in different sorts of proceedings. The general guardian ad litem may be appointed in proceedings concerning family supervision and placement orders, conflicts regarding foster care and when there is an injunction to request treatment. The separate legal representative is mandatory in authorization for secure youth care proceedings and in disputes on the execution of care and supervision orders and may be involved in proceedings on the termination of care and supervision orders, the revocation of written instructions concerning these orders, the adjustment of care orders due to changed circumstances and in requests to end or shorten the placement in care.²³²

The last two categories, **international child abduction** and (international and national) **adoption** proceedings, are more narrow. All jurisdictions have one form of representation available in international child abductions proceedings, the Australian independent children's lawyer, the French children's lawyer, the Dutch general guardian ad litem and the South African legal representative.²³³ Child representation in adoption proceedings is less uniform in the four jurisdictions. In New South Wales, Australia all three forms of child representation (independent legal representative, direct legal representative & guardian ad litem) can be appointed, just as in South Africa, where both a legal representative or curator ad litem can be appointed.²³⁴ In the Netherlands, only the filiation guardian ad litem can be appointed in adoption proceedings.²³⁵ Finally, in France representation can only occur in one very specific national adoption proceeding, namely in the revocation of a simple adoption both the ad hoc administrator and children's lawyer can represent the child.²³⁶

The similarities and differences between jurisdictions in the categories of cases in which a child can be represented, the different forms of representation and the different general tasks of these representation forms combined lead to a variety of representations as presented in Table 4.

²³¹ S. 55 jo. 45 and 150 to 160 SACA 2005. See section 5.4.1.2 of the annexes.

 ²³² Art. 6.1.10(4) Youth Care Act, Art. 1:262b jo. 1:265k(1), Art. 1:261, 1:264, 1:265d and 1:265g(2) DCC. See section 5.3.1.2 of the annexes.
 ²³³ S. 68L(3) AFLA 1975, Art. 388-1 FCC, Art. 1:250 DCC, and S. 279 SACA 2005. See sections 5.1.1.2, 5.2.1.2, 5.3.1.2

²³³ S. 68L(3) AFLA 1975, Art. 388-1 FCC, Art. 1:250 DCC, and S. 279 SACA 2005. See sections 5.1.1.2, 5.2.1.2, 5.3.1.2 and 5.4.1.2 of the annexes.

²³⁴ S. 9 Adoption Act 2000 and S. 233(1)(c) SACA 2005. See sections 5.1.1.2 and 5.4.1.2 of the annexes.

²³⁵ See Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 4 and 6; and section 5.3.1.2 of the annexes.

²³⁶ Art. 307-1 FCC. See section 5.2.1.2 of the annexes.

		General type	of representation
Categories of proceedings	Jurisdictions	Best interests representative	Separate legal representative
	Australia	X (ICL)	
Parental Authority	France	X (AHA)	X (CL)
	The Netherlands	X (GGAL)	
	South Africa	X (CAL)	X (LR)
	Australia	X (ICL)	
Parentage	France	X (AHA)	
	The Netherlands	X (FGAL)	
	South Africa		X (LR)
	Australia	X (NSW: ILR, GAL)	X (NSW: DLR)
Child Protection	France	X (AHA)	X (CL)
	The Netherlands	X (GGAL)	X (SLR)
	South Africa		X (LR)
	Australia	X (ICL)	
International Child	France		X (CL)
Abduction	The Netherlands	X (GGAL)	
	South Africa		X (LR)
	Australia	X (NSW: ILR, GAL)	X (NSW: DLR)
Adoption	France	X (AHA)	X (CL)
	The Netherlands	X (FGAL)	
	South Africa	X (CAL)	X (LR)

Table 4: Forms of representation available in different categories of proceedings in each jurisdiction per type of representation

Abbreviations:

AHA: Ad hoc administrator CAL: Curator ad litem CL: Children's lawyer DLR: Direct legal representative FGAL: Filiation guardian ad litem GAL: Guardian ad litem GGAL: General guardian ad litem ICL: Independent children's lawyer ILR: Independent legal representative LR: Legal representative SLR: Separate legal representative

Three evaluative conclusions can be drawn from the table and discussion above and by considering Article 12 UNCRC.

Firstly, that in all five categories of proceedings each jurisdiction has **at least one form of representation for children**. At first sight, it thus appears to comply with Article 12 UNCRC which grants children the right to express their views in 'all matters affecting the child' (para. 1) and more specifically, 'in any judicial and administrative proceedings affecting the child' (para. 2). There should be no limitation to the types of proceedings in which children can be heard.²³⁷ With regards to representation in the family law proceedings studied in this research, only one jurisdiction fulfills this requirement. In New South Wales, Australia, there are no limits to the types of proceedings in the five categories above. That is not the case in France, the Netherlands or South Africa. For example, in France representation in child protection proceedings is only available in select types of proceedings. According to the Committee, there should also be no distinction made between proceedings initiated

²³⁷ UNCRC General Comment No. 12 2009, para. 32. See section 2.2.1 and 2.2.2 of the annexes.

by the children themselves or those initiated by others.²³⁸ This requirement is met in all the jurisdictions.

The second conclusion that can be drawn is that there are differences between the types of proceedings in how children can be represented. There is a clear difference between the horizontal 'private' family law disputes, e.g. parental authority and parentage proceedings, and the vertical and more 'public' family law disputes, e.g. child protection and adoption proceedings. In the latter it is more common to provide children with a form of separate legal representation, instead of the best interests representation more generally afforded in disputes between parents. Why? Generally, it is considered that in vertical proceedings the child may require more procedural protection from the state interference in their private family matters and that the effects of the decision are more severe, e.g. children may be removed from their families. However, taking horizontal family law disputes less seriously and thereby differentiating between the two general types of proceedings, is not in conformity with Article 12 UNCRC.²³⁹ It is interesting to note that in the category of international child abduction proceedings, the split is equal between best interests and separate representation. Only South Africa provides for a strong mandatory right to representation.²⁴⁰ Australia applies additional requirements making an appointment exceptional, in France the child can only be assisted in their right to be heard, and in the Netherlands representation is possible in theory but in practice rarely employed due to the short timeframe available for international child abduction proceedings.²⁴¹

The third and final conclusion concerns the **division of tasks and types of cases**. In in some jurisdictions there is a clear division of tasks between the forms of representation and there is a clear division between the types of cases, while in other jurisdictions there is a clear division of tasks but no division between types of cases as both forms of representation can function in similar proceedings. Whether these divisions do or do not exist is relevant with regards to the children's right to choose the manner by which their views are heard as provided by Article 12 UNCRC. As discussed in section 3.1 above, the existence of multiple forms of representation in principle provides for a choice, but the actual choice depends on whether multiple forms of representation are accessible in the same types of proceedings. The clearest division of both tasks and types of proceedings is in the Netherlands, in each sort of proceeding there is only one available form of representation, except in a few very limited and exceptional child protection proceedings.²⁴² This means that, in practice, there is no choice for the child between the types of representation. In Australia there is the clear division of types of proceedings between the federal form of representation, the independent children's lawyer, and the New South Wales' forms of representation, the independent legal representative, the direct legal representative and the guardian ad litem. The latter forms are available in similar proceedings.

²³⁸ UNCRC General Comment No. 12 2009, para. 33.

²³⁹ Daly 2016, p. 7.

 $^{^{240}}$ S. 279 SACA 2005. See section 5.4.1.2 of the annexes.

²⁴¹ S. 68L(3) AFLA 1975, Art. 388-1 FCC, and Jonker et al. 2015, p. 45. See sections 5.1.1.2, 5.2.1.2 and 5.3.1.2 of the annexes.

²⁴² See sections 5.3.1.2 and 5.3.2.3 of the annexes.

therefore appearing to grant children a choice in the manner of representation. However, this choice may be limited by the further requirements discussed in the next section. Having two (or more) forms of representation available in the same types of proceedings also occurs to a limited extent in both France and South Africa. Again, providing the appearance of a choice, although in both these jurisdictions the question which of the forms of representation is actually available to the child in a specific procedure depends on the further requirements discussed below.

3.3. Requirements for representation

If children are involved in one the proceedings described above in which representation is available, then the question remains whether children are always represented? There are only three situations in which child representation is mandatory. These are the appointment of a filiation guardian ad litem in pending parentage proceedings and the appointment of a separate legal representative in authorization of secure youth care proceedings in the Netherlands,²⁴³ and the French ad hoc administrator who must be appointed if requested by the child's parent(s) within the framework of Article 383 FCC.²⁴⁴ By making representation mandatory in these proceedings, the Dutch and French legislatures must have considered these situations extraordinary. With regards to the separate legal representative in the authorization of secure youth care proceedings in the Netherlands, it was indeed the case that the legislature wanted to provide extra protection to the child due to the drastic effect of the decision.²⁴⁵

Although it is commendable that representation is not limited to any specific requirements in these situations, it does conflict with one of the core values of Article 12 UNCRC. Article 12 UNCRC comprises a right for the child to participate in proceedings, not an obligation.²⁴⁶ A child should have the option not to express his or her views. Therefore, it may be problematic that in the three above mentioned situations, children have mandatory representation. However, mandatory representation does not have to mean that children are obliged to express views. As long as children can opt not to express their views even though they are represented in these three situations, the standard of Article 12 UNCRC is not threatened.

A second issue concerning the mandatory representation and Article 12 UNCRC is that according to the Committee there should be no distinction made between proceedings initiated by the child versus those initiated by others.²⁴⁷ In the Netherlands, the distinction made between the mandatory filiation guardian ad litem in pending proceedings versus child related requirements for the appointment of a filiation guardian ad litem when proceedings are to be instituted by the child is contrary to Article 12(2) UNCRC. A filiation guardian ad litem should be mandatory no matter who institutes the proceedings. By applying additional requirements when the child initiates the proceedings, the Netherlands limits the child's opportunity to participate in parentage proceedings.

Aside from these three situations, further requirements are applied for all other types of proceedings and forms of representation. These requirements can either be child related, e.g. determining from what age or level of maturity children can be represented, or conflict related, e.g. determining when a representative can be appointed. Both sorts of requirements will be discussed in the following section for all jurisdictions and forms of representation.

²⁴³ Art. 1:212 DCC and Art. 6.1.10(4) Youth Care Act. See section 5.3.1.5 of the annexes.

 ²⁴⁴ Massip 1995; and Fédération nationale des administrateurs ad hoc 2009, p. 42 and 44. See section 5.2.1.5 of the annexes.
 ²⁴⁵ Kamerstukken II 2005/06, 30644, 3, p. 23. See also, Van Teeffelen 2008; and section 5.3.1.1 of the annexes.

²⁴⁶ UNCRC General Comment No. 12 2009, para. 16 and 22; Sutherland 2013, p. 344; and section 2.2.1.

²⁴⁷ UNCRC General Comment No. 12 2009, para. 33.

3.3.1. Child related requirements for representation

In the four studied jurisdictions, there are three ways in which child related factors play a role as a requirement or otherwise.

First of all, it is possible that there is **no explicit child related requirement**, although child related factors, such as the age of the child, may still influence the appointment or the general task of the representative. In Australia, the independent children's lawyer is not subject to specific child related requirements, but the criteria from the *Re K* decision by which judges determine whether or not to use their discretion, do contain two child related factors.²⁴⁸ This means that the appointment can be set aside if it is considered inappropriate due to the child's age or maturity or that the appointment may be required because the child is of mature years and has strong views. The South African curator ad litem is not restricted by any child related factors and the legal representative is also appointed regardless of the age or maturity of the child.²⁴⁹ However, the role of the legal representative is dependent on child related factors.²⁵⁰ If the child is sufficiently mature, developed and wishes to participate directly then the legal representative takes instructions from the child. If not, when the child is very young and unable to give instructions, the legal representative may still be appointed but will function as a best interests advocate. In the Netherlands, the general guardian ad litem is appointed at the discretion of the court and the child's age or level of maturity can be a factor taken into account, in addition to other relevant circumstances, but is not a hard requirement.²⁵¹

Secondly, child related factors can be a requirement which must be achieved, a **'positive' requirement** in the sense of children having capability or having reached a certain age. The simplest version hereof, is that children must be 12 years or older to have a separate legal representative in the Dutch care and supervision order proceedings.²⁵² For the appointment of a filiation guardian ad litem to initiate parentage proceedings on behalf of the child the Dutch courts have discretion. A general requirement often applied by the Dutch courts is that the child should be sufficiently able to foresee the consequences of the application in the parentage proceedings.²⁵³ In New South Wales, Australia, the direct legal representative will be appointed if the child is presumed capable of giving proper instructions²⁵⁴ and in France, if the child is *'capable de discernement*' a children's lawyer can represent the child.²⁵⁵

The last option is that child related factors are used as a **'negative' requirement**, meaning that the child is considered incapable or too young and thus requires a representative. This is the case for both the New South Wales' independent legal representative and the French ad hoc administrator

 $^{^{248}}$ S. 6(a) AFLA 1975. *Re K* [1994] FLC 92-461; Inquiry into the Australian Legal Aid System 1997, para. 5.16; Carson et al. 2014, p. 59; and Kaspiew et al. 2013, p. xi. See sections 5.1.1.4 and 5.1.1.5 of the annexes.

²⁴⁹ See section 5.4.1.4 of the annexes.

²⁵⁰ See Boezaart 2013a, p. 369; Cleophas & Assim 2015, p. 300; and section 5.4.2 of the annexes.

²⁵¹ See Jansen 2016b; and section 5.3.1.4 of the annexes.

²⁵² Art. 1:261, 1:262b, 1:264, 1:265d and 1:265g DCC. See section 5.3.1.4 of the annexes.

²⁵³ See Schrama 2015; and section 5.3.1.4 of the annexes.

²⁵⁴ S. 99A(1)(a) and 99A(2)(a) CCPA (NSW) and S. 122(3)(c)(i) Adoption Act 2000. See section 5.1.1.4 of the annexes.

²⁵⁵ Art. 388-1 FCC and Art. 1182 FCCP. See section 5.2.1.4 of the annexes.

who are both appointed when the requirements for, respectively, the direct legal representative or children's lawyer are not met. If the child is not presumed capable of giving proper instructions then an independent legal representative will be appointed.²⁵⁶ In France, if the child is not '*capable de discernement*' then an ad hoc administrator can be appointed for the child.²⁵⁷ Lastly, in New South Wales, Australia a guardian ad litem can be appointed if there are special circumstances which warrant the appointment, these special circumstances include when the child has special needs because of their age, disability or illness, and in child protection procedures, when the child is not capable of giving proper instructions.²⁵⁸

In Table 5 below, the three ways in which child related factors can play a role as a requirement discussed above have been presented per jurisdiction. The table shows that while in the Netherlands and South Africa the task of the representatives has no impact on the requirements applied, in Australia – specifically, New South Wales – and France there is a difference. The separate legal representatives, the direct legal representative and the children's lawyer, require capability (as defined in italics), while the best interests representative, the independent legal representative and the ad hoc administrator, require incapability. Thus, in these two jurisdictions more developed children who are capable of giving instructions or expressing their views are granted the chance to instruct their representative, while the (generally younger) children who are incapable of doing so are not.

	Australia	France	The Netherlands	South Africa
No (explicit) requirement	ICL	-	GGAL	LR
				CAL
'Positive' requirement	DLR	CL	FGAL	-
			SLR	
+↑ -↓	Capability of giving proper instructions	Capable de discernement	-	-
'Negative' requirement	ILR	AHA	-	-
	GAL			

Table 5: The role of child related requirements for the representation forms in the different jurisdictions*

* The different forms of representation have been shaded according to their general task: Light gray = best interests representatives Abbreviations:

AHA: Ad hoc administrator CAL: Curator ad litem CL: Children's lawyer DLR: Direct legal representative

FGAL: Filiation guardian ad litem GAL: Guardian ad litem GGAL: General guardian ad litem ICL: Independent children's lawyer ILR: Independent legal representative LR: Legal representative SLR: Separate legal representative

Open requirements for representation

Three of the child related requirements discussed above are open requirements, leaving a certain discretion to judges. All three of these are linked to capability: the New South Wales' requirement of

²⁵⁶ S. 99A(1)(a) and 99A(2)(a) CCPA (NSW) and S. 122(3)(c)(i) Adoption Act 2000. See section 5.1.1.4 of the annexes.

²⁵⁷ Art. 388-1 FCC and Art. 1182 FCCP. See section 5.2.1.4 of the annexes.

²⁵⁸ S. 123(2) Adoption Act 2000 and s. 100(2) CCPA (NSW).

being 'capable of giving proper instructions', the French requirement of being '*capable de discernement*' and the Dutch requirement of 'being sufficiently able to foresee the consequences of a parentage application'. What is meant by these requirements?

In New South Wales, the requirement of being capable of giving proper instructions is supplemented by rebuttable presumptions in the legislation and guidelines in the Representation Principles for Children's Lawyers. A child is presumed capable from the age of 12 onwards in child protection proceedings and from the age of 10 onwards in adoption proceedings.²⁵⁹ The capability must be determined per child, by considering their willingness to participate, ability to communicate, age, level of education, cultural context and degree of language acquisition, and not by assessing the child's 'good judgment' or level of maturity.²⁶⁰

The French requirement of being '*capable de discernement*' is taken directly from the French version of Article 12 of the UNCRC. In France, there is no age requirement applied to the determination of the child's capability to express their views. The judge must determine the child's capability individually in each separate case. Generally, however, children are considered capable from the age of 7 onwards.²⁶¹

In the Netherlands, the requirement of being sufficiently able to foresee the consequences of an application only concerns limited cases, when a filiation guardian ad litem is requested to initiate proceedings on behalf of the child. In these situations, generally children from the age of 12 onwards are considered sufficiently capable, although in exceptional cases the requirement has been circumvented to allow for the appointment of a guardian ad litem for (very) young children, even if they are not sufficiently able to foresee the consequences.²⁶²

Evaluation of requirements in light of Article 12 UNCRC

Article 12 UNCRC refers to child related factors twice: the capability of the child and the capacity of the child. Regarding when children should be provided with a means to participate, only the capability of the child is relevant. The issue of capacity only concerns that weight should be given to the child's views 'in accordance with the age and maturity of the child'. This is for the decision maker to decide after hearing the views of the child. On the other hand, the right to be heard should be granted to the child 'capable of forming his or her own views', according to Article 12 UNCRC. So how do the child related requirements in the four jurisdictions relate to the capability requirements contained in the UNCRC?

Firstly, it is crucial that there should not be an age limit in law or in practice.²⁶³ Age should not be an automatic barrier.²⁶⁴ Of the representation forms discussed, only one has a simple age limit

²⁵⁹ S. 99A(1)(a) and 99A(2)(a) CCPA (NSW) and S. 122(3)(c)(i) Adoption Act 2000. See section 5.1.1.4 of the annexes.

²⁶⁰ Principles C1 and C2 of the Representation Principles for Children's Lawyers 2014. See section 5.1.1.4 of the annexes.

²⁶¹ See Attias 2012; Rongé 2008, p. 45; and section 5.2.1.4 of the annexes.

²⁶² See Van Teeffelen 2008; Schrama 2015; and section 5.3.1.4 of the annexes.

²⁶³ UNCRC General Comment No. 12 2009, para. 21.

in law – the Dutch separate legal representative in care and supervision order proceedings. This runs contrary to Article 12 UNCRC. Of course, in other jurisdictions age limits are also seen applied in practice as a means by which to elucidate open requirements. For example, in New South Wales the rebuttable presumptions in legislation consist of an age limit. Compared to the age limit for the Dutch separate legal representative, this is not a 'barrier' as it is a presumption which must be tested per child in combination with other factors, e.g. the child's ability to communicate, and can therefore be debunked.

A second important aspect is that decision makers, e.g. the courts, should start with the presumption of a capable child and then determine on a case-by-case basis whether the child is not capable.²⁶⁵ Whether this occurs can, generally, not be found specifically in the requirements themselves, as it is a mindset that must be adopted by judges. However, it would be good if the necessity of such a mindset was made more explicit.

The final issue regarding capability, is how it should be understood. As discussed in section 2.2.1, how the UNCRC understands capacity remains unclear. According to the Committee, the child need only have a 'sufficient understanding', but what this means exactly is vague.²⁶⁶ The same can be said about the French requirement of being '*capable de discernement*' and the Dutch requirement of being sufficiently able to foresee the consequences. Both remain vague. In fact, the Committee has criticized the French requirement, stating that in practice the interpretation and determination of this requirement may possibly discriminate and deny the child of their right to be represented and heard.²⁶⁷ It is ironic that the Committee does so, when not providing for any clarity themselves. The manner in which New South Wales has elaborated their open requirements of being capable of giving proper instructions is commendable. Although the child's ability to communicate and their willingness to participate may be open to discussion, clarifying specific factors creates legal equality between children in the assessment of their capability.

3.3.2. Conflict related requirements

In addition to certain specific child related factors, in some jurisdictions other requirements must also be met for the appointment of a representative for the child. These can be classified as conflict related requirements, as the specifics of the conflict make it so that the child requires representation. The additional requirements found in all four jurisdictions are similar, in the sense that they can be classified as one of the two conflict related requirements.

²⁶⁴ UNCRC Recommendations: Day of General Discussion 2006, para. 51.

²⁶⁵ UNCRC General Comment No. 12 2009, para. 20. See also, Krappmann 2010, p. 507; Lansdown 2011, p. 20; and section 2.2.1.

²⁶⁶ UNCRC General Comment No. 12 2009, para. 21.

²⁶⁷ UNCRC Concluding Observations France 2004, para. 21. s

Necessary in the child's best interests

The first shared conflict related requirement is that representation is necessary in the child's best interests. This requirement is applied to all the Australian representation forms. With regards to the independent children's lawyer, the independent legal representative, and the direct legal representative the non-exhaustive criteria provided for in *Re K* guide the courts in deciding whether the child's interests require representation.²⁶⁸ The courts can appoint a representative, according to these criteria, for example if there are allegations of abuse, issues of parental alienation or significant medial, psychiatric or psychological illnesses or disorders involved. In international child abduction proceedings these criteria are supplemented by the extra requirement of 'exceptional circumstances'.²⁶⁹ The New South Wales' guardian ad litem also requires special circumstances, namely that the court is of the opinion that the child will benefit from the appointment.²⁷⁰ In the Netherlands, the appointment of the general guardian ad litem is also subject to the requirement that it is necessary in the best interests of the child.²⁷¹ This is also the case for the South African legal representative.²⁷² The appointment of a South African curator ad litem is also solely guided by the best interests of the child, although the additional requirement is that the conflict must be one of four specific instances.²⁷³ One of these falls under the second shared conflict related requirement discussed in the following, the other three are that the child does not have a parent or guardian, that the parent or guardian cannot be found or that the parent or guardian unreasonably refuses to assist the minor in the legal proceedings.

The requirement that representation should be in the child's best interests for the appointment to occur, ties into the symbiotic relationship between Article 3 and Article 12 UNCRC.²⁷⁴ While it is positive that the best interests of the child guide the appointment of a representative as a requirement in the situations above, it should not be too high of a hurdle in practice. States should not use the best interests principle to withhold children the opportunity to express their views or to trump their right to be heard.²⁷⁵ This is especially so, because an adult's – subjective – assumption of what is in the best interests of the child can impede the child's opportunity to participate in proceedings.²⁷⁶ The requirement of representation being necessary in the child's best interests should therefore be used cautiously. The courts in Australia, the Netherlands and South Africa should not abuse the requirement to prevent the representation of children.

²⁶⁸ S. 68L(2) AFLA 1975, S. 99(1) CCPA (NSW), S. 122(2)(b) Adoption Act 2000, and *Re K* [1994] FLC 92-461. See also, UNCRC Initial Periodic Report of Australia 1996, para. 248; and section 5.1.1.5 of the annexes.

⁹ S. 68L(3) AFLA 1975. See section 5.1.1.5 of the annexes.

²⁷⁰ S. 100(1)(B) CCPA (NSW) and S. 123(1)(B) Adoption Act 2000.

²⁷¹ See section 5.3.1.5 of the annexes. ²⁷² S. 6(4) Divorce Act 1979 and S. 55 SACA 2005. See section 5.4.1.5 of the annexes.

²⁷³ See Ter Haar 2016; and section 5.4.1.5 of the annexes.

²⁷⁴ UNCRC General Comment No. 12 2009, para. 74.

²⁷⁵ Hodgkin & Newell 2007; Lansdown 2011, p. 33; and section 2.3.2.

²⁷⁶ See Archard & Skivenes 2009, p. 2; Raitt 2004, p. 153; Thomas & O'Kane 1998, p. 151; and section 2.3.2.

Conflict of interests

The second shared conflict related requirement is the requirement of a conflict of interests between the child and the parent(s) or guardian(s). This requirement is found explicitly in France for the ad hoc administrator and for the Dutch general guardian ad litem. In France, the conflict of interests must actually exist for the appointment according to Article 383 FCC. For the appointment on the basis of Article 388-2 FCC, it is sufficient that it appears to exist or is highly likely to arise. While the requirement of a conflict of interests is relatively subjective and vague, in France the conflict must be between the child and the parent(s) and must be sufficiently contradictory or divergent. A conflict between the parents is not enough for the appointment of an ad hoc administrator.²⁷⁷ In the Netherlands, the requirement of a conflict of interests is understood broadly. It includes both conflicts between the child and the parent(s) or conflicts between the two parents.²⁷⁸ The requirement of a conflict of interests also plays a role in Australia and South Africa, albeit more limited. In Australia, the *Re K* criteria include the situation of an intractable conflict between the parents involving the child, and in South Africa, one of the instances in which the curator ad litem can be appointed is when the interests of the child clash with those of the parents or guardians or if the possibility of a clash exists.²⁷⁹

The requirement of a conflict of interests between the child and the parent(s) or between the two parents is a logical one. The Committee itself has emphasized that there is often a risk in matters concerning children that there is a conflict between the child and their parent(s) or between the two parents themselves.²⁸⁰ Therefore, in those situations it is vital to appoint a representative, who is not a family member, for the child.

3.3.3. Combined requirements for representation

Other than the three specific situations in which the appointment of a representative is mandatory, all the other representation forms in each of the four jurisdictions are subject to either child related or conflict related requirements or a combination of both as shown in Table 6.

²⁷⁷ See Fédération nationale des administrateurs ad hoc 2009, p. 23-24; and section 5.2.1.5 of the annexes.

²⁷⁸ Werkproces benoeming bijzondere curator o.g.v. Art. 1:250 BW 2014, p. 3. See also, Jansen 2016a, p. 2179; Ter Haar 2015; and section 5.3.1.5 of the annexes. ²⁷⁹ See *Re K* [1994] FLC 92-461; Boezaart & de Bruin 2011, p. 422-423; Boezaart 2013b, p. 709-710; and sections 5.1.1.5

²¹⁹ See *Re K* [1994] FLC 92-461; Boezaart & de Bruin 2011, p. 422-423; Boezaart 2013b, p. 709-710; and sections 5.1.1.5 and 5.4.1.5 of the annexes.

²⁸⁰ UNCRC General Comment No. 12 2009, para. 36.

Table 6: Requirements	for the forms	of representation pe	r jurisdiction
-----------------------	---------------	----------------------	----------------

			Child	l related requ	irements	Conflict related requirements		
re	Forms of presenta		'Positive'	'Negative'	Not explicit	t Necessary in Conflict of N child's interests interests		
	ICL				Х	Х	(x)	
AU		ILR		Х		Х	(x)	
	NSW	DLR	Х			Х	(x)	
		GAL		Х		Х		
FR	AHA*			Х			Х	
	CL		Х					Х
	GGAL				Х	Х	Х	
NL	FGAL*	¢	Х					Х
	SLR*		Х					Х
SA	LR				Х	Х		
	CAL				Х	Х	(x)	

(x): Is a factor but not a hard requirement.

* In three situations the appointment is mandatory: the AHA when requested in Art. 383 FCC; the FGAL in pending parentage proceedings; and the SLR in the authorization of secure youth care proceedings.

Abbreviations:

AHA: Ad hoc administrator CAL: Curator ad litem

CL: Children's lawyer

DLR: Direct legal representative

FGAL: Filiation guardian ad litem GAL: Guardian ad litem GGAL: General guardian ad litem ICL: Independent children's lawyer ILR: Independent legal representative LR: Legal representative SLR: Separate legal representative

3.4. Appointment of a child representative

Knowing which forms of representation are available, in which types of proceedings they are available and what the applicable requirements are in those situations must be accompanied by an understanding of who makes the decision to appoint a child representative and at which point in the proceedings this decision can be made. This section covers these two aspects, first discussing who can request and/or appoint a child representative and then discussing when the appointment can occur and when the task of the representative ends.

3.4.1. Who appoints the child representative

In all four jurisdictions, it is either the court or the child who can opt to appoint a representative, with the court capable of doing so either *ex officio* or on request. An overview hereof is included in Table $7.^{281}$

	Forms of		Cou	ırt			
re	representation		Ex officio	At the request of:	Child		
AU	ICL		ICL X		X	X: child, welfare organization, any other person	
		ILR	Х	X?	\mathbf{X}^{1}		
	NSW	DLR	Х	X?	X ¹		
		GAL	Х	X?			
FR	AHA		X	X: mandatory at the request of a legal representative, discretion at the request of the child or public prosecutor			
	CL				Х		
	GGAL		Х	X: child, legal representative, person with 'family life', guardianship agency, or other interested party			
NL	FGAL		X: mandatory when pending	X: child, legal representative, legal parent without parental authority, natural father with 'family life'			
	SLR				X: mandatory in authorization of secure youth care proceedings		Х
SA	LR		Х		X^2		
	CAL		Х	X: child, child's relative, other persons with an interest			

Table 7: Decision maker for each form of representation per jurisdiction

?: Unclear whether, but likely, possible at request and unclear that if so, at whose request.

¹: Requires leave of court.

²: Requires consent from the Legal Aid Board and in divorce, care or maintenance cases written consent from the Regional Operations Executive.

<u>Abbreviations:</u> AHA: Ad hoc administrator CAL: Curator ad litem

CL: Children's lawyer

DLR: Direct legal representative

FGAL: Filiation guardian ad litem GAL: Guardian ad litem GGAL: General guardian ad litem ICL: Independent children's lawyer ILR: Independent legal representative LR: Legal representative SLR: Separate legal representative

²⁸¹ For further information see sections 5.1.1.6 (AU), 5.2.1.6 (FR), 5.3.1.6 (NL), and 5.4.1.6 (SA) of the annexes.

The table shows that the decision is mostly in the hands of the courts, with the Australian and South African exceptions requiring the leave of court or consent from the Legal Aid Board. Children are only free to appoint the most traditional lawyer forms – the French children's lawyer and the Dutch separate legal representative – themselves without requiring any further consent. From the table it can also be concluded that when the representative can be requested, the child is always offered the opportunity to request the court appointment of a representative.

Evaluation in light of Article 12 UNCRC

Neither Article 12 UNCRC nor the General Comments made by the Committee, refer to whom may or should request and/or appoint a representative for the child. It appears to be a decision left to the State parties and perhaps falls under the requirement that the child's right to be heard in judicial proceedings should be provided 'in a manner consistent with the procedural rules of national law'. What has been said by the Committee on this topic, however, related to the French situation in which the hearing and representation of a child through the children's lawyer is subject to the child's own request. The Committee was concerned, saying that this 'may give rise to discrimination and inconsistencies in practice'.²⁸² This concern can equally apply to the Dutch separate legal representative. Therefore, it seems that having the court decide whether to appoint a representative, with the opportunity for the child to request the appointment, is the most appropriate framework. Even though, according to Milne, this only grants children a limited amount of independence as they remain dependent on the court as to whether they can participate.²⁸³

3.4.2. Duration of child representation

The timing of child representation in family law proceedings lacks a clear structure in most jurisdictions.²⁸⁴ In **South Africa** it is unclear from what moment the legal representative or curator ad litem can be appointed and when their function ends. In practice, they are generally appointed at the start of proceedings, although it can be earlier as their role can be to assist the child in getting access to court.²⁸⁵ The start and end of the **French** children's lawyer's function is also unclear, more clarity exists with regards to the ad hoc administrator.²⁸⁶ The appointment of an ad hoc administrator on the ground of Article 388-2 FCC can only occur in an ongoing procedure, while the appointment on the ad hoc administrator is limited and the completion of the mission means the end of their appointment.²⁸⁷

²⁸² UNCRC Concluding Observations France 2009, para. 39.

²⁸³ Milne 2015, p. 18. See also section 2.1.

²⁸⁴ Taylor et al. 2012, p. 669.

²⁸⁵ S. 14 SACA 2005. See section 5.4.1.3 of the annexes.

²⁸⁶ See section 5.2.1.3 of the annexes.

²⁸⁷ See Fédération nationale des administrateurs ad hoc 2009, p. 10-11; Lebrun 2001, p. 33; and section 5.2.1.3 of the annexes.

The **Dutch** guardians ad litem also have limited tasks, making the duration of their task dependent on the assignment description given by the court.²⁸⁸ The general guardian ad litem can be appointed at any phase of the procedure, including prior to the start of proceedings in theory.²⁸⁹ The filiation guardian ad litem will always be involved from the very start of proceedings, either being appointed to initiate proceedings on behalf of the child or being immediately appointed once proceedings have been initiated by another party. The Dutch separate legal representative will also be immediately appointed at the start of authorization for secure youth care proceedings, but can become involved at any point in other proceedings as children are free to decide.²⁹⁰

In Australia, the timing of the representatives is most clearly defined. The independent children's lawyer, independent legal representative, direct legal representative and the guardian ad litem can all be appointed in the early phases of proceedings.²⁹¹ The appointment of an independent children's lawyer ends when a final decision has been made or when they seek the court to discharge them of their appointment.²⁹² Both the independent legal representative and direct legal representative can be dismissed by the courts at any time for any reason and the direct legal representative can also be dismissed by the child personally.²⁹³ The guardian ad litem's function will generally end after the court's decision, although they may commence an appeal on behalf of the child.²⁹⁴

All in all, the child representation forms in all four jurisdictions generally function from the start of family law proceedings until a decision has been made or they have been discharged by the court or due to the completion of their mission. Whether this complies with Article 12 UNCRC is difficult to say, as it is a subject which is not discussed by the Article or by the Committee. However, having representation available from the start of proceedings but also allowing for the appointment of a representative at a later point in time makes representation more accessible than a more limited frame within which a representative could be appointed. This accessibility supports the 'opportunity' for the child to be heard in judicial proceedings as required by Article 12(2) UNCRC.²⁹⁵

²⁸⁸ See De Graaf & Limbeek 2011; and section 5.3.1.3 of the annexes.

²⁸⁹ Kamerstukken II 2004/05, 301245, 3, p. 13; Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 5.

²⁹⁰ Art. 6.1.10(4) Youth Care Act. See section 5.3.1.3 of the annexes.

²⁹¹ See Care and Protection Practice Standards 2015, principles 2.2.2(1), 2.2.4, 2.3.2(1), and 2.3.4; Kaspiew et al. 2013, p. 18; and section 5.1.1.3 of the annexes. 292 Constant

Guidelines for Independent Children's Lawyers 2013, para. 6.6.

²⁹³ See the explanatory text relating to Principle J2 of the Representation Principles for Children's Lawyers 2014; and section 5.1.1.3 of the annexes.

²⁹⁴ Bao-Er 2006, p. 5.

²⁹⁵ See UNCRC General Comment No. 12 2009, para. 34; Lansdown 2011, p. 52; and section 2.2.2.

3.5. The function, work and payment of the child's representative

The general task of child representatives is important, but cannot be fully understood without knowing which function requirements are attached, how their task ought to be completed and who bears the costs of their work. This section will delve into each of these three aspects separately.

3.5.1. Function requirements

For the function of a child representative in each of the four jurisdictions a variety of requirements are applied. The function can be fulfilled by different types of practitioners or even by lay persons. Although it might be expected that the function of all forms of child representatives are fulfilled by a legal practitioner, this is not always the case.

In Table 8, the required professional background of each form of representation and whether they must be registered on some form of list or as a member of a panel is presented.

				Type of practitioner					
re	Forms of representation		Legal practitioner	Social or behavioral science practitioner	Other	Register			
	ICL		Х			Х			
AU		ILR	Х			Х			
	NSW	DLR	Х			Х			
		GAL		Х		Х			
FR	AHA		X	X	X: family member or person close to the child	Х			
	CL		Х			(X: voluntary)			
	GGAL		Х	Х		Х			
NL	FGAL		Х			Х			
	SLR		Х						
SA	LR		Х						
	CAL		Х						

Table 8: Function requirements of which type of practitioners and their registration in relation to the forms of representation

Abbreviations: AHA: Ad hoc administrator CAL: Curator ad litem CL: Children's lawyer DLR: Direct legal representative

FGAL: Filiation guardian ad litem GAL: Guardian ad litem GGAL: General guardian ad litem ICL: Independent children's lawyer ILR: Independent legal representative LR: Legal representative SLR: Separate legal representative

The table evidently shows that most forms of representation require a legal practitioner.²⁹⁶ There are three exceptions. The guardian ad litem in New South Wales cannot be a legal practitioner, but must be a person with the necessary qualification and experience in social, health, or behavioral sciences.²⁹⁷ Both the French ad hoc administrator and the Dutch general guardian ad litem can either be a legal practitioner or another person. The ad hoc administrator will generally be a family member or other person close to the child, unless the judge finds this not to be in the interests of the child, in which

²⁹⁶ For the extended information see sections 5.1.2.1 (AU), 5.2.2.1 (FR), 5.3.2.1 (NL), and 5.4.2.1 (SA) in the annexes.

²⁹⁷ See section 5.1.2.1 of the annexes.

case a natural or legal person will be appointed from a special list provided for by law.²⁹⁸ The natural or legal person can have a legal, social or behavioral sciences background.²⁹⁹ In the Netherlands in theory, anyone can be appointed as a general guardian ad litem. However, the list of guardians ad litem from which the court chooses, includes both legal practitioners, mediators and experts from the social and behavioral sciences.³⁰⁰

With regard to the (professional) background of child representatives, the Committee on the Rights of the Child has not placed any restrictions. Instead it has stated that parents, lawyers or other persons are suitable, although parents less so due to the risk of conflict with – the views of – their child. Thus, the options of legal practitioners and social or behavioral practitioners are both suitable. However, the fact that in South Africa, both the legal representative and the guardian ad litem can only be legal practitioners and that the Constitution specifically refers to legal practitioners, has been deemed contradictory to the UNCRC by Sloth-Nielsen.³⁰¹ She argues that this limits what was originally meant by the drafters of the Convention. This may be the case, although it is also rare that other professionals are appointed in the other jurisdictions and the Committee has, as of yet, not criticized this status quo. That the French ad hoc administrator may be a parent, unless a judge finds this ill-advised, in which case another professional will be appointed, does appear to fit well within the guidance of the Committee.

Registers of child representatives

Table 8 also shows that most representatives are appointed from a register. More specifically, various types of registers exist. In New South Wales, Australia the independent children's lawyer, direct legal representative and independent legal representative are all either 'inhouse' practitioners or on the panel for appointment.³⁰² Guardians ad litem must also be chosen from the panel to which they can be appointed. In France, the ad hoc administrator with a professional nature can only be appointed from the list created per Article R. 53 French Code of Criminal Procedure (Code de procédure pénale; hereafter FCPC).³⁰³ A children's lawyer can voluntarily join the regional bar group of children's lawyers provided for by the national charter on the role and functioning of the children's lawyer of 2008.³⁰⁴ For the Dutch guardians ad litem there are also two lists composed by the Legal Aid Board to which professionals can be added.³⁰⁵ In South Africa, both forms of representative are not registered.

²⁹⁸ Art. R. 53 FCPC. See section 5.2.2.1 of the annexes.

 $^{^{299}}$ See section 5.2.2.1 of the annexes.

 $^{^{300}}$ See section 5.3.2.1 of the annexes.

 ³⁰¹ Sloth-Nielsen 2008, p. 498.
 ³⁰² See Kaspiew et al. 2013, p. 15; Ross 2013b, p. 335; and section 5.1.2.1 of the annexes.

³⁰³ See section 5.2.2.1 of the annexes.

³⁰⁴ Chartre nationale de la defense des mineurs 2008. See section 5.2.2.1 of the annexes.

³⁰⁵ See Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 6; Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 3; and section 5.3.2.1 of the annexes.

Although a curator ad litem will generally be an advocate of the High Court, an attorney can also fulfill the function.³⁰⁶

The existence of panels or registers of child representatives is positive, as it provides for experienced and skilled representatives. In addition, once appointed to the register or panel it is often the case that the representatives must adhere to certain codes of conduct. For example, in New South Wales the representatives must adhere to the practice standards once appointed to the panel and in France many children's lawyers have voluntarily bound themselves to best practice guidelines.³⁰⁷ This follows the Committees recommendation that codes of conduct should be developed for child representatives in accordance with Article 12 UNCRC.³⁰⁸

Required experience and skills of a child representative

For most forms of representation certain experience and skills are required. The most common being that the representative must be able to communicate effectively with the child and other parties,³⁰⁹ must have knowledge of and experience with the issues facing the child, e.g. parental conflicts,³¹⁰ must have knowledge of and experience with the law,³¹¹ and should have mediation and advocacy skills.³¹² In Australia, an independent children's lawyer must also complete a training program and can follow a professional development program.³¹³ These required skills tie in perfectly with what the Committee has said about the topic. Namely, that the representative should have sufficient knowledge and understanding of the proceedings and have experience in communicating and working with children.³¹⁴

3.5.2. The work of the child representatives

Knowing who can function as a child representative is important, knowing how they ought to complete their task as child representative even more so. As was previously discussed in section 3.1, the forms of representation can be categorized by their general task of either representing the child's best interests or representing the child's views as directed. But how ought the general task of these forms of representation to be completed in the four jurisdictions?

³⁰⁶ See Boezaart 2013b, p. 713-714; and section 5.4.2.1 of the annexes.

³⁰⁷ Kaspiew et al. 2013, p. 103; and Chartre nationale de la defense des mineurs 2008. See sections 5.1.2.1 and 5.2.2.1 of the annexes.

³⁰⁸ UNCRC General Comment No. 12 2009, para. 37.

³⁰⁹ See, for example, the Australian ICL, ILR, DLR and GAL in section 5.1.2.1, the Dutch GGAL in section 5.3.2.1, and the South African LR in section 5.4.2.1 of the annexes. ³¹⁰ See, for example, the Australian ICL, ILR, DLR and GAL in section 5.1.2.1, the French AHA in section 5.2.2.1, and the

³¹⁰ See, for example, the Australian ICL, ILR, DLR and GAL in section 5.1.2.1, the French AHA in section 5.2.2.1, and the Dutch GGAL in in section 5.3.2.1 of the annexes.

³¹¹ See, for example, the Australian GAL in section 5.1.2.1, the Dutch FGAL in in section 5.3.2.1, and the South African LR in section 5.4.2.1 of the annexes.

³¹² See, for example, the Australian GAL in section 5.1.2.1, the Dutch GGAL and FGAL in in section 5.3.2.1, and the South African LR in section 5.4.2.1 of the annexes.

³¹³ See Carson et al. 2014, p. 60; Kaspiew et al. 2013, p. 107; and section 5.1.2.1 of the annexes.

³¹⁴ UNCRC General Comment No. 12 2009, para. 36.

An important part of the answer to this question concerns how the representatives ought to **communicate with the child**. All eleven forms of representative are advised to speak to the child in person and must inform the court of the child's views.³¹⁵ For some forms of representation, additional requirements are in place for hearing the child. For example, the Dutch general guardian ad litem is advised to speak to children from the age of 12 onwards themselves, but to involve a psychologist when speaking to younger children.³¹⁶ Or, in Australia, the direct legal representative and independent legal representative, must use language appropriate for the specific child, employing an interpreter if necessary. This means that (potentially) vulnerable or marginalized children are also considered, a positive detail considering the right to non-discrimination in Article 2 UNCRC and its relationship to Article 12 UNCRC.³¹⁷

In addition to hearing the child's views, most forms of representation are required to **inform the child** on their task, the course of the proceedings, and the decision made by the court.³¹⁸ Again some restrictions may apply. For example, in France children should only be informed of the final decision once they have reached the age of six and should not be informed of the specific contents of the files in child protection proceedings by the children's lawyer.³¹⁹ Some forms of representation – the Australian independent children's lawyer, independent legal representative and direct legal representative, the French children's lawyer, the Dutch separate legal representative and the South African legal representative – specifically ought to have continued contact with the child throughout the proceedings.³²⁰ This means that the child will have a sort of 'lawyer-client' relationship with the representative, can contact them with questions and will be informed of all steps taken.³²¹

Aside from contact with the child, most forms of representatives will also be required to **obtain and review further relevant information or evidence**. One aspect hereof is that the representative may need to speak to other involved parties or persons about the conflict or the situation of the child. This may, for example, be included in the instructions for the Dutch guardians ad litem.³²² The representative may also be required to draw certain evidence to the courts attention, e.g. concerning situations of domestic abuse.³²³

³¹⁵ See sections 5.1.2.2 (AU), 5.2.2.2 (FR), 5.3.2.2 (NL), and 5.4.2.2 (SA) of the annexes.

³¹⁶ See Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 10; and section 5.3.2.2 of the annexes.

³¹⁷ UNCRC General Comment No. 12 2009, para. 77 and 78. See section 2.3.1.

³¹⁸ See sections 5.1.2.2 (AU), 5.2.2.2 (FR), 5.3.2.2 (NL), and 5.4.2.2 (SA) of the annexes.

³¹⁹ Art. 1187 FCCP. See section 5.2.2.2 of the annexes.

³²⁰ See sections 5.1.2.2 (AU), 5.2.2.2 (FR), 5.3.2.2 (NL), and 5.4.2.2 (SA) of the annexes.

³²¹ See e.g. the DLR in New South Wales, Australia, Principle D2 of the Representation Principles for Children's Lawyers 2014; Care and Protection Practice Standards 2015, para. 2.2.3; and section 5.1.2.2 of the annexes.

³²² See Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 7; and section 5.3.2.2 of the annexes.

³²³ See e.g. the ICL in Australia, Guidelines for Independent Children's Lawyers 2013, para. 7; and section 5.1.2.2 of the annexes.

Evaluation in light of the implementation steps for Article 12 UNCRC

The above corresponds with three of the Committee on the Rights of the Child's five steps to 'effectively realize' the child's right to be heard in proceedings.³²⁴ The first step, the **preparation** of children through informing them of the available options, the consequences thereof and the impact of their views on the proceedings, ties in to the obligation of representatives to inform the child at the start of the proceedings.³²⁵ The second step, hearing the child, is clearly provided for as all eleven representation forms must speak to the child in person to obtain their views. The Committee requires that this must be done in an 'enabling and encouraging' context, in which the representative seriously listens to and talks to the child.³²⁶ Taking the child seriously, by having a continued dialogue, is provided for by the forms of representation mentioned above who must have continued contact with the child throughout the proceedings. Another good example are the Australian representatives, who ought to 'strive for a relationship of trust and respect' and should, when speaking to the child, encourage them to ask further questions and answer these appropriately.³²⁷ Lastly, the fourth step, giving **feedback** to the child regarding how their views have been taken into account, is also covered by the requirement that the representatives inform the child about the decision of the court.³²⁸

The third and fifth steps of the Committee's five steps are not covered by the work obligations of representatives above. The third step, the assessment of the capacity of the child, is not included, because it is generally the decision-maker, e.g. the court, who must determine the capacity of the child and the weight to be accorded to their views (see also section Error! Reference source not found. above).³²⁹ Only in South Africa, do legal representatives themselves need to determine the capacity of the child. As discussed in section 3.3.1 above, this is because their role is dependent on whether the child is sufficiently mature, developed and wishes to participate. If this is the case, then the legal representative is a client-directed representative, if not then a best interests representative. The legal representative needs to determine the child's capacity at the outset of the proceedings on a case-by-case basis, thereby conforming to the step provided by the Committee.³³⁰ Whether children are provided access to complaints, remedies and redress procedures if their right to be heard has been breached, the fifth step, has not been explored in this research.³³¹

³²⁴ UNCRC General Comment No. 12 2009, para. 40. See section 2.2.3.

³²⁵ UNCRC General Comment No. 12 2009, para. 41.

³²⁶ UNCRC General Comment No. 12 2009, para. 42.

³²⁷ See Guidelines for Independent Children's Lawyers 2013, para. 6.6; Care and Protection Practice Standards 2015, para. 2.2.3 and 2.3.3; and section 5.1.2.2 of the annexes. ³²⁸ UNCRC General Comment No. 12 2009, para. 45.

³²⁹ UNCRC General Comment No. 12 2009, para. 44.

³³⁰ See Boezaart 2013a, p. 369; Cleophas & Assim 2015, p. 300; and section 5.4.2 of the annexes.

³³¹ UNCRC General Comment No. 12 2009, para. 46 and 47. See section 2.2.3.

Additional specific duties of representatives

The type of general task determines some additional specific duties of the representatives. These will be discussed per type of general task.

Specific duties for the separate legal representatives

With regards to representation forms functioning as a separate legal representative – the Australian direct legal representative, French children's lawyer, Dutch separate legal representative and South African legal representative – it is logical that they must seek the child's instructions. How? By speaking to the child directly. In Australia, it is also emphasized that the direct legal representative should ensure that the child understands the instructions they are giving.³³² In South Africa, legal representatives also ought to apply their legal knowledge and expertise to 'translate' the child's views to legal jargon.³³³ The general task of a separate legal representative also calls for these forms of representation to appear in court and function either directly as a lawyer in the proceedings or to accompany the child and assist them in court.³³⁴ Only the South African legal representative does not function as a lawyer in court, but will make submissions to the court.³³⁵

Specific duties for best interests representatives

On the other hand, the forms of representation functioning as a best interests representative do not generally function as a lawyer in court (except the independent legal representative in Australia). Instead, the Dutch guardians ad litem and the South African curator ad litem are required to submit a written report to the court, the Australian independent children's lawyer makes oral submissions to the court, and the Australian guardian ad litem, French ad hoc administrator and South African curator ad litem instruct a lawyer to act in accordance with their instructions.³³⁶

The most interesting question is how these representatives determine what they ought to represent, namely how do they determine what is in the best interests of the child? A few common standards apply. The first standard is that the representatives must work independently, acting impartially from the court and other parties.³³⁷ Exclusively representing the interests of the child is in line with the Committee's understanding of Article 12 UNCRC.³³⁸ Secondly, the best interests must be determined objectively, taking into account the available information and the child's views but not the representatives 'own opinion'.³³⁹ The latter is important for compliance with Article 12 UNCRC.

³³² Principle D2 of the Representation Principles for Children's Lawyers 2014 and Care and Protection Practice Standards 2015, para. 2.2.3. See section 5.1.2.2 of the annexes.

³³³ See Sloth-Nielsen 2008, p. 504; and section 5.4.2.2 of the annexes.

³³⁴ See sections 5.1.2.2 (AU), 5.2.2.2 (FR), and 5.3.2.2 (NL) of the annexes.

³³⁵ See Boezaart 2013a, p. 369; and section 5.4.2.2 of the annexes.

³³⁶ See sections 5.1.2.2 (AU), 5.2.2.2 (FR), 5.3.2.2 (NL), and 5.4.2.2 (SA) of the annexes.

³³⁷ See the Australian ICL and ILR in section 5.1.1, the French AHA in section 5.2.2.2, and the Dutch FGAL in section 5.3.1 of the annexes.

³³⁸ UNCRC General Comment No. 12 2009, para. 36 and 37.

³³⁹ See the Australian ICL, ILR and GAL in sections 5.1.1 and 5.1.2.2, the French AHA in section 5.2.2.2, the Dutch FGAL and GGAL in section 5.3.1 and 5.3.2.2, and the South African CAL in section 5.4.1 of the annexes.

It is crucial that the representative does not merely represent their own opinion regarding the child's situation, but takes into account the child's views.³⁴⁰ Specifically, the representative ought to transmit the child's views correctly to the court according to the Committee.³⁴¹ In Australia and France, this is explicitly reaffirmed by requiring the representative to inform the court of the child's views when they differ from their view of what is in the child's best interests.³⁴² Finally, in the Netherlands, it is also emphasized that the filiation guardian ad litem should take into account both the immediate short term and the long term interests of the child.³⁴³ For example, by considering the effects of the child having only one legal parent following the denial of parentage. The Australian independent legal representative should also aim at achieving what is in the long term best interests of the child.³⁴⁴

3.5.3. Payment of the child representatives

The payment of the child representatives is a very practical question. However, it is important as it indicates how accessible the representation is for children. As Article 12 paragraph 2 refers to an 'opportunity to be heard', the Committee has noted that representation in legal proceedings must be accessible and barriers, such as costs, must be eliminated.³⁴⁵ If there is no legal aid available and the parties are ordered to pay all the costs, then the representation is less accessible and might be limited to children of well-off parents.³⁴⁶

Of all the forms of representation in the jurisdictions, only one is always automatically fully financed by the State: the guardian ad litem in New South Wales, Australia. An allowance is paid by the panel of the Department of Justice directly.³⁴⁷ For all other forms of representation the child or the parties will be required to pay the costs with, or without, legal aid available. In the most generous situation, legal aid is provided without any further requirements. This is the case for the independent and direct legal representatives in New South Wales in child protection proceedings, the Dutch separate legal representative in secure youth care proceedings, and the French children's lawyer provided for in the framework of Article 388-1 FCC.³⁴⁸ On the opposite end of the scale is the French ad hoc administrator, for which no legal aid is available, although the State Treasury does advance the payment before recovering the costs from the party ordered to pay.³⁴⁹

In all the other situations and for the other forms of representation legal aid is available depending on varying requirements. The most common requirement being the general means and

³⁴⁰ See Lansdown 2011, p. 25; and section 2.2.2.

³⁴¹ UNCRC General Comment No. 12 2009, para. 36 and 37.

³⁴² See Guidelines for Independent Children's Lawyers 2013, para. 5.1, 5.3 and 5.4; Fédération nationale des administrateurs ad hoc 2009, p. 62; and sections 5.1.2.2 and 5.2.2.2 of the annexes.

³⁴³ See Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 6; Schrama 2015; Vlaardingerbroek 2001, p. ¹⁰⁵; and section 5.3.2 of the annexes. ³⁴⁴ See Principle B3 and the explanatory text accompanying Principle E2 (part 2) of the Representation Principles for

Children's Lawyers 2014; and section 5.1.2.2. ³⁴⁵ UNCRC General Comment No. 12 2009, para. 34.

³⁴⁶ Daly 2016, p. 6.

³⁴⁷ See Guardian ad Litem Handbook 2012, para. 35; and section 5.1.1.7 of the annexes.

³⁴⁸ See sections 5.1.1.7 (AU), 5.2.1.7 (FR), and 5.3.1.7 (NL) of the annexes.

³⁴⁹ Art. 1210-3 FCCP. See section 5.2.1.7 of the annexes.

merits test, applied to the Australian independent children's lawyer in child maintenance and international child abduction proceedings and the independent and direct legal representatives in adoption proceedings, the South African curator ad litem, and the Dutch filiation guardian ad litem, and the Dutch general guardian ad litem when it does not concern a conflict between the child and the parent(s).³⁵⁰ If the proceedings do concern a conflict between the child and their parent, then legal aid is automatically available for the Dutch general guardian ad litem without any further requirements.³⁵¹ Legal aid can be granted in parental authority related proceedings when an Australian independent children's lawyer is appointed by the court and legal aid is considered reasonable.³⁵² Legal aid is provided for the South African legal representative upon court order in most types of proceedings if the 'substantial injustice' test is fulfilled.³⁵³ The 'substantial injustice' test is contained in the South African constitution but remains ambiguous.³⁵⁴ Finally, in divorce proceedings the South African courts cannot appoint a legal representative at state expense, but children may directly request a representative at the legal aid board themselves.³⁵⁵

The practical issue of legal costs can be a major obstacle for the child to exercise their right to representation. However, other than for the French administrator ad hoc, either legal aid is available for the child to cover the costs or the representative works at State expense. This appears to alleviate the potential issue of accessibility. In practice, it remains the question whether legal aid boards accept the application and have the resources to fund child representation in family law disputes.

³⁵⁰ See sections 5.1.1.7 (AU), 5.3.1.7 (NL), and 5.4.1.7 (SA) of the annexes.

³⁵¹ See UNCRC Third Periodic Report of the Netherlands 2008, para. 63; and section 5.3.1.7 of the annexes.

³⁵² See Policies Legal Aid New South Wales, 5.3.3 and 5.4.4; and section 5.1.1.7 of the annexes.

³⁵³ See Legal Aid Guide 2014, para. 4.18.1, p. 65; and section 5.4.1.7 of the annexes.

³⁵⁴ S. 28(1)(h) South African Constitution 1996. See section 5.4.1.5 of the annexes.

³⁵⁵ See Heaton 2012, p. 402; Legal Aid Guide 2014, para. 4.18.7, p. 70; and section 5.4.1.7 of the annexes.
3.6. Summary and combined evaluation

In this chapter the legal frameworks of child representation in Australia, France, the Netherlands and South Africa have been compared systematically and evaluated in light of the child's right to be heard and represented in Article 12 UNCRC. In this section, the information from the above is summarized and compiled in the form of a compliance with the UNCRC 'report card' per jurisdiction as shown in Table 9 below.

In Chapter 2 on children's right to participation in the UN Convention on the Rights of the Child, Article 12 was analyzed to determine the key standards set by the UNCRC. The UNCRC standards formed the criteria by which the legal frameworks of the four jurisdictions were evaluated. These standards together also form the subjects on the 'report card' below. There are nine main criteria (in bold), which often consist of multiple sub-criteria. For example, the main criteria of capability, that the child is capable of forming his or her own views, consists of three sub-criteria. The Committee has clarified that there should be no age limit and that decision-makers should start from a presumption of capability to determine capability on a case-by-case basis. For the rest, the Committee remains vague on how States ought to determine capability. That is why the fourth criteria is whether in the jurisdiction the requirement of capability is 'less ambiguous than the Committee'. The compliance with these standards will now be summarized and briefly discussed.

'Assure' the child's right to be heard

The first main criterion is that State parties ought to 'assure' the child's right to be heard, creating a strict legal obligation and requiring them to take proactive steps in assuring this right. The strict legal obligation of having a manner by which children are heard or represented in family law proceedings is met by all jurisdictions as they have at least two forms of representation and additional means by which children can express their views. With regards to proactive steps, differences could be found between the jurisdictions, explaining the differences in grading. While South Africa actively and promptly reformed their legal framework in response to the UNCRC, Australia and France took more time to respond to the requirements of Article 12 UNCRC, and the Netherlands did not consider reform necessary solely on the ground of the UNCRC.

In all matters and proceedings affecting the child

The second main criterion is that children have the right to be heard in 'all matters affecting the child', according to Article 12 paragraph 1, and in 'any judicial and administrative proceedings affecting the child', according to Article 12 paragraph 2. The Committee thrice emphasizes that there should be no limitations or distinctions. Firstly, there should be no limits regarding the types of proceedings. Regarding family law proceedings, this is only the case in Australia. In France, the Netherlands and South Africa there are representatives in all categories of proceedings, but there are

limitations. In various proceedings representatives are not provided for, as discussed in section 3.2. Secondly, there should be no distinctions between the types of proceedings. In all jurisdictions, there is not an extensive distinction, but there is a clear preference for providing separate legal representation forms in vertical family law disputes and more best interests representation forms in the horizontal family law disputes. Finally, there should be no distinction with regards to whom initiated the proceedings. In Australia, France and South Africa this does not occur. In the Netherlands, there is a distinction made regarding the filiation guardian ad litem and the requirements that must be met for appointment (see section 3.3).

	AU (NSW)	FR	NL	SA
'Assure' child's right:				
- Strict legal obligation	++	++	++	++
- Take proactive steps	+	+	~	++
In all matters & proceedings affecting child	:			
- No limits regarding types of proceedings	++	+/_	+/_	+/
- No distinction between types	+	+	+	+
- No distinction between who initiated	++	++	+	++
Capability:		·		
- No age limit	+	+	-	++
- Presumption of capability case-by-case	~	~	~	~
- Less ambiguous than the Committee	++	-	-	n.a.
Capacity: applicable?				Role of LR ++
'Freely' & 'opportunity': Not an obligation	++	+/_	+/_	++
Through a representative:				
- Choice for child	+	+/_	_	+/_
- Represent child's views, not own views	+	+	+	+
- (Potentially a) professional	++	++	++	++
- Knowledge, understanding & experience	++	+	++	++
Accessibility:				
- Legal costs	++	+/_	+	+
- Child may request	++	++	++	++
- Not dependent on a decision	-	+/_	-	-
Child's best interest: not protectionist	~	++	~	~
Implementation steps:			•	
- Prepare and inform the child	+	+	+	+
- Hear child directly	++	++	++	++
- Give feedback to child	+	+	+	+
++: Very good +/-: Both sufficient a	nd insufficient asp	ects	~ : Neur	tral

 Table 9: Compliance of the legal representation frameworks in four jurisdictions with Article 12 UNCRC

 ALL (NSW)

+: Sufficient

+/- : Boin sufficient and insufficient aspects - : Insufficient ~ : Neutral n.a. : Not applicable

Capability of the child to express views

With regards to capability, as discussed above, there are three sub-criteria. As South Africa does not have any child related requirements for the appointment of either the legal representative or curator ad litem, it definitely does not have an age limit and the other criteria are not applicable. The Netherlands does have an age requirement for the separate legal representative in various proceedings, in the other two jurisdictions age requirements are more implicit, as they can be found in their capability

requirements. For example, age is used in the rebuttable presumptions of (in)capability in New South Wales. The second sub-criterion is that decision-makers should employ a presumption of capability and then determine the capability of the child on a case-by-case basis. None of the jurisdictions have an explicit presumption of capability for all children in their law, but they also do not have a presumption of incapability and the child related requirements, if used, are of a personal nature. Thus, all jurisdictions can be qualified as neutral for this matter. Finally, regarding the open requirement employed in three of the jurisdictions for capacity, only the requirement in New South Wales is much less ambiguous than the standard of the Committee. The requirement includes clear criteria as to when the child should be considered capable of giving proper instructions.

Capacity of the child

The fourth main criterion concerns the capacity of the child: in accordance with the child's age and maturity due weight should be given to the views of the child. As discussed above, this is generally an issue for the decision-maker who must weigh the views of the child. However, for the legal representative in South Africa it is also important as the child's capacity determines the legal representative's role. According to the Committee, children's capacity cannot be determined solely on the basis of one the factors. Both age and maturity must be taken into account on a case-by-case basis, and this is also exactly what South African law prescribes.

'Freely' an 'opportunity' to express views

The following main standard, contained both in paragraph 1 and 2 of Article 12 UNCRC, is that the child has the right to be heard and represented, but is not obliged to. In Australia and South Africa this is very much the case. In the France, there is one situation in which representation is obligatory with or without the child's consent, and in the Netherlands there are two. Thus, in both jurisdictions there are good practices, but the mandatory representation is questionable. In these situations, the child must be afforded the option not to express views to meet the standard of Article 12 UNCRC.

Through a representative

The sixth, and probably most important, standard concerns the right of the child to be heard through a representative according to Article 12 paragraph 2 UNCRC. On the basis of the Committee's General Comment, four sub-criteria can be determined. The first is that the child has the choice to be represented. In principle, the fact that each jurisdiction has two or more forms of representation in addition to other means by which they can be heard, provides children with choices. However, these are not limitless choices. The types of proceedings are one potential limit. In the Netherlands there is only one available form of representation per type of proceeding. This means, for example, that the child can only choose for the general guardian ad litem in parental authority proceedings in which a

choice is available between the two forms of representation. In New South Wales, the choice between the representatives is the most far-reaching. However, in all three jurisdictions the child related and conflict related requirements can still limit the remaining choices.

The second sub-criteria concerning representation is that the representative should exclusively represent the child's views and, if required to determine the child's best interests, should do so objectively based on evidence and by including the child's views. In all jurisdictions, the representatives only represent the child. However, as there are many best interests representatives, the issue of objectively presenting the best interests of the child is more difficult. All representatives are required to do so, but whether their final consideration on the best interests is an objective one in practice remains to be seen.

The third and fourth sub-criteria concern the professional background of the representative and that they must have sufficient knowledge, understanding and experience. All jurisdictions and their representatives fulfill both requirements. Albeit that the French are more relaxed concerning the knowledge, understanding and experience required for the administrator ad hoc and children's lawyer.

Accessibility

The seventh main criterion is that representation should be accessible. According to the Committee, potential barriers should be eliminated. The hurdle of legal costs is the lowest for children in New South Wales, Australia and the highest for children (and their parents) in France. There is no legal barrier in any of the jurisdictions that prevents children from being able to request legal representatives, but the final decision is – by and large – left to the courts or other panel of adults (e.g. Legal Aid Board). This limits the autonomy of children in their choice whether or not to be represented in family law disputes.

Child's best interests and Article 3

In light of Article 3 UNCRC, the best interests of the child should be a primary consideration in family law proceedings and the implementation of Article 12 UNCRC. However, it should not function as a protectionist argument to obstruct the child's right to be represented. Which is what may possibly occur in practice with the Australian, Dutch, and South African requirements that the appointment of a representative must be necessary in the best interests of the child.

Implementation steps of Article 12

Finally, three of the Committee's five steps for the implementation of Article 12 are relevant in light of the legal framework of representation. All three – the preparation of the child by providing information, the hearing of the child and the giving of feedback to the child – must be undertaken by the representation forms of the jurisdictions. Although the specifics may differ, see section 3.5.2, the overall picture is a positive one.

4. Conclusion

In this research the child's right to be heard and represented in Article 12 UNCRC has been thoroughly examined. The four legal frameworks of child representation in Australia, France, the Netherlands and South Africa were systematically studied and compared. The latter was evaluated using the standards derived from the human rights analysis to tackle the main research question: To what extent is the minimum international human rights standard for child representation provided by Article 12 UNCRC complied with by the legal framework for family law proceedings in four jurisdictions?

The answer to this question is contained in the entire third chapter, specifically in the overview of Table 9 and summary in section 3.6 above. Each of the jurisdictions' legal representation frameworks have been shown to have advantages and disadvantages in relation to the standard of Article 12 UNCRC. But, for the short answer to the main research question, how do the advantages and disadvantages add up? In Figure 2, the overall compliance of the jurisdictions' representation frameworks with the UNCRC are shown.³⁵⁶





Compliance with the minimum standards of Article 12 UNCRC

All in all, the four jurisdictions meet most of the minimum standards contained in Article 12 UNCRC to provide children with representation in family law proceedings. Of course, there are differences between the jurisdictions. The legal representation framework of Australia, specifically New South Wales, scores the highest in compliance. With four forms of accessible representation available in all types of family law proceedings, and extensive detailed laws and guidelines containing clear requirements, the framework answers best to the standards of the UNCRC. Australia is closely followed by South Africa, another jurisdiction with detailed provisions and guidelines for the appointment and functioning of the legal representative and with an additional, less regulated,

³⁵⁶ The total compliance score has been calculated based on the scores in Table 9 of section 3.6. Each sign was given a score: ++ = 2, + = 1, +/- = 0,5, ~ = 0 and -= -1. These were totaled up and then divided by the number of scores given to calculate the mean score. The mean score of each jurisdiction is: Australia 1.3, France 1.0, the Netherlands 0.7, and South Africa 1.2. The jurisdictions were placed along the scale in Figure 4 with the symbols given the same score as used for the calculation.

representative in common law: the curator ad litem. The two civil law jurisdictions, France and the Netherlands have frameworks of representation which conform less with the minimum standards of the UNCRC. Although both remain on the positive side of the scale, several improvements can be made. In both jurisdictions, the limits in the types of cases, the ambiguity of their capability requirements, and the lack of choice for the child are issues. With the Netherlands having the additional drawback of an age limit for the separate legal representative.

However, the criticism should not be reserved solely for the Netherlands or France. There is room for improvement in the legal frameworks of all jurisdictions. For example, concerning the level of dependency the child has on an adult decision-maker for the appointment of a representative or the explicit presumption of capability for all children prior to a case-by-case examination. Aside from the existing issues in these jurisdictions, the question also remains how these legal frameworks function in practice. This has not been examined in this research, but the manner in which these frameworks are implemented may reveal more imperfections.

Margins of discretion in Article 12 UNCRC

Although the Committee on the Rights of the Child provides further clarification on Article 12 UNCRC, through their General Comments and in Concluding Observations, certain margins of discretion remain. With regards to the right to representation two main issues have been left open by the Committee. The first issue is that the Committee remains very ambiguous as to how the child's capability should be determined. The second issue that has been sidetracked by the Committee is which form(s) of representation should be offered to children. The Committee spares no words on which types of representative should be made available or which form would be most effective. As Parkes remarked, it is an area shrouded in confusion.³⁵⁷

The comparative research has shown that these margins of discretion afforded by the Committee have been dealt with by the different jurisdictions in a variety of manners. For example, the open capability requirements for representation in France and the Netherlands mirror the ambiguity of the Committee, (potentially) impeding children's right to representation. The Committee could learn something from New South Wales in this regard. The open capability requirement there has been additional given shape with rebuttable presumptions and a variety of further factors that must be considered on a case-by-case basis. Concerning the second main margin, the form(s) of representation, this research has identified two general forms of representative and the separate legal representative. In addition, this research has shown that although two general forms of representation have their own peculiarities.

³⁵⁷ Parkes 2013, p. 255.

Where there is a margin of discretion, differences are to be expected between jurisdictions. In fact, differences may even be applauded as a variety of best practices may develop in each State. Through the use of comparative research, States can then learn from each other and potentially transplant best practices through legal reform in their own systems. However, further guidance by the Committee regarding these margins in Article 12 might also be constructive. Clarification of the minimum standards to be adhered to by States would raise the bar for national jurisdictions. For example, if the requirement of capability would be further specified children would have more certainty as to when they have the right to express their views. Of course, reservations must be expressed. If the Committee stipulates further guidelines, then these should be (empirically) tested best practices which are suitable for implementation in all legal cultures and systems. The Committee should not express which form of representation ought to be provided and thereby exclude other forms, without certainty that it is important that further comparative and empirical research is conducted as to child representation to inform the Committee and States of existing best practices and the variety of options available.

Novel aspects of this research

This comparative research included two novel aspects: a widened geographical scope and a focus on family law proceedings in the broad sense. This research departed from the 'Anglocentric focus' found in existing comparative research by including a mixed jurisdiction, a common law jurisdiction, and two civil law jurisdictions. When looking at the historical development of the representation forms in each of the jurisdictions, the legal traditions visibly had an influence. By including four jurisdictions with varying traditions, more divergent approaches have been studied. In general, it can be concluded from Figure 2 that the civil law jurisdiction (France and the Netherlands) could learn from the legal framework in the common law jurisdiction (Australia) and mixed jurisdiction (South Africa). The focus on family law proceedings in the broad sense revealed the inconsistencies between types of proceedings. These inconsistences include the differences between horizontal and vertical family law disputes and between the tasks of representatives in varying proceedings. By looking at all types of family law proceedings, this research has presented a more complete picture of child representation.

The thorough comparison and evaluation of child representation in four jurisdictions has set out the legal frameworks and their compliance with Article 12 UNCRC. It has revealed good practices, which can be of value for other legislatures and the Committee on the Rights of the Child. It has also exposed shortcomings. Further normative comparative, multi-disciplinary, and empirical research is required to transform the conclusions from this exploratory research into conclusions for practical application. To conclude, although room for improvement remains for the four jurisdictions as well as for the guidance by the Committee, the jurisdictions have already made significant strides in complying with the minimum standard of the child's right to be represented in family law proceedings. Looking back at the old Victorian proverb, it is time for a new maxim in accordance with Article 12 UNCRC to encourage legislatures and the judiciary to take children seriously in family law proceedings:

'Children should be seen, and, if they so wish, heard and represented'

Bibliography

Primary sources

Australian Law Reform Commission 1997

Australian Law Reform Commission, *Seen and heard: priority for children in the legal process*, Sydney: Commonwealth of Australia 1997.

Care and Protection Practice Standards 2015

Legal Aid New South Wales, *Care and Protection Practice Standards*, 2015, available at: www.legalaid.nsw.gov.au/__data/assets/pdf_file/0005/7088/Care-and-Protection-Panel-Practice-Standards.pdf (last visited 19 March 2017).

Chartre nationale de la defense des mineurs 2008

Commission « droit des mineurs » de la Conference des bâtonniers, 'La Chartre nationale de la defense des mineurs', *Journal du droit des jeunes*, No. 275, 2008, pp. 42-43.

Further Revised Explanatory Memorandum Family Law Amendment Bill 2000

The Parliament of the Commonwealth of Australia, House of Representatives, *Further Revised Explanatory Memorandum Family Law Amendment Bill 2000.*

Guardian ad Litem Handbook 2012

State of New South Wales Department of Attorney General and Justice, *Guardian ad Litem Handbook*, Sydney: State of New South Wales 2012.

Guide Méthodologique Administrateur ad hoc 2014

Direction des Affaires criminelles et des Grâces, Direction des affaires civiles et du sceau, Direction des services judiciaires, and Service de l'accès au droit et à la justice et de l'aide aux victimes, *Guide Méthodologique Administrateur ad hoc: Représentation judiciaire et accompagnement des enfants victimes d'infractions*, Paris: Ministère de Justice 2014, available at: www.justice.gouv.fr/publication/guide_aah.pdf (last visited 19 March 2017).

Guidelines for Independent Children's Lawyers 2013

Family Court of Australia, *Guidelines for Independent Children's Lawyers*, 2013, available at: www.familycourt.gov.au/wps/wcm/connect/fcoaweb/about/policies-and-procedures/guidelinesindependent-childrens-lawyer (last visited 19 March 2017).

Guidelines Legal Aid New South Wales

Legal Aid New South Wales, *Guidelines*, available at: www.legalaid.nsw.gov.au/for-lawyers/policyonline/guidelines (last visited 20 December 2016).

Inquiry into the Australian Legal Aid System 1997

Senate Legal and Constitutional References Committee, *Inquiry into the Australian Legal Aid System:* Second Report, Canberra: Commonwealth of Australia 1997.

Inschrijvingsvoorwaarden advocatuur 2016

Raad voor Rechtsbijstand, *Inschrijvingsvoorwaarden advocatuur 2016*, 2016, available at: www.rvr.org/binaries/content/assets/rvrorg/advocaten/overaanvragen/inschrijven/inschrijvingsvoorwaarden-advocatuur-2016-versie-1-00---ondertekend-doorbestuur.pdf (last visited 19 March 2017).

Legal Aid Guide 2014

Legal Aid South Africa, *Legal Aid Guide 2014*, 13th edition, 2014, available at: www.legal-aid.co.za/wp-content/uploads/2012/03/Legal-Aid-Guide-2014.pdf (last visited 19 March 2017).

Kamerstukken I 2008/09, 30145, E

Nader Memorie van Antwoord Wet bevordering voortgezet ouderschap en zorgvuldige scheiding, *Kamerstukken I* 2008/09, 30145, E.

Kamerstukken II 1992/93, 23012, 3

Memorie van Toelichting Nadere regeling van het gezag over en van de omgang met minderjarige kinderen, *Kamerstukken II* 1992/93, 23012, 3.

Kamerstukken II 2004/05, 30145, 3

Memorie van Toelichting Wet bevordering voortgezet ouderschap en zorgvuldige scheiding, *Kamerstukken II* 2004/05, 301245, 3.

Kamerstukken II 2005/06, 30644, 3

Memorie van Toelichting Wijziging van de Wet op de jeugdzorg met betrekking tot jeugdzorg waarop aanspraak bestaat ingevolge de wet in gesloten setting (gesloten jeugdzorg), *Kamerstukken II* 2005/06, 30644, 3.

Pathways for Children 2004

Family Law Council, *Pathways for Children: A Review of Child Representation in Family Law*, Canberra: Commonwealth of Australia 2004.

Policies Legal Aid New South Wales

Legal Aid New South Wales, *Policies*, available at: www.legalaid.nsw.gov.au/for-lawyers/policyonline/policies (last visited 19 March 2017).

Representation Principles for Children's Lawyers 2014

The Law Society of New South Wales, *Representation Principles for Children's Lawyers*, 4th edition, 2014, available at: www.lawsociety.com.au/cs/groups/public/documents/internetcontent/026517.pdf (last visited 19 March 2017).

Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014

Landelijk Overleg Vakinhoud Familie- en Jeugdrecht, *Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW*, 2014, available at: www.rechtspraak.nl/SiteCollectionDocuments/Richtlijnbenoeming-bijzondere-curator-ogv-artikel-212-boek-1-BW.pdf (last visited 19 March 2017).

Senate Legal and Constitutional Affairs Committee 2011

Senate Legal and Constitutional Affairs Committee, *Report on International parental child abduction to and from Australia*, Canberra: Commonwealth of Australia 2011.

Rapport Staatscommissie Herijking ouderschap 2016

Staatscommissie Herijking ouderschap, *Kind en Ouders in de 21^{ste} Eeuw*, Den Haag: Staatscommissie Herijking ouderschap 2016.

UNCHR Question of a Convention on the Rights of the Child 1980

United Nations Economic and Social Council Commission on Human Rights, *Question of a Convention on the Rights of the Child: Note verbale dated 5 October 1979 addressed to the Division of Human Rights by the Permanent Representation of the Polish People's Republic to the United Nations in Geneva*, 17 January 1980, E/CN.4/1349*.

UNCHR Report of the Working Group 1981

United Nations Economic and Social Council Commission on Human Rights, *Report of the Working Group on a Draft Convention on the Rights of the Child*, 17 February 1981, E/CN.4/L.1575

UNCHR Draft Convention on the Rights of the Child 1987

United Nations Economic and Social Council Commission on Human Rights, *Draft Convention on the rights of the child, text of the draft convention as adopted by the Working Group (First to ninth sessions: 1979-1987),* 16 October 1987, E/CN.4/1988/WG.1/WP.1.

UNCHR Report of the Working Group 1989

United Nations Economic and Social Council Commission on Human Rights, *Report of the Working Group on a Draft Convention on the Rights of the Child*, 2 March 1989, E/CN.4/1989/48.

UNCRC Concluding Observations Australia 2005

United Nations Committee on the Rights of the Child, *Consideration of reports submitted by States parties under Article 44 of the Convention: Concluding observations – Australia*, 20 October 2005, CRC/C/15/Add.268.

UNCRC Concluding Observations France 2004

United Nations Committee on the Rights of the Child, Consideration of reports submitted by States parties under Article 44 of the Convention: Concluding observations – France, 30 June 2004, CRC/C/15/Add.240.

UNCRC Concluding Observations France 2009

United Nations Committee on the Rights of the Child, Consideration of reports submitted by States parties under Article 44 of the Convention: Concluding observations – France, 22 June 2009, CRC/C/FR/CO/4.

UNCRC General Comment No. 5 2003

United Nations Committee on the Rights of the Child, General comment no. 5 (2003): General measures of implementation of the Convention on the Rights of the Child, 27 November 2003, CRC/GC/2003/5.

UNCRC General Comment No. 10 2007

United Nations Committee on the Rights of the Child, *General Comment No. 10: Children's Rights in Juvenile Justice*, 25 April 2007, CRC/C/GC/10.

UNCRC General Comment No. 12 2009

United Nations Committee on the Rights of the Child, *General Comment No. 12: The Right of the Child to be Heard*, 1 July 2009, CRC/C/GC/12.

UNCRC General Comment No. 14 2013

United Nations Committee on the Rights of the Child, *General Comment No. 14: The right of the child to have his or her best interests taken as a primary consideration*, 29 May 2013, CRC/C/GC/14.

UNCRC Initial Periodic Report of Australia 1996

United Nations Committee on the Rights of the Child, *Consideration of reports submitted by States parties under Article 44 of the Convention: Initial reports of State parties due in 1993 – Australia, 8 January 1996*, CRC/C/8/Add.31.

UNCRC Initial Periodic Report of South Africa 1998

United Nations Committee on the Rights of the Child, *Consideration of reports submitted by States parties under Article 44 of the Convention: Initial reports of State parties due in 1997 – South Africa, 22 May 1998*, CRC/C/51/Add.2.

UNCRC Recommendations: Day of General Discussion 2006

United Nations Committee on the Rights of the Child, *Recommendations: Day of General Discussion* on the Right of the Child to be Heard, 29 September 2006.

UNCRC Third and Fourth Periodic Report of France 2008

United Nations Committee on the Rights of the Child, *Consideration of reports submitted by States parties under Article 44 of the Convention: Third and fourth periodic reports of State parties due in 2007 – France, 21 February 2008, CRC/C/FRA/4.*

UNCRC Third Periodic Report of the Netherlands 2008

United Nations Committee on the Rights of the Child, *Consideration of reports submitted by States parties under Article 44 of the Convention: Third periodic reports of State parties due in 2007 – the Netherlands*, 23 July 2008, CRC/C/NLD/3.

Werkproces benoeming bijzondere curator o.g.v. Art. 1:250 BW 2014

Landelijk Overleg Vakinhoud Familie- en Jeugdrecht, *Werkproces benoeming bijzondere curator o.g.v. Art. 1:250 BW*, 2014, available at: www.rechtspraak.nl/SiteCollectionDocuments/Werkproces-benoeming-bijzondere-curator-ogv-artikel-250-boek-1-BW.pdf (last visited 19 March 2017).

Case Law

<u>Australia</u> *Re K* [1994] FLC 92-461. *In the Matter of P and P* [1995] FLC 92-615. *De L* [1996] FLC 92-706. *Re Oscar* [2002] NSWSC 453 (21 May 2002).

France

Cour de Cassation 1ére Chambre civile, 23 février 1999 (Bull. N°66).

The Netherlands

Hoge Raad 4 February 2005, ECLI:NL:HR:2005:AR4850. Hoge Raad 6 October 2006, ECLI:NL:HR:2006:AV9444.

South Africa

Du Toit v Minister of Welfare and Population Development (Lesbian and Gay Equality Project as Amicus Curiae) (CCT 40/01) [2002] ZACC 20 (10 September 2002).

Soller NO v G [2003] (5) SA 430 (W).

Centre for Child Law and Another v Minister of Home Affairs and Others [2005] (6) SA 50 (T) (13 September 2004).

S v M (Centre for Child Law as Amicus Curiae) (CCT 53/06) [2007] ZACC 18 (26 September 2007). AD v DW (Centre for Child Law as Amicus Curiae; Department for Social Development as Intervening Party) [2007] ZACC 27 (7 December 2007).

Legal Aid Board v R and another [2009] (2) SA 262 (D).

S v J (695/10) [2010] ZASCA 139 (19 November 2010).

Legal Aid Board in re Four Children (512/10) [2011] ZASCA 39 (29 March 2011).

Ex parte Centre for Child Law in re: six minor children [2012] Unreported case no. 11762/2012. *B v G* [2012] 2 SA 329 (SGJ HC).

Literature

Archard & Skivenes 2009

D. Archard & M. Skivenes, 'Balancing a Child's Best Interests and a Child's Views', *The International Journal of Children's Rights*, Vol. 17, No. 1, 2009, pp. 1-21.

Asser 1957

C. Asser, 'Handleiding tot de beoefening van het Nederlands Burgerlijk Recht: Eerste Deel Personenrecht', 9th edition (J. Wiarda ed.), Zwolle: W.E.J. Tjeenk Willink 1957.

Attias 2012

D. Attias, 'Les nouveaux droits de l'enfant sont-ils compatibles avec sa protection et ne mettent-ils pas en danger l'autorité parentale?', *Issu de Petites affiches*, No. 50, 2012.

Atwood 2011

B.A. Atwood, 'Representing Children Who Can't or Won't Direct Counsel: Best Interests Lawyering or No Lawyer at All?', *Arizona Legal Studies* (Discussion paper No. 11-16), 2011.

Avenard 2015

G. Avenard, 'Le droit de s'exprimer', L'école des parents, No. 615, 2015, pp. 32-34.

Bala, Birnbaum & Bertrand 2013

N. Bala, R. Birnbaum & L. Bertrand, 'Controversy About the Role of Children's Lawyers: Advocate or Best Interests Guardian? Comparing Practices in Two Canadian Jurisdictions with Different Policies for Lawyers', *Family Court Review*, Vol. 51, No. 4, 2013, pp. 681-97.

Ballard et al. 2014

R.H. Ballard et al., 'Hearing the Voice of the Child in Divorce', in: M.K. Miller, J. Chamberlain & T. Wingrove, *Psychology, Law, and the Wellbeing of Children*, Oxford: Oxford University Press 2014.

Bao-Er 2006

D. Bao-Er, 'Care and Protection: The role of guardians ad litem in the Children's Court', *New South Wales Law Society Journal*, Vol. 44, No. 4, 2006, pp. 60-63, available at: ssrn.com/abstract=1143743 (last visited 19 March 2017), (SSRN version used with pp. 1-7).

Barrie 2013

G.N. Barrie, 'Giving due consideration to views expressed by the child in family law proceedings: the Australian experience and lessons for South Africa', *Tydskrif vir die Suid-Afrikaanse Reg*, No. 1, 2013, pp. 124-138.

Bazin 2014

E. Bazin, 'De l'audition du mineur dans les procedures relatives à l'autorite parentale', *Issue du Gazette du Palais*, No. 200, 2014.

Bell 2015

F. Bell, *Discussion paper: Facilitating the Participation of Children in Family Law Processes*, Lismore: Southern Cross University 2015.

Bilson & White 2005

A. Bilson & S. White, 'Representing Children's Views and Best Interests in Court: An International Comparison', *Child Abuse Review*, Vol. 14, 2005, pp. 220-239.

Birnbaum & Bala 2009

R. Birnbaum & N. Bala, 'The Child's Perspective on Legal Representation: Young Adults Report on their Experiences with Child Lawyers', *Canadian Journal of Family Law*, Vol. 25, 2009, pp. 11-71.

Bisschop 1904

W.R. Bisschop, 'Roman-Dutch Law in South-Africa', *The Law Quarterly Review*, Vol. 20, No. 1, 1904, pp. 41-52.

Black 1994

M. Black, *Monitoring the Rights of Children: Global Seminar Report*, Florence: UNICEF International Child Development Centre 1994.

Boele-Woelki 2008

K. Boele-Woelki, 'What comparative family law should entail', *Utrecht Law Review*, Vol. 4, No. 2, 2008, pp. 1-24.

De Boer 1992

Th.M. de Boer, 'Vergelijkenderwijs: de inspiratie van buitenlands recht', *WPNR*, Vol. 123, No. 6033, 1992, pp. 39-48.

Boezaart 2013a

T. Boezaart, 'Listening to the Child's Objection', *New Zealand Law Review*, Vol. 26, No. 3, 2013, pp. 357-372.

Boezaart 2013b

T. Boezaart, 'The role of a curator *ad litem* and children's access to the courts', *De Jure*, 2013, pp. 707-726.

Boezaart & de Bruin 2011

T. Boezaart & D.W. de Bruin, 'Section 14 of the Children's Act 38 of 2005 and the child's capacity to litigate', *De Jure*, Vol. 44, No. 2, 2011, pp. 416-438.

Broekhuijsen-Molenaar 2016

A.M.L. Broekhuijsen-Molenaar, 'Commentaar op artikel 1:212 BW', in: J.H. Nieuwenhuis, C.J.J.M. Stolker & W.L. Valk (eds.), *Tekst & Commentaar Burgerlijk Wetboek*, Deventer: Kluwer 2016.

De Bruin 2010

D.W. de Bruin, *Child Participation and Representation in Legal Matters* [diss. University of Pretoria], available at: www.repository.up.ac.za/dspace/bitstream/handle/2263/27414/Complete.pdf?sequence=9 (last visited on 19 March 2017).

Duval 2008

N. Duval, 'Commentaires: La Chartre nationale de la defense des mineurs', *Journal du droit des jeunes*, No. 275, 2008, pp. 43-44.

Carson et al. 2014

R. Carson et al., 'The role and efficacy of Independent Children's Lawyers: Findings from the AIFS Independent Children's Lawyer Study', *Family Matters*, No. 94, 2014, pp. 58-69.

Cashmore 2011

J. Cashmore, 'Children's participation in family law decision-making: Theoretical approaches to understanding children's views', *Children and Youth Services Review*, Vol. 33, No. 4, 2011, pp. 515-520.

Cleophas & Assim 2015

K-A. Cleophas & U.M. Assim, 'Child Participation in Family Law Matters Affecting Children in South Africa', *European Journal of Law Reform*, Vol. 17, No. 2, 2015, pp. 294-304.

Daly 2016

A. Daly, 'Hearing Children in Proceedings – What we can learn from global practice', *Seen and Heard*, Vol. 26, No. 1, 2016, pp. 1-11.

Demarchi 2010

J-R. Demarchi, 'Une justice familiale en mutation (Bilan de l'evolution legislative relative à l'audition de l'enfant)', *Issu de Petites affiches,* No. 54, 2010.

Elrod 2007

L.D. Elrod, 'Client-Directed Lawyers for Children: It is the "Right Thing to Do', *Pace Law Review*, Vol. 27, 2007, pp. 869-920.

Favre-Lanfray 2008a

G. Favre-Lanfray, 'Les multiples facettes de l'administrateur ad hoc', in: Observatoire national de l'enfance en danger, *Dossier thématique de l'Oned 2008-1: Les administrateurs ad hoc*, 2008, pp. 2-10, available at: www.onpe.gouv.fr/system/files/publication/dossierthematique_administrateursadhoc_2008_5.pdf (last visited 19 March 2017).

Favre-Lanfray 2008b

G. Favre-Lanfray, 'Les missions d'administrateur ad hoc au penal et au civil', in: Observatoire national de l'enfance en danger, *Dossier thématique de l'Oned 2008-1: Les administrateurs ad hoc*, 2008, pp. 11-13, available at: www.onpe.gouv.fr/system/files/publication/dossierthematique_administrateursadhoc_2008_5.pdf (last visited 19 March 2017).

Fédération nationale des administrateurs ad hoc 2009

Fédération nationale des administrateurs ad hoc, Rapport: La représentation « ad hoc » du mineur,Tullins:FE.N.A.A.H.2009,availableat:www.reforme-enfance.fr/images/documents/rapportfenaah.pdf (last visited 19 March 2017).

Ferreira 2013

S. Ferreira, 'It is My Adoption, But is Anyone Listening to Me?', *New Zealand Law Review*, Vol. 26, No. 3, 2013, pp. 335-355.

Fernando 2013

M. Fernando, 'How Can We Best Listen to Children in Family Law Proceedings?', *New Zealand Law Review*, Vol. 26, No. 3, 2013, pp. 387-407.

Fernando 2014

M. Fernando, 'Family Law Proceedings and the Child's Right to be Heard in Australia, the United Kingdom, New Zealand, and Canada', *Family Court Review*, Vol. 52, No. 1, 2014, pp. 46-59.

Fortin 2005

J. Fortin, Children's Rights and the Developing Law, Cambridge: Cambridge University Press 2005.

Fossier 2002

T. Fossier, 'La responsabilité civile dans la protection de l'enfance: le cas de l'administrateur ad hoc', *Issu de Gazette du Palais*, No. 15, 2002.

Fossier 2006

T. Fossier, 'L'administrateur ad hoc: actualité d'une vieille question', *Journal du droit des jeunes*, No. 254, 2006, pp. 16-22.

Glenn 2012

H.P. Glenn, 'The aims of comparative law', in: J.M. Smits (ed.), *Elgar Encyclopedia of Comparative Law*, Cheltenham: Edward Elgar Publishing 2012, pp. 65-74.

Gouttenoire 2006

A. Gouttenoire, 'Les modes de participation de l'enfant aux primordial judiciaires', *Cahiers de la recherche sur les droits fondamentaux*, No. 5, 2006, pp. 59-64.

De Graaf & Limbeek 2011

J.H. de Graaf & M.M.C. Limbeek, 'Het CCRA Seminar: De Rechten van het Kind en Echtscheiding', *FJR*, 2011/4.

Grevot 2010

A. Grevot, 'L'administrateur ad hoc: l'enfant délaissé du dispositif de protection de l'enfance', *Journal du droit des jeunes*, No. 297, 2010, pp. 12-19.

Ter Haar 2015

J.H.M. ter Haar, 'De processuele handelingsonbekwaamheid van de minderjarige', EB, 2015/22.

Ter Haar 2016

J.H.M. ter Haar, 'Commentaar op artikel 1:250 BW', in: J.H. Nieuwenhuis, C.J.J.M. Stolker & W.L. Valk (eds.), *Tekst & Commentaar Burgerlijk Wetboek*, Deventer: Kluwer 2016.

Hale 2006

B. Hale, 'Children's Participation in Family Law Decision-Making: Lessons from Abroad', *Australian Journal of Family Law*, Vol. 20, No. 2, 2006, pp. 119-126.

Hart 1992

R.A. Hart, *Children's Participation: From Tokenism to Citizenship*, Florence: UNICEF International Child Development Centre 1992.

Heaton 2012

J. Heaton, 'South Africa: Changing the contours of child and family law', in: E.E. Sutherland (ed.), *The Future of Child and Family Law: International Predictions*, New York: Cambridge University Press 2012, pp. 398-428.

Helmholz 1978

R.H. Helmholz, 'The Roman Law of Guardianship in England: 1300-1600', *Tulane Law Review*, Vol. 52, No. 2, 1978, pp. 223-257.

Herbots & Put 2015

K. Herbots & J. Put, 'The Participation Disc: A Conceptual Analysis of (a) Child('s Right to) Participation', *International Journal of Children's Rights*, Vol. 23, 2015, pp. 154-188.

Hodgkin & Newell 2007

R. Hodgkin & P. Newell, *Implementation Handbook for the Convention on the Rights of the Child*, New York: UNICEF 2007.

Jansen 2016a

E. Jansen, 'De eigen(aardige) procesbevoegdheid van de minderjarige', *NJB* 2016/1563, pp. 2177-2183.

Jansen 2016b

E. Jansen, 'Commentaar op artikel 250 Boek 1 BW', in: S.F.M. Wortmann (ed.), *Groene Serie Personen- en Familierecht*, Deventer: Kluwer 2016.

Jonker et al. 2015

M. Jonker et al., Internationale kinderontvoering: De uitvoeringspraktijk van inkomende zaken in Nederland, Engeland & Wales, Zweden en Zwitserland, Den Haag: WODC 2015.

Justen 2014

M. Justen, 'Audition et intérêt supérieur de l'enfant danse le contentieux familial', *Journal du droit des jeunes*, No. 338-339, 2014, pp. 38.

Kaspiew et al. 2013

R. Kaspiew et al., *Independent Children's Lawyer Study: Final Report*, Canberra: Attorney-General's Department 2013.

Kassan 2003

D. Kassan, 'The Voice of the Child in Family Law Proceedings', *De Jure*, Vol. 36, No. 2, 2003, pp. 164-179.

Kaufman & Flekkoy 1998

N.H. Kaufman & M.G. Flekkoy, 'Participation Rights of the Child: Psychological and Legal Considerations', *Children's Legal Rights Journal*, Vol. 18, No. 1, 1998, pp. 15-29.

Kentie & Hendriks 2013

D.J.D. Kentie & J.A.M. Hendriks, 'Artikel 1:250 BW, de bijzondere curator, in het belang van vechtscheidingskinderen?', *EB*, 2013/93.

Klaas 2009

J. Klaas, 'De bijzondere curator ex art. 1:250 BW; enkele bijzondere toepassingen uit de praktijk', *FJR*, 2009/79.

Koh Peters 2005

J. Koh Peters, 'How Children are Heard in Child Protective Proceedings, in the United States and Around the World in 2005: Survey Findings, Initial Observations, and Areas for Further Study', *Nevada Law Journal*, Vol. 6, No. 3, 2005, pp. 966-1073.

Krappmann 2010

L. Krappmann, 'The weight of the child's view (Article 12 of the Convention on the Rights of the Child)', *International Journal of Children's Rights*, Vol. 18, 2010, pp. 501-513.

Krause 2006

H.D. Krause, 'Comparative Family Law: Past Traditions Battle Future Trends – and Vice Versa', in:M. Reimann & R. Zimmermann (eds.), *The Oxford Handbook of Comparative Law*, Oxford: Oxford University Press 2006, pp. 1099-1129.

Lansdown 2011

G. Lansdown, Every Child's Right to be Heard: A Resource Guide on the UN Committee on the Rights of the Child General Comment No. 12, London: Save the Children Fund 2011.

Lansdown 2016

G. Lansdown, 'Best interests of the child and the right to be heard', in: M. Sormunen (ed.), *The best interests of the child – A dialogue between theory and practice*, Strasbourg: Council of Europe 2016, pp. 31-35.

Lebrun 2001

P-B. Lebrun, 'L'administrateur ad hoc', Journal du droit des jeunes, No. 207, 2001, pp. 33-34.

Le Mintier 1994

B. Le Mintier, 'Présentation de la loi n° 93-22 du 8 janvier 1993 modifiant le Code civil relative à l'État civil, à la famille et aux droits de l'enfant et instituant le Juge aux Affaires Familiales', *Revue juridique de l'Ouest*, No. 1, 1994, pp. 1-31.

Van Leuven & de Klerk 2013

C.A.R.M. van Leuven & M.J. de Klerk, 'Formele en informele rechtsingang(en) minderjarige', *EB* 2013/32.

Van Leuven et al. 2015

C.A.R.M. van Leuven et al., 'Pilot: de gedragskundige als bijzondere curator', FJR, 2015/57.

Liefaard & Doek 2015

T. Liefaard & J.E. Doek, 'Kinderrechten in de rechtspraak: een internationaal perspectief', *FJR*, 2015/20.

Liefaard & Vonk 2016

T. Liefaard & M. Vonk, 'The Rights of the Child in the Netherlands: A Family Law Perspective', in:
O. Cvejić Jančić (ed.), *The Rights of the Child in a Changing World: 25 Years after The UN Convention on the Rights of the Child*, Dordrecht: Springer 2016, pp. 313-330.

Limbeek & Bruning 2015

M. Limbeek & M. Bruning, 'The Netherlands: Two Decades of the CRC in Dutch Case Law', in: T. Liefaard & J.E. Doek (eds.), *Litigating the Rights of the Child*, Dordrecht: Springer 2015, pp. 89-104.

Lok & Vonk 2016

B.L. Lok & M. Vonk, 'De bijzondere curator in afstammingszaken en het belang van het kind', *FJR*, 2016/10.

Lowe & Murch 2001

N. Lowe & M. Murch, 'Children's Participation in the family justice system – translating principles into practice', *Child & Family Law Quarterly*, Vol. 13, No. 2, 2001, pp. 137-158.

Lücker-Babel 1995

M-F. Lücker-Babel, 'The right of the child to express views and to be heard: An attempt to interpret Article 12 of the UN Convention on the Rights of the Child', *The International Journal of Children's Rights*, Vol. 3, No. ³/₄, 1995, pp. 391-404.

Lundy 2007

L. Lundy, "Voice' is not enough: 93rimordial93zing Article 12 of the United Nations Convention on the Rights of the Child', *British Educational Research Journal*, Vol. 33, No. 6, 2007, pp. 927-942.

Massip 1995

J. Massip, 'Administration légale et autorité parentale (a)', Issu de Defrénois, No. 8, 1995.

McIntosh, Bryant & Murray 2008

J.E. McIntosh, D. Bryant & K. Murray, 'Evidence of a Different Nature: The Child-Responsive and Less Adversarial Initiatives of the Family Court of Australia', *Family Court Review*, Vol. 46, No. 1, 2008, pp. 125-136.

Milne 2015

B. Milne, *Rights of the Child: 25 years after the adoption of the UN Convention*, Cham: Springer 2015.

Morag, Rivkin & Sorek 2012

T. Morag, D. Rivkin & Y. Sorek, 'Child Participation in the Family Courts – Lessons from the Israeli Pilot Project', *International Journal of Law, Policy and the Family*, Vol. 26, No. 1, 2012, pp. 1-30.

Nicola 2010

F.G. Nicola, 'Family Law Exceptionalism in Comparative Law', *The American Journal of Comparative Law*, Vol. 58, 2010, pp. 777-810.

Oderkerk 2001

M. Oderkerk, 'The Importance of Context: Selecting Legal Systems in Comparative Legal Research', *Netherlands International Law Review*, Vol. 48, No. 3, 2001, pp. 293-318.

Oderkerk 2015

M. Oderkerk, 'The Need for a Methodological Framework for Comparative Legal Research: Sense and Nonsense of 'Methodological Pluralism' in Comparative Law', *Rabels Zeitschrift für ausländisches und internationales Privatrecht*, Vol. 79, No. 3, 2015, pp. 589-623.

Örücü 2007

E. Örücü, 'Methodological Aspects of Comparative Law', *European Journal of Law Reform*, Vol. 8, No. 1, 2007, pp. 29-42.

Parkes 2013

A. Parkes, *Children and International Human Rights Law: The Right of the Child to be Heard*, London: Routledge 2013.

Parkinson 2001

P. Parkinson, 'The child participation principle in child protection law in New South Wales', *The International Journal of Children's Rights*, Vol. 9, No. 3, 2001, pp. 259-272.

Parkinson & Cashmore 2008

P. Parkinson & J. Cashmore, *The Voice of a Child in Family Law Disputes*, Oxford: Oxford University Press 2008.

Pérez Manrique 2007

R.C. Pérez Manrique, 'Participación Judicial de los Niños, Niñas y Adolescentes', *UNICEF Justicia y Derechos del Niño*, No. 9, 2007, pp. 251-277.

Pieters 2008

I.J. Pieters, 'De bijzondere curator: Quo vadis?', FJR, 2008/78.

Pieters 2012

I.J. Pieters, 'Waar staat de bijzondere curator in het huidige rechtsbestel inmiddels?', FJR, 2012/36.

Raitt 2004

F.E. Raitt, 'Judicial discretion and methods of ascertaining the views of a child', *Child & Family Law Quarterly*, Vol. 16, No. 2, 2004, pp. 151-164.

Rešetar & Emery 2008

B. Rešetar & R.E. Emery, 'Children's Rights in European Legal Proceedings: Why Are Family Practices So Different From Legal Theories?', *Family Court Review*, Vol. 46, No. 1, 2008, pp. 65-77.

Rongé 2008

J-L. Rongé, 'Réflexions: La Chartre nationale de la defense des mineurs', *Journal du droit des jeunes*, No. 275, 2008, pp. 45-46.

Ross 2013a

N.M. Ross, 'Children's Lawyers: "Seeing" Children', *New Zealand Law Review*, Vol. 26, No. 3, 2013, pp. 409-427.

Ross 2013b

N.M. Ross, 'Different Views? Children's Lawyers and Children's Participation in Protective Proceedings in New South Wales, Australia', *International Journal of Law, Policy and the Family*, Vol. 27, No. 3, 2013, pp. 332-358.

Scherpe 2016

J.M. Scherpe, 'European family law – Introduction to the book set', in: J.M. Scherpe (ed.), *European Family Law Volume I: The Impact of Institutions and Organisations on European Family Law*, Cheltenham: Edward Elgar 2016, pp. ix-xii.

Schrama 2015

W.M. Schrama, 'Commentaar op artikel 212 Boek 1 BW', in: S.F.M. Wortmann (ed.), *Groene Serie Personen- en Familierecht*, Deventer: Kluwer 2015.

Shackel 2016

R. Shackel, 'The UN Convention on the Rights of the Child: Tracing Australia's Implementation of the Provisions Relating to Family Relations', in: O. Cvejić Jančić (ed.), *The Rights of the Child in a Changing World: 25 Years after The UN Convention on the Rights of the Child*, Dordrecht: Springer 2016, pp. 37-60.

Shier 2001

H. Shier, 'Pathways to Participation: Openings, Opportunities and Obligations: A New Model for Enhancing Children's Participation in Decision-making, in line with Article 12.1 of the United Nations Convention on the Rights of the Child', *Children & Society*, Vol. 15, 2001, pp. 107-117.

Skelton et al. 2010

A. Skelton et al., *Family Law in South Africa*, Cape Town: Oxford University Press South Africa 2010.

Sloth-Nielsen 1996

J. Sloth-Nielsen, 'The contribution of children's right to the reconstruction of society: Some implications of the constitutionalisation of children's right in South Africa', *The International Journal of Children's Rights*, Vol. 4, No. 4, 1996, pp. 323-344.

Sloth-Nielsen 2002

J. Sloth-Nielsen, 'Children's Rights in the South African Courts: An overview since ratification of the UN Convention on the Rights of the Child', *The International Journal of Children's Rights*, Vol. 10, No. 2, 2002, pp. 137-156.

Sloth-Nielsen 2008

J. Sloth-Nielsen, 'Realising children's rights to legal representation and to be heard in judicial proceedings: an update', *South African Journal on Human Rights*, Vol. 24, No. 3, 2008, pp. 495-524.

Smits 2015

V.M. Smits, 'Participatie van het kind bij het ouderschapsplan: een tweeledig onderzoek naar de werking van artikel 815 lid 4 Rv', *FJR*, 2015/44.

Strutz & Verhagen 2015

M. Strutz & E. Verhagen, 'The Minor in Divorce-Related Judicial Proceedings in the Netherlands and Germany', *European Journal of Law Reform*, Vol. 17, No. 2, 2015, pp. 305-316.

Sutherland 2012

E.E. Sutherland, 'Imperatives and challenges in child and family law: Commonalities and disparities', in: E.E. Sutherland (ed.), *The Future of Child and Family Law: International Predictions*, New York: Cambridge University Press 2012, pp. 1-46.

Sutherland 2013

E.E. Sutherland, 'Listening to the Voice of the Child: The Evolution of Participation Rights', *New Zealand Law Review*, Vol. 26, No. 3, 2013, pp. 335-355.

Sutherland 2014

E.E. Sutherland, 'Listening to the child's voice in the family setting: from aspiration to reality', *Child & Family Law Quarterly*, Vol. 26, No. 2, 2014, pp. 152-172.

Taylor, Tapp & Henaghan 2007

N. Taylor, P. Tapp & M. Henaghan, 'Respecting Children's Participation in Family Law Proceedings', *International Journal of Children's Rights*, Vol. 15, 2007, pp. 61-82.

Taylor et al. 2012

N. Taylor et al., 'International Models of Child Participation in Family Law Proceedings following Parental Separation/Divorce', *International Journal of Children's Rights*, Vol. 20, 2012, pp. 645-673.

Van Teeffelen 2007

P.A.J.Th. van Teeffelen, 'De stem van het kind in echtscheidings-, gezags- en omgangs-zaken', *FJR*, 2007/45.

Van Teeffelen 2008

P.A.J.Th. van Teeffelen, 'FJR-Studiemiddag: bijzondere curator', FJR, 2008/34.

Van Teeffelen 2013

P.A.J.Th. van Teeffelen, 'Formele en informele rechtsingang(en) minderjarige', EB, 2013/62.

Thomas 2007

N. Thomas, 'Towards a Theory of Children's Participation', *International Journal of Children's Rights*, Vol. 15, No. 2, 2007, pp. 199-218.

Thomas & O'Kane 1998

N. Thomas & C. O'Kane, 'When children's wishes and feelings clash with their 'best interests'', *The International Journal of Children's Rights*, Vol. 6, 1998, pp. 137-154.

Tisdall & Morrison 2012

E.K.M. Tisdall & F. Morrison, 'Children's Participation in Court Proceedings when Parents Divorce or Separate: Legal Constructions and Lived Experiences', in: M. Freeman, *Law and Childhood Studies*, Oxford: Oxford University Press 2012, pp. 156-173.

Tisdall et al. 2004

E.K.M. Tisdall et al., 'Children's Participation in Family Law Proceedings: A Step Too Far or a Step Too Small', *Journal of Social Welfare and Family Law* Vol. 26, No. 1, 2004, pp. 17-33.

Treseder 1997

P. Treseder, *Empowering children and young people: training manual; promoting involvement in decision-making*, London: Save the Children 1997.

Van Triest 2004

A. van Triest, 'Het kinderverhoor in het ressort Den Bosch onder de loep', FJR, 2004/4.

Veegens 1923

J.D. Veegens, 'Schets van het Nederlandsch Burgerlijk Recht: Eerste Deel Personenrecht', 3rd edition (A.S. Oppenheim ed.), Haarlem: H.D. Tjeenk Willink & Zoon 1923.

Vlaardingerbroek 2001

P. Vlaardingerbroek, 'De bijzondere curator en het afstammingsrecht', FJR, No. 4, 2001, pp. 101-107.

Van Wamelen

C. van Wamelen, 'De Omgangscurator', FJR, No. 10, 1995, pp. 217.

Zaal & Skelton 1998

N. Zaal & A. Skelton, 'Providing Effective Representation for Children in a New Constitutional Era: Lawyers in the Criminal and Children's Courts', *South African Journal on Human Rights*, Vol. 14, No. 4, 1998, pp. 539-559.

Other sources

Hale 2011

B. Hale, 'Can you hear me, Your Honour?', Hershman Levy Memorial Lecture 2011, available at: www.alc.org.uk/uploads/Can_you_hear_me,_Your_Honour._Memorial_lecture_2011.pdf (last visited 19 March 2017).

Ligue des droits de l'Homme 2012

Ligue des droits de l'Homme, 'La 99rimordi de l'avocat est 99rimordial aux côtés de l'enfant', 19 April 2012, available at: www.ldh-france.org/La-presence-de-l-avocat-est/ (last visited 19 March 2017).

5. Annexes

- 5.1. Australia (New South Wales)
 - 5.1.1. What forms of representation are available for children in family law proceedings in Australia (New South Wales) and how are they regulated?

Four forms of representation are available for children in family law proceedings in Australia, more specifically in the state New South Wales. Federal law provides for the independent children's lawyer in Section 68L of the Family Law Act 1975 (hereafter AFLA 1975). The independent children's lawyer represents the child's interests in the legal proceedings.³⁵⁸ This form of best interests representation is similar to the independent legal representative in New South Wales provided for in s. 99A(2) of the Children and Young Persons (Care and Protection) Act 1998 (NSW) (hereafter CCPA (NSW)) and s. 122(2)(b) of the Adoption Act 2000 (NSW).³⁵⁹ In New South Wales, children can instruct a direct legal representative following s. 99A(1) CCPA (NSW) and s. 122(2)(b) of the Adoption Act 2000, who will act on their instructions.³⁶⁰ Lastly, a Guardian ad Litem can be appointed for the child following s. 100 CCPA (NSW) and s. 123 of the Adoption Act 2000. The guardian ad litem is a 'person appointed by the court to make decisions on behalf of a child in the legal proceedings'.³⁶¹

5.1.1.1. When were the forms of representation introduced or amended?

Most of the forms of representation in Australia are relatively young, direct representation of children since the 1970s and best interests representation in the following decades.³⁶² Only the Guardian ad Litem has historical roots in English common law which was the basis of Australian family law prior to the Family Law Act of 1975.³⁶³

Fifteen years prior to Australia's ratification of the UNCRC, s. 65 of the original Family Law Act 1975 established a statutory form of child representation in the line of Article 12 UNCRC.³⁶⁴ As Part VII of the AFLA 1975 was extremely brief, comprising of only 10 pages, there was no guidance regarding the role of the child's representative.³⁶⁵ Following the ratification of the UNCRC the role of this 'separate representative' was developed and defined by case law, especially in the 1990s, for example in In the Matter of P and P and Re K.³⁶⁶ In the latter case the Family Court also acknowledged the role of the UNCRC in developing the guidelines on the appointment of separate

³⁵⁸ Barrie 2013, p. 128.

³⁵⁹ Representation Principles for Children's Lawyers 2014, p. 8.

³⁶⁰ Fernando 2013, p. 395.

³⁶¹ Representation Principles for Children's Lawyers 2014, p. 8.

 ³⁶² Ross 2013a, p. 411.
 ³⁶³ Australian Law Reform Commission 1997, para. 13.10; and Bates 2013, p. 48.

³⁶⁴ McIntosh, Bryant & Murray 2008, p. 128.

³⁶⁵ McIntosh, Bryant & Murray 2008, p. 128.

³⁶⁶ In the Matter of P and P [1995] FLC 92-615 and Re K [1994] FLC 92-461. See also, Inquiry into the Australian Legal Aid System 1997, para. 5.23; Carson et al. 2014, p. 59; and Kaspiew et al. 2013, p. 4.

representatives.³⁶⁷ In 2004, the Family Law Council published their 'Pathways for Children: A Review of Child Representation in Family Law' report in response to a Federal Government directive.³⁶⁸ While the report did not recommend any major changes to the role of child representatives as developed by case law,³⁶⁹ it did recommend the clarification and consolidation of best interests representatives as independent children's lawyers through further legislative recognition.³⁷⁰ Essentially, the Family Law Council recommended the codification of the pre-existing practices in case law.³⁷¹ These recommendations were to a great extent adopted and implemented into the Family Law Act by the Family Law Amendment (Shared Parental Responsibility) Act 2006.³⁷² This Amendment Act drastically extended Part VII of the AFLA 1975 to approximately 200 pages, amongst others by introducing the independent children's lawyer with a best interests role and clear statutory guidelines.³⁷³ Part VII was further amended in 2011, when the fulfillment of Australia's obligations as a signatory to the UNCRC was introduced as an additional object of the Part in s. 60(B)(4).³⁷⁴ This reaffirms how the introduction of the independent children's lawyers is an important aspect of Australia fulfilling the obligations under Article 12 UNCRC.³⁷⁵

In New South Wales, the representation of children in care proceedings was introduced by the Children (Care and Protection) Act 1987 (NSW), as the previous act, the Child Welfare Act 1939 (NSW) made no mention of represented children. Section 66(2) of the Children (Care and Protection) Act 1987 allowed for the child to be separately represented without offering any further guidance regarding the role of such a representative. It was the Children and Young Persons (Care and Protection) Act 1998 which further expanded and explicitly introduced the direct legal representative and the independent legal representative in s. 99A. The new act, coming eight years following the ratification of the UNCRC, also incorporates the child's right to participate as a fundamental principle in s. 10.³⁷⁶ In doing so, the Act has strengthened children's participation rights in child care proceedings as well as all other decisions made within the framework of the Act, for example by social workers.³⁷⁷

5.1.1.2. In which types of cases can they be represented?

The independent children's lawyers can be appointed in federal family law proceedings under the Family Law Act 'in which a child's best interests are, or a child's welfare is, the paramount, or

³⁶⁷ McIntosh, Bryant & Murray 2008, p. 128.

³⁶⁸ Pathways for Children 2004 and Bell 2015, p. 3.

³⁶⁹ Pathways for Children 2004 and Bell 2015, p. 3.

³⁷⁰ Pathways for Children 2004, p. 29 and 43; Bell 2015, p. 3; and Kaspiew et al. 2013, p. 4.

³⁷¹ Kaspiew et al. 2013, p. 4. ³⁷² Bell 2015, p. 3.

 ³⁷³ Barrie 2013, p. 125-127.
 ³⁷⁴ Family Law Legislation Amendment (Family Violence and Other Measures) Act 2011. See also, Carson et al. 2014, p. ^{58.} ³⁷⁵ Kaspiew et al. 2013, p. xi.

³⁷⁶ Ross 2013b, p. 334.

³⁷⁷ Parkinson 2001, p. 260.

relevant, consideration' per s. 68L AFLA 1975.³⁷⁸ Originally, the role was limited to 'proceedings with respect to the custody, guardianship or maintenance of, or access to, a child of a marriage' (s. 65 (old) AFLA 1975). However, later amendments increased the scope of the independent children's lawyer's role.³⁷⁹ Now the types of cases include proceedings on parental authority (s. 61A to 61F), parental plans (s. 63A to 63H), parenting orders (s. 64A to 65ZD), child maintenance orders (s. 66A to 66X), and parentage proceedings (s. 69P to 69ZD). Proceedings concerning an international child abduction are also included, albeit that s. 68L(3) applies further requirements for the appointment of independent children's lawyer in those cases. The scope is also no longer limited to children of a marriage, but also includes cases concerning children from *de facto* partners, children born as a result of artificial conception procedures or surrogacy agreements, and adopted children (see s. 60EA to 60HB).

In New South Wales, the direct legal representative, independent legal representative, and Guardian ad Litem can be appointed in child protection proceedings and in adoption proceedings.³⁸⁰ In child protection proceedings, the participation principle of s. 10 CCPA (NSW) explicitly lists certain decisions which will likely have a significant impact on the child's life and thus especially require representation. The list names proceedings concerning emergency or ongoing care, the development and review of care plans, the provision of counselling or treatment services and contact with family members.³⁸¹ In adoption proceedings, the participation principle of s. 9 Adoption Act 2000 also lists decisions of significant impact, which are proceedings related to the placement for adoption of the child, the development of adoption plans, the application for the adoption order, and related to contact with birth parents of the child.

5.1.1.3. When can children be represented in family law proceedings in Australia (New South Wales)?

Independent children's lawyers

Independent children's lawyers can be, and often are, appointed at an early stage in the family law proceedings.³⁸² This why they also receive legal aid funding for the first stage of proceedings in most States and Territories until the end of proceedings, if not for post-decision orders.³⁸³ For example, in New South Wales, the independent children's lawyer is funded for three stages: 1) the procedural hearing to the listing for hearing, 2) the preparation for the final hearing, 3) the final hearing (and potentially post-decision orders).³⁸⁴ According to the Federal Guidelines for independent children's lawyers, the representation can end either because the decision has been made or because the

³⁷⁸ Kaspiew et al. 2013, p. xi.

³⁷⁹ See also, Inquiry into the Australian Legal Aid System 1997, para. 5.5. ³⁸⁰ Bell 2015, p. 2; and Ross 2013b, p. 332.

³⁸¹ S. 10(3) CCPA (NSW). See also Parkinson 2001, p. 260.

³⁸² Kaspiew et al. 2013, p. 18.

³⁸³ Kaspiew et al. 2013, p. 19.

³⁸⁴ Kaspiew et al. 2013, p. 19.

independent children's lawyer has sought the court to discharge the appointment. The latter can be done because the independent children's lawyer is of the opinion that he or she no longer serves a useful purpose, the relationship with the child has broken down irretrievably, the representation is contrary to the best interests of the child, or because practical circumstances make continuation impossible.³⁸⁵

Direct legal representative and independent legal representative

The direct legal representative and independent legal representative will generally be appointed at the start of the child protection or adoption proceedings and will represent the child throughout the course of the proceedings.³⁸⁶ Direct legal representatives can be dismissed by the children themselves at any time and regardless of who appointed the representative.³⁸⁷ The court should be informed hereof by the representative. Independent legal representatives cannot be dismissed by the children themselves.³⁸⁸ Following s. 99(3) CCPA (NSW) both types of legal representatives may be dismissed by the court at any time and for any reason, in adoption proceedings only for the reason that 'the child informs the Court that he or she does not wish to be represented' (s. 122(9) Adoption Act 2000).

Guardian ad Litem

A Guardian ad Litem will most likely also be appointed in the early phases of a proceeding, although it is not exactly clear. It is also not clear when the Guardian ad Litem's standing terminates, however case law has shown that it can continue after the final decision, because the Guardian ad Litem may commence an appeal.³⁸⁹ What is clear is that the appointment of a Guardian ad Litem should not be done too hastily, because once a Guardian ad Litem is appointed the child can no longer instruct a direct legal representative.³⁹⁰ That is also why an order for a Guardian ad Litem is rarely made, because as explained by Hamilton J in Re Oscar, it is generally sufficient and less intrusive for the interests of the child to be protected by a direct or independent legal representative.³⁹¹

5.1.1.4. What requirements are set for the children, e.g. age, level of maturity?

Independent children's lawyer

In the AFLA 1975 no requirements are set regarding the children for the appointment of an independent children's lawyer. However, the order that the court can make under s. 5 for the independent children's lawyer to find out the child's views on the matter can be set aside by the independent children's lawyer if complying with the order would be inappropriate due to the child's

³⁸⁵ Guidelines for Independent Children's Lawyers 2013, para. 6.6.

 ³⁸⁶ Care and Protection Practice Standards 2015, principles 2.2.2(1), 2.2.4, 2.3.2(1), and 2.3.4.
 ³⁸⁷ Principle J2 of the Representation Principles for Children's Lawyers 2014.

³⁸⁸ See the explanatory text relating to Principle J2 of the Representation Principles for Children's Lawyers 2014.

³⁸⁹ Bao-Er 2006, p. 5.

³⁹⁰ Bao-Er 2006, p. 2.

³⁹¹ Re Oscar [2002] NSWSC 453 (21 May 2002), para. 7; and Bao-Er 2006, p. 2.

age or maturity (s. 6(a) AFLA 1975). No information is given either in the statute or in the Federal Guidelines when this would be the case.

Direct legal representative and independent legal representative

In New South Wales, requirements are set regarding the child for the appointment of direct and independent legal representatives both in child protection proceedings and adoption proceedings.

In child protection proceedings, the age and/or capability of the child determines what form of representative is appointed, if a child is presumed capable of giving proper instructions a direct legal representative is appointed, if not an independent legal representative.³⁹² Two rebuttable presumptions apply. According to s. 99B CCPA (NSW), children are not presumed to be capable of giving proper instructions under the age of 12. From the age of 12 and over, children are presumed capable of giving instructions and this presumption is not simply rebutted if the child has a disability (s. 99C CCPA (NSW)).

In adoption proceedings, the age and/or capability of the child plays a similar role with regards to the form of representative appointed.³⁹³ Section 122(4) of the Adoption Act 2000 also includes a rebuttable presumption, but then that a child from the age of 10 and over is capable of giving proper instructions and that it may not be rebutted simple because the child has a disability. More importantly in adoption proceedings, the consent of the child is required if the child is between 12 and 18 years of age according to s. 55 of the Adoption Act 2000. However, this consent may be dispensed by the court, following s. 69, if the child is 'in such a physical or mental condition as not to be capable of properly considering the question of consent'.

The rebuttable assumptions in both child protection and adoption proceedings raise the question as to how the child's capability of giving proper instructions should be determined. The NSW Representation Principles for Children's Lawyers give further guidance. The determination of capability should be based on 'the child's willingness to participate and ability to communicate' not 'any assessment of the 'good judgment' or level of maturity of the child'.³⁹⁴ It must be determined per child and the representative should consider whether 'a perceived incapacity could be overcome by developmentally appropriate communication or by adopting a different approach in taking instructions'.³⁹⁵ More generally the representative should boost the child's capacity by taking into account the 'child's age, level of education, cultural context and degree of language acquisition' when communicating with the child.³⁹⁶ However, if the capacity of the child to give proper instructions is in fact limited, but the child can give limited instructions then the representative should directly represent the child only regarding those matters, or if the child has a disability, then the representative

³⁹² S. 99A(1)(a) and S. 99A(2)(a) CCPA (NSW). See also Ross 2013b, p. 334.

 $^{^{393}}$ S. 122(3)(c)(i) Adoption Act 2000 requires an independent legal representative is the child is not capable of giving proper instructions, otherwise a direct legal representative suffices.

³⁹⁴ Principle C1 of the Representation Principles for Children's Lawyers 2014.

³⁹⁵ Explanation following Principle C1 of the Representation Principles for Children's Lawyers 2014.

³⁹⁶ Principle C2 of the Representation Principles for Children's Lawyers 2014.

should 'seek help from appropriate service providers' to enhance the child's capacity to give instructions.³⁹⁷

Guardian ad Litem

In regards to the appointment of a Guardian ad Litem in New South Wales one of the main requirements, that there are special circumstances which warrant the appointment, are factors concerning the child. Section 123(2) of the Adoption Act 2000 states that special circumstances may include that the child has special needs because of age, disability or illness. In s. 100(2) CCPA (NSW), the exact same thing is stated with the addition of the special circumstance that the child is not capable of giving proper instructions.

5.1.1.5. What other requirements are applied, e.g. conflict of views between child and parents?

An independent children's lawyer will not be appointed automatically in every case, but instead the court will only do so if it is of the opinion that the child's interests 'ought to be independently represented' in the proceedings, according to s. 68L(2) AFLA 1975.³⁹⁸ The Australian courts make use of the non-exhaustive criteria provided in *Re K* to assist them in deciding whether or not to use their discretion to appoint an independent children's lawyer.³⁹⁹ If the cases are relatively straightforward, courts will generally not appoint an independent children's lawyer.⁴⁰⁰ In Re K, a decision from 1994, the Full Court determined thirteen grounds in which a child's interests will normally require an independent children's lawyer.⁴⁰¹ The thirteen grounds are, that there is a proceeding involving: 1. Allegations of child abuse, 2. An apparently intractable conflict between the parents, 3. Parental alienation, 4. Issues of cultural or religious differences affecting the child, 5. Issues concerning the sexual preference of the parent(s) which is likely to impinge upon the child's welfare, 6. Alleged anti-social conduct by the parent(s) which seriously impinges on the child's welfare, 7. Significant medical, psychiatric or psychological illness or personality disorder in relation to the parent(s) or child, 8. Parents who neither seem suitable custodians, 9. A child of mature years expressing strong views, which involve changing a long standing custodial arrangement or a complete denial of access to one parent, 10. The potential permanent removal of the child from the jurisdiction, 11. A proposal to separate siblings, 12. Custody cases where none of the parties are legally represented, 13. Applications relating to the medical treatment of children. Australian access to justice

³⁹⁷ Principle C3 of the Representation Principles for Children's Lawyers 2014.

³⁹⁸ Fernando 2014, p. 48. ³⁹⁹ *Re K* [1994] FLC 92-461; Inquiry into the Australian Legal Aid System 1997, para. 5.16; Carson et al. 2014, p. 59; and Kaspiew et al. 2013, p. xi. ⁴⁰⁰ Fernando 2014, p. 48.

⁴⁰¹ Re K [1994] FLC 92-461. See also, Inquiry into the Australian Legal Aid System 1997, para. 5.15 for the full and extensive list of grounds.

research shows that independent children's lawyer s are especially appointed when the first, third, sixth or seventh ground is involved.⁴⁰²

With regards to international child abduction cases, s. 68L(3) AFLA 1975 places an additional requirement on the appointment of an independent children's lawyer. The court may only do so if it considers that there are 'exceptional circumstances that justify doing so'. Although in a previously leading case, *De L* the High Court of Australia found that separate representation should always be afforded to children in international abduction cases,⁴⁰³ the effect of the decision was reversed in the Family Law Amendment Act 2000 by including para. 3 to s. 68L AFLA 1975. The explanation to do so was that the Hague Convention requires a prompt return without an inquiry into 'the reasons for the abduction' or 'the best interests of the child' and thus no separate representative is required unless in exceptional circumstances.⁴⁰⁴ The requirement of 'exceptional circumstances' has been criticized recently, including by the Chief Justice of the Family Court, as it 'may now be too restrictive'.⁴⁰⁵

In New South Wales, the CCPA (NSW) and Adoption Act 2000 apply the same relatively vague requirement for appointing either a **direct legal representative** or **independent legal representative**. Following s. 99(1) CCPA (NSW) and s. 122(2)(b) Adoption Act 2000, the court may appoint a representative 'if it appears to the court that the child needs to be represented in any proceedings before it'. There is no further specification as to when this is the case, but it will probably at least be so when any of the thirteen grounds of *Re K* are involved. With regards to the **Guardian ad Litem**, s. 100(1)(B) CCPA (NSW) and s. 123(1)(B) of the Adoption Act 2000 again apply the same requirement as a separate option to the child-related factor discussed above of 'special circumstances', namely 'if the court is of the opinion that the child will benefit from the appointment'. This is likewise a vague requirement, but as an example, in *Re Oscar* the court was of the opinion that the child would benefit from a Guardian ad Litem because the child was almost 12, so his wishes were a relevant consideration, but he was (likely) suffering serious developmental impairment or psychological harm because of the particularly bitter and complicated history of the matter.⁴⁰⁶

5.1.1.6. Who decides whether the child will be represented?

The decision whether the child will be represented is made by the courts in Australia. An **independent children's lawyer** can be appointed by the court either on its own initiative or on the

⁴⁰² Kaspiew et al. 2013, p. 5.

⁴⁰³ De L [1996] FLC 92-706.

⁴⁰⁴ Further Revised Explanatory Memorandum Family Law Amendment Bill 2000, para. 293.

⁴⁰⁵ Senate Legal and Constitutional Affairs Committee 2011, p. 42; and Kaspiew et al. 2013, p. 5.

⁴⁰⁶ *Re Oscar* [2002] NSWSC 453 (21 May 2002), para. 4 and 7.
application of the child, an organization concerned with the welfare of the child or any other person, following s. 68L(4) AFLA 1975.407

In New South Wales, a direct legal representative or independent legal representative may either be appointed by the court or requires the leave of court if retained directly by the child or the parents or guardians of the child.⁴⁰⁸ A **Guardian ad Litem** can only be appointed by the court, see s. 100(1) CCPA (NSW) or s. 123(1) of the Adoption Act 2000. If a representative was first appointed for child, but they determine that the child is unable to instruct the representative, then the representative should inform the court and apply for an appointment of a Guardian ad Litem for the child, whose instructions the representative can then act on.⁴⁰⁹

5.1.1.7. How is the child's representative financed?

Legal aid is a matter left to the Legal Aid commissions in each State or Territory. In principle, the costs of a representative is either borne by the parties equally or by a legal aid grant.⁴¹⁰ Speaking generally, across Australia the Legal Aid Commissions will mostly provide legal aid for independent children's lawyers if appointed by the court.⁴¹¹ In New South Wales, it depends on the type of family law proceeding whether legal aid is available and which test must be satisfied. Legal aid for the independent children's lawyer in proceedings on parenting matters or parenting orders is granted if the court has appointed the lawyer and legal aid is considered reasonable, no means or merits test is applied.⁴¹² In proceedings on child maintenance or international child abductions a means and merits test is applied.⁴¹³ With regard to the direct legal representative or independent legal **representative**, legal aid is provided in care proceedings without a means or merits test,⁴¹⁴ but is only provided in adoption proceedings in exceptional circumstances in addition to the means or merits test.⁴¹⁵ If legal aid is not provided, independent children's lawyer, direct and independent legal representatives, must be paid for by the parties, the parents, equally, taking into consideration their capacity to pay, their legally aided status and previous contributions.⁴¹⁶

The Guardian ad Litem is appointed in New South Wales from the Guardian ad Litem Panel which falls under the responsibility of the Department of Justice. This means that when a Guardian ad Litem is appointed he or she is paid an allowance in accordance with the scheduled fee through the Guardian ad Litem Panel.⁴¹⁷

⁴⁰⁷ Barrie 2013, p. 128.

⁴⁰⁸ See the explanatory text accompanying Principle A1 and Principle A2 of the Representation Principles for Children's Lawyers 2014. ⁴⁰⁹ Principle B2 and the accompanying explanatory text of the Representation Principles for Children's Lawyers 2014.

⁴¹⁰ Parkinson & Cashmore 2008, p. 51.

⁴¹¹ Kaspiew et al. 2013, p. 17.

⁴¹² Policies Legal Aid New South Wales, 5.3.3 and 5.4.4. ⁴¹³ Policies Legal Aid New South Wales, 5.10.3 and 5.12.

⁴¹⁴ Policies Legal Aid New South Wales, 5.16.10.

⁴¹⁵ Policies Legal Aid New South Wales, 5.18.

⁴¹⁶ Guidelines Legal Aid New South Wales, 2.4.

⁴¹⁷ Guardian ad Litem Handbook 2012, para. 35.

5.1.2. What is the task of the child's representative in family law proceedings in Australia (New South Wales)?

Independent children's lawyer

Under the Family Law Act 1975, s. 68LA sets out the task of the independent children's lawyer. Independent children's lawyers represent the child's best interests in the court proceedings. They are not the child's legal representative and thus are not obliged to act on the child's instructions (s. 68LA(4) AFLA 1975).⁴¹⁸ Instead, s. 68LA(2) and (3) require that they 'form an independent view' of what is in the best interests of the child and act and/or make submissions to the court in the proceedings accordingly, including appealing the orders of the court if necessary.⁴¹⁹ To do so, the independent children's lawyer must 'act impartially' by being truly independent of the court and the parties and focusing fully and objectively on the child's best interests.⁴²⁰ The independent children's lawyer should bring before the court the child's best interest as drawn from and supported by evidence, not from their own opinion of the case.⁴²¹ The independent children's lawyer's task also includes ensuring that any views expressed by the child (note, the child is not required to express views - s. 60CE AFLA 1975) are fully put before the court.⁴²² This does not mean that the independent children's lawyer must meet with the child themselves.⁴²³ If these views do not concur with what the independent children's lawyer has submitted as being in the best interests of the child, that is not a problem as long as it has been made clear to the court.⁴²⁴ Finally, the independent children's lawyer also has the task of acting as an 'honest broker' on behalf of the child in the facilitation of any agreements and must 'endeavor to minimize the trauma to the child associated with the proceedings'.425

Direct legal representative

The role of the direct legal representative in New South Wales is to act on the instructions of the child, who is their client, as well as ensuring that their views are placed before the court and that all relevant evidence is adduced according to s. 99D(a) CCPA (NSW) and s. 122(3).⁴²⁶ According to Parkinson, the role of a client-directed legal representative is more important in care proceedings than

⁴¹⁸ See also, Guidelines for Independent Children's Lawyers 2013, para. 2; and Fernando 2013, p. 396.

⁴¹⁹ Guidelines for Independent Children's Lawyers 2013, para. 6.11.

⁴²⁰ See, s. 68LA(5)(a) AFLA 1975 and the Guidelines for Independent Children's Lawyers 2013, para. 4 and 6.4.

⁴²¹ Guidelines for Independent Children's Lawyers 2013, para. 4.

 ⁴²² See, s. 68LA(5)(b) AFLA 1975 and the Guidelines for Independent Children's Lawyers 2013, para. 4.
 ⁴²³ Fernando 2014, p. 49.

⁴²⁴ Guidelines for Independent Children's Lawyers 2013, para. 2; and Fernando 2013, p. 395.

⁴²⁵ See, s. 68LA(5)(d) and (e) AFLA 1975 and the Guidelines for Independent Children's Lawyers 2013, para. 6.4.

⁴²⁶ See also Principle A1 of the Representation Principles for Children's Lawyers 2014; Parkinson 2001, p. 267; and Ross 2013b, p. 335.

in parental disputes, because the child protection authority will already promote the best interests of the child.427

Independent legal representative

The role of the independent legal representative in New South Wales is more akin to that of an independent children's lawyer. Independent legal representatives are best interests representatives as they do not act on a child's instructions and do not have a client,⁴²⁸ unless a Guardian ad Litem has been appointed, in which case the independent legal representative acts on their instructions (s. 99D(b)(i) CCPA (NSW) and s. 123(4) Adoption Act 2000).⁴²⁹ Their role is thus to present evidence to the court on what is in the best interests of the child and to make applications, submissions or appeals accordingly, but also to present the child's views (s. 99D(b) CCPA (NSW) and s. 123(7) Adoption Act 2000).

Guardian ad Litem

The role of the Guardian ad Litem in New South Wales is to 'safeguard and represent the interests of the child' and to instruct the legal representative of the child, s. 100(3) CCPA (NSW) and s. 123(3) Adoption Act 2000. This role is relatively vague as no further statutory guidance is provided.⁴³⁰

5.1.2.1. What are the function requirements for the child's representative?

Independent children's lawyers, direct legal representatives, and independent legal representatives are all legal practitioners who are either 'inhouse', employed directly by the State's Legal Aid Commission, or work privately but are appointed to a panel from which they can be appointed.⁴³¹ In either case it is required that the practitioner has a minimum of 5 years experience in family law and, for independent children's lawyers, have completed the independent children's lawyer training program.⁴³² In New South Wales, once practitioners have been appointed to the panel they must agree and adhere to the practice standards.⁴³³ These practice standards require that the lawyers must be able to communicate effectively with children and must have an understanding of issues facing children in proceedings.⁴³⁴ New South Wales also has a professional development program available to all independent children's lawyers.⁴³⁵

⁴²⁷ Parkinson 2001, p. 268.

⁴²⁸ Principle A2 of the Representation Principles for Children's Lawyers 2014; and Ross 2013, p. 335.

⁴²⁹ Principle A3 of the Representation Principles for Children's Lawyers 2014.

⁴³⁰ Bao-Er 2006, p. 1.
⁴³¹ Kaspiew et al. 2013, p. 15; and Ross 2013b, p. 335.

⁴³² Carson et al. 2014, p. 60.

⁴³³ Kaspiew et al. 2013, p. 103.

⁴³⁴ Care and Protection Practice Standards 2015, para. 1.2.

⁴³⁵ Kaspiew et al. 2013, p. 107.

Guardians ad Litem in New South Wales are generally not lawyers, instead they are persons with qualifications and experience in social, health or behavioural sciences. People can apply to become a member of the Guardian ad Litem panel, if they have 'mediation, advocacy and decision making skills, the ability to communicate effectively with various professionals and family members, basic knowledge of legal proceedings and the legal process, knowledge of issues affecting children and young people, people with illness, disability or disorder which may affect their decision-making capacity'.⁴³⁶

5.1.2.2. How should the child's representative complete their task?

Independent children's lawyer

The Family Court's Guidelines for Independent Children's Lawyers provide guidance as to how independent children's lawyer ought to complete their task.⁴³⁷ Some of the most important guidelines will be briefly touched upon. Firstly, while the independent children's lawyer is not required to meet with the child, they are encouraged and expected to, unless exceptional circumstances should prevent them from doing so.⁴³⁸ Secondly, the independent children's lawyer should provide and explain certain information to the child concerning their role (including the limitations, e.g of confidentiality), the court processes, if they have made submissions contrary to the child's views why they have done so, and at the conclusion of the proceedings the orders made by the court and their effect.⁴³⁹ Thirdly, the independent children's lawyer should strive for a relationship of trust and respect in which the child is provided the opportunity to express his or her views, which will subsequently be fully put before the court, even if they disagree that the child's views are in his or her best interests.⁴⁴⁰ The independent children's lawyer should also draw evidence to the court's attention concerning the child's best interests, but also if there is a situation of family violence and abuse.⁴⁴¹

Direct legal representative and independent legal representative

In New South Wales, the Representation Principles for Children's Lawyers and the Care and Protection Practice Standards give further guidance as to how direct and independent legal representatives should fulfill their task. A few general guidelines apply. The representative should see the child as soon as possible before going to court, in a place and at a time when it is 'comfortable and convenient' for the child, if necessary with the support of a trusted adult, as well as after any hearings or court orders.⁴⁴² The representative should use 'language appropriate to the age, maturity, level of

⁴³⁶ Guardian ad Litem Handbook 2012, para. 12.

⁴³⁷ Carson et al. 2014, p. 60.

⁴³⁸ Guidelines for Independent Children's Lawyers 2013, para. 6.2; Carson et al. 2014, p. 60; and Fernando 2014, p. 49.

⁴³⁹ Guidelines for Independent Children's Lawyers 2013, para. 5.1, 5.2, 5.4 and 6.10.

⁴⁴⁰ Guidelines for Independent Children's Lawyers 2013, para. 5.1, 5.3, and 5.4.

⁴⁴¹ See s. 68LA(5)(c) and Guidelines for Independent Children's Lawyers 2013, para. 7.

⁴⁴² Principle D1, D4 and D5 of the Representation Principles for Children's Lawyers 2014; and Care and Protection Practice Standards 2015, para. 2.2.3 and 2.3.3.

education, cultural context and degree of language proficiency of the child' when communicating with the child,⁴⁴³ if necessary employing an interpreter to do so.⁴⁴⁴ The representative should inform and explain to the child what the role of the representative is, the proceedings, and the confidential nature of the relationship and should also prepare the child for the end of the relationship.⁴⁴⁵ The representative should in this communication also encourage the child to ask further questions and answer them appropriately.⁴⁴⁶

Direct legal representative

The direct legal representative has to maintain a lawyer-client relationship with the child.⁴⁴⁷ This means that the representative should meet with the child and maintain in contact with them throughout the proceedings,⁴⁴⁸ should represent the child competently and professionally,⁴⁴⁹ and owes the same duty of confidentiality as if the child were an adult.⁴⁵⁰ Besides the general information requirements, the direct legal representative should ensure that the child is sufficiently informed of the nature of the proceedings, the available options, possible consequences, the lawyer-client relationship, and all the relevant documents.⁴⁵¹ Besides informing the child, the direct legal representative should also seek the child's instructions on those matters and should ensure that they understand these instructions.452

Independent legal representative

The independent legal representative must retain a relationship with the child, to provide the child with information and to 'explore' the child's views and the extent to which the child can participate in the proceedings.⁴⁵³ In that respect the independent legal representative also has a duty of confidentiality.⁴⁵⁴ The independent legal representative's task to represent the child's best interests must be done 'in a competent and professional way'.⁴⁵⁵ What does this mean? Firstly, that the best interests must be determined on the basis of 'objective criteria addressing the child's specific needs and preferences' and that the representative should aim at achieving the child's long term best

⁴⁴³ Principle D6 of the Representation Principles for Children's Lawyers 2014.

⁴⁴⁴ Care and Protection Practice Standards 2015, para. 2.2.3 and 2.3.3.

⁴⁴⁵ Principle D3, E1, F2 and J1 of the Representation Principles for Children's Lawyers 2014.

⁴⁴⁶ Care and Protection Practice Standards 2015, para. 2.2.3 and 2.3.3.

⁴⁴⁷ Principle D2 of the Representation Principles for Children's Lawyers 2014; and Care and Protection Practice Standards 2015, para. 2.2.2.

⁴⁴⁸ Care and Protection Practice Standards 2015, para. 2.2.2.

⁴⁴⁹ Principle E2 (part 1) of the Representation Principles for Children's Lawyers 2014.

⁴⁵⁰ Principle F1 of the Representation Principles for Children's Lawyers 2014.

⁴⁵¹ Principle D2 of the Representation Principles for Children's Lawyers 2014; and Care and Protection Practice Standards 2015, para. 2.2.3. ⁴⁵² Principle D2 of the Representation Principles for Children's Lawyers 2014; and Care and Protection Practice Standards

^{2015,} para. 2.2.3. ⁴⁵³ Care and Protection Practice Standards 2015, para. 2.3.2 and 2.3.3.

⁴⁵⁴ Principle F1 of the Representation Principles for Children's Lawyers 2014.

⁴⁵⁵ Principle E2 (part 2) of the Representation Principles for Children's Lawyers 2014.

interests through the court's decision.⁴⁵⁶ Secondly, representative should do so by participating in all court proceedings, obtaining relevant materials, presenting evidence, and ensuring that the child's views are included in the proceedings.⁴⁵⁷

Guardian ad Litem

There are no clear guidelines as to have a Guardian ad Litem should fulfill their task, however the Guardian ad Litem takes the position of a party and should thus act in that manner.⁴⁵⁸ The case of *Re* Oscar has also clarified that the Guardian ad Litem can, amongst other things, review the materials in the proceedings and interview the child.⁴⁵⁹ The Guardian ad Litem should also give instructions to the legal representative on behalf of the child.

5.1.2.3. Is the child representative the only option to be heard, are other options available and can they be complementary?

With regard to the previously discussed forms of child representation in Australia, specifically New South Wales, a few comments can be made with regards to complementarity. An independent children's lawyer cannot act complementary to a direct or independent legal representative or to a Guardian ad Litem, because the independent children's lawyer is provided for in the Family Law Act 1975 for certain types of family law proceedings which are not provided for in the New South Wales legislation on care or adoption proceedings. In these care or adoption proceedings, a direct legal representative and independent legal representative can also not act in the same proceedings. The child will have either or, according to Principle B4 a representative cannot act 'simultaneously' as both types of representative, if the need arises for a change of form of representation the court should be informed.⁴⁶⁰ However, the forms of legal representative can act together with a Guardian ad Litem. In care proceedings, the independent legal representative acts on the instructions of the Guardian ad Litem (s. 99D(b)(i) CCPA (NSW)), in adoption proceedings it is the direct legal representative (s. 122(3)(c)(ii) and s. 123(4) Adoption Act 2000).

Regarding the other options for the child to be heard, it is important to realize that a representative is only one of the three co-existing mechanisms, together with the family report and the judicial interviews, by which the child's views can be heard in family law proceedings under the Family Law Act 1975.⁴⁶¹ Family reports are written by family consultants, who are social workers or

⁴⁵⁶ Principle B3 and the explanatory text accompanying Principle E2 (part 2) of the Representation Principles for Children's Lawyers 2014.

 ⁴⁵⁷ Care and Protection Practice Standards 2015, para. 2.3.4.
 ⁴⁵⁸ Bao-Er 2006, p. 4.

⁴⁵⁹ *Re Oscar* [2002] NSWSC 453 (21 May 2002); and Bao-Er 2006, p. 4.

⁴⁶⁰ Representation Principles for Children's Lawyers 2014, see also Principle B2 which states that the representative can only proceed if he or she is certain of what type of representative they are. ⁴⁶¹ Kaspiew et al. 2013, p. 4.

psychological counsellors.⁴⁶² These consultants will speak to the child to ascertain their views and write a report concerning those views as well as other matters concerning the child's welfare (s. 62G AFLA 1975).⁴⁶³ The other option for the court is that a judge, or a judicial officer, may interview a child upon their discretion.⁴⁶⁴ If child expresses clear views relevant to the case during this interview, the judge may opt to order a family report.⁴⁶⁵ A family report, judicial interview and independent children's lawyer can be complementary, if necessary.

⁴⁶² Parkinson & Cashmore 2008, p. 48.
⁴⁶³ Barrie 2013, p. 129.
⁴⁶⁴ Barrie 2013, p. 133.
⁴⁶⁵ Barrie 2013, p. 129.

5.2. France

5.2.1. <u>What forms of representation are available for children in family law</u> proceedings in France and how are they regulated?

In French family law proceedings, children can be represented in two manners. The first form of representation is the institute of the *Administrateur ad hoc* (hereafter: ad hoc administrator). While there is no legal definition of an ad hoc administrator, legal doctrine defines it as 'a natural or legal person, appointed by a magistrate, who substitutes the parents in exercising the rights of the non-emancipated child, in the child's name and place within the limited assignment entrusted to him'.⁴⁶⁶ The ad hoc administrator can function in criminal, civil and administrative cases, ranging from family law proceedings, criminal proceedings with child victims to the representation of unaccompanied minor refugees.⁴⁶⁷ The second form of representation, the *avocat d'enfant* (herafter: children's lawyer), grants children the opportunity to be assisted by a lawyer in certain cases.⁴⁶⁸ The children's lawyer is not a separate institution, as such, but special guidelines have been set up for lawyers working for children.⁴⁶⁹

5.2.1.1. When were the forms of representation introduced or amended?

The **ad hoc administrator** was first introduced in 1910 in order to mitigate paternal power over children's property and was limited to issues of property.⁴⁷⁰ In 1964, Article 389-3 French Civil Code (*Code Civil;* hereafter FCC) was introduced for the ad hoc administrator in guardianship matters concerning children's property.⁴⁷¹ The **children's lawyer** was first introduced in the context of child protection proceedings in Article 1186 French Code of Civil Procedure (*Code de procédure civile;* hereafter FCCP) in 1981.⁴⁷² The role of the ad hoc administrator gradually expanded to other types of cases (e.g. for criminal proceedings involving child abuse in 1989⁴⁷³). In 1993 the role was extended to include family law proceedings of a non-patrimonial nature.⁴⁷⁴ The law which introduced Article 388-2 FCC, was also crucial in providing children with a general right to be heard by the judge or a third party in family law proceedings in Article 388-1 FCC, with or without the support of a children's lawyer.⁴⁷⁵ This 1993 law was introduced with the aim of the French government to bring

⁴⁶⁶ Favre-Lanfray 2008b, p. 11; and Fédération nationale des administrateurs ad hoc 2009, p. 10.

⁴⁶⁷ Favre-Lanfray 2008a, p. 2.

⁴⁶⁸ Avenard 2015, p. 34.

⁴⁶⁹ See, Chartre nationale de la defense des mineurs 2008.

⁴⁷⁰ Loi du 6 avril 1910 pour la bonne administration des biens des mineurs. See also, Fédération nationale des administrateurs ad hoc 2009, p. 12; and Grevot 2010, p. 12.

⁴⁷¹ Loi n°64-1230 du 14 décembre 1964 portant modification des dispositions du code civil relatives à la tutelle et à l'émancipation. See also, Fossier 2002.
⁴⁷² Décret n°81-500 du 12 mai 1981 instituant les dispositions des Livres III et IV du nouveau Code de procedure civile et

 ⁴⁷² Décret n°81-500 du 12 mai 1981 instituant les dispositions des Livres III et IV du nouveau Code de procedure civile et modifiant certaines dispositions de ce code.
 ⁴⁷³ Loi n° 89-487 du 10 juillet 1989 relative à la prévention des mauvais traitements à l'égard des mineurs et à la protection

⁴⁷³ Loi n° 89-487 du 10 juillet 1989 relative à la prévention des mauvais traitements à l'égard des mineurs et à la protection de l'enfance. See also, Grevot 2010.

⁴⁷⁴ Fédération nationale des administrateurs ad hoc 2009, p. 12; and Grevot 2010, p. 12.

⁴⁷⁵ Loi n°93-22 du 8 janvier 1993 modifiant le code civil relative à l'état civil, à la famille et aux droits de l'enfant et instituant le juge aux affaires familiales. See also, Demarchi 2010.

the French Civil Code in conformity with the UNCRC, following the ratification of the UNCRC by France in 1990.⁴⁷⁶ More amendments occurred in 2007, which introduced the obligation to inform the child on their right to be heard and to be assisted by a children's lawyer,⁴⁷⁷ and in 2009, when Articles 338-1 to 338-12 FCCP were introduced clarifying how children should be heard.⁴⁷⁸ In 2015, the (old) Article 389-3 FCC was abrogated and the content concerning the appointment of an ad hoc administrator was instead moved to an amended Article 383 FCC by an Ordonnance with the aim of modifying and simplifying family.⁴⁷⁹ Most recently, in 2016, Article 388-2 was amended to include a section on the appointment of an ad hoc administrator in child protection proceedings.⁴⁸⁰

5.2.1.2. In which types of cases can they be represented?

Children can be represented by an **ad hoc administrator** in a variety of civil, criminal and administrative law proceedings. In civil cases, the ad hoc administrator can be appointed in a variety of procedures. There are two legal grounds for civil cases, Article 383 and Article 388-2 FCC, but they are complex and to a certain extent incoherent as debate remains regarding in which types of cases an ad hoc administrator can be appointed.⁴⁸¹

Article 383 FCC is relatively clear as it allows an ad hoc administrator to represent children in proceedings concerning their property, e.g. in a succession, when there is a conflicting interest between the child and their legal representative, for example their parents.⁴⁸² It is important to realize that the ad hoc administrator is not the same as a guardian (*tutelle*), the ad hoc administrator has a precise and limited mission as a judicial representative while the guardian has a continual function as the replacement legal representative.⁴⁸³ An ad hoc administrator can replace the judicial representation of children in conflict with their guardian, however.⁴⁸⁴

Article 388-2 FCC allows for the appointment of an ad hoc administrator in many other family law matters when the interests of the child and their parent(s) conflict. However, it is debated whether the ad hoc administrator can only represent the child who is not a party to proceedings or can only represent the child who is a party to the proceedings. In principle, as emphasized by Gouttenoire, the ad hoc administrator can act in cases in which children are a party but do not have the capacity to represent themselves, thus should be represented by their legal representative, but due to a conflict of interests can or should not be.⁴⁸⁵ This would include the proceedings in which the child has the (exclusive) right to establish or contest maternity or paternity (Art. 325 and 327 jo. 328 FCC or Art.

⁴⁷⁶ Attias 2012; Bazin 2014; and Le Mintier 1994, p. 1.

⁴⁷⁷ Loi n°2007-308 du 5 mars 2007 portant réforme de la protection juridique des majeurs. See also, Demarchi 2010.

⁴⁷⁸ Décret n°2009-572 du 20 mai 2009 relatif à l'audition de l'enfant en justice. See also, Demarchi 2010.

⁴⁷⁹ Ordonnance n°2015-1288 du 15 octobre 2015 portant simplification et modernisation du droit de la famille.

⁴⁸⁰ Loi n°2016-297 du 14 mars 2016 relative à la protection de l'enfant.

⁴⁸¹ Grevot 2010, p. 13.

⁴⁸²₄₈₂ Massip 1995.

⁴⁸³ Fédération nationale des administrateurs ad hoc 2009, p. 10.

⁴⁸⁴ Favre-Lanfray 2008b, p. 11.

⁴⁸⁵ Gouttenoire 2006, p. 62.

334 FCC),⁴⁸⁶ child protection proceedings (Art. 375 FCC),⁴⁸⁷ proceedings for the revocation of a simple adoption (Art. 370-1 FCC).⁴⁸⁸ and proceedings on the grandparent's right of access and visitation (Art. 371-4 FCC).⁴⁸⁹ On the other hand, an ad hoc administrator can also be appointed in types of cases where children are not a party to the proceedings as has been implicitly accepted by the Cour de Cassation and in legal doctrine.⁴⁹⁰ While some authors still find it pointless to appoint an ad hoc administrator in cases when the child cannot intervene,⁴⁹¹ it is possible for the appointment of an ad hoc administrator in procedures concerning parental authority, e.g. the litigation of visitation, access, and custody after divorce, etc. 492

In a variety of family law matters, the children's lawyer can represent the child when the child is a party or can assist the child in cases when the child is not a party.⁴⁹³ In child protection proceedings, to which the child is a party (Art. 375 FCC), the child with discernment has the right to appoint a children's lawyer (Art. 1186 FCCP).⁴⁹⁴ The same applies to the revocation of a simple adoption (Art. 370-1 FCC). Even if the child is not a party but has the right to be heard, such as in proceedings within the framework of parental authority or international child abduction, Article 388-1 FCC allows for the child to be assisted by a children's lawyer.⁴⁹⁵

5.2.1.3. When can children be represented in family law proceedings in France?

The ad hoc administrator of Article 383 FCC can be appointed prior to the start of a proceeding, while the appointment on the basis of Article 388-2 FCC can only occur in an ongoing procedure.⁴⁹⁶ An ad hoc administrator has a precise and limited mission, which means that their functioning is limited to the time necessary to complete the mission.⁴⁹⁷ This also highlights one of the differences between the ad hoc administrator and the guardian (*tutelle*), who has a continuing function.⁴⁹⁸ There is no specific start or end to the task of the children's lawyer. They can assist and/or represent the child prior to, during and after the proceedings. The law places no restrictions, but does require that a children's lawyer appointed by the judge according to Article 1186 FCCP must be designated to the child within eight days of the appointment.

⁴⁸⁶ Favre-Lanfray 2008a, p. 3; and Fédération nationale des administrateurs ad hoc 2009, p. 17-18.

⁴⁸⁷ Fédération nationale des administrateurs ad hoc 2009, p. 14.

⁴⁸⁸ Favre-Lanfray 2008a, p. 3; and Grevot 2010, p. 12.

⁴⁸⁹ Fédération nationale des administrateurs ad hoc 2009, p. 15.

⁴⁹⁰ Cour de Cassation 1ére Chambre civile, 23 février 1999 (Bull. n°66). See also, Fédération nationale des administrateurs ad hoc 2009, p. 26-27; and Gouttenoire 2006, p. 62.

Fédération nationale des administrateurs ad hoc 2009, p. 15.

⁴⁹² Gouttenoire 2006, p. 62; and Grevot 2010, p. 12.

⁴⁹³ Gouttenoire 2006, p. 63.
⁴⁹⁴ Attias 2012; Eglin 2006, p. 36; and Ligue des droits de l'Homme 2012.

⁴⁹⁵ Avenard 2015, p. 34; and Ligue des droits de l'Homme 2012.

⁴⁹⁶ Fédération nationale des administrateurs ad hoc 2009, p. 26; and Grevot 2010, p. 13.

⁴⁹⁷ Fédération nationale des administrateurs ad hoc 2009, p. 10-11; and Lebrun 2001, p. 33.

⁴⁹⁸ Fédération nationale des administrateurs ad hoc 2009, p. 11; and Lebrun 2001, p. 33.

5.2.1.4. What requirements are set for the children, e.g. age, level of maturity?

The important requirement for children, both with regards to the **ad hoc administrator** and the **children's lawyer**, is whether the child is *capable de discernement*, so whether the child is capable of forming his or her own views. When this requirement is achieved, the child has the right to be heard (Art. 388-1 FCC and Art. 1182 FCCP) and thus the right to have a children's lawyer.⁴⁹⁹ The requirement is relevant for the ad hoc administrator in a negative sense. If the child is not *capable de discernement* in cases where the child can normally act as a party by themselves and there is a conflict of interests with the parent(s) or guardian, then an ad hoc administrator can be appointed.⁵⁰⁰ When is a child *capable de discernement*? There is no age attached to the requirement, instead the judge must decide per case.⁵⁰¹ According to Attias, children are generally considered capable of forming their own views in France at the age of 7 years old.⁵⁰²

5.2.1.5. What other requirements are applied, e.g. conflict of views between child and parents?

An **ad hoc administrator** is not appointed automatically, the additional requirement of a conflict of interests between the child and the legal representative(s) must be met.⁵⁰³ Article 383 FCC requires the actual existence of a conflict of interests at the time of the judge's decision to appoint an ad hoc administrator, while Article 388-2 FCC only requires the appearance of a conflict of interests.⁵⁰⁴ This means that it could also concern a future conflict of interests, but this conflict must be sufficiently serious, close and threatening.⁵⁰⁵ What is meant by a 'conflict of interests' in both of these articles is very subjective and vague, which on the one hand means that a great variety of cases is included, but also means that the appointment of an ad hoc administrator is very dependent on the judge's discretion.⁵⁰⁶ However, the conflict of interests is generally understood as only concerning conflicts between the child's interests and the parents' interests, not conflicts between the parents, and as having to be sufficiently contradictory or divergent not simply distinct.⁵⁰⁷ According to the *Fédération nationale des administrateurs ad hoc* the judge does not have any discretion when the legal representative of the child requests an ad hoc administrator on the ground of Article 383 FCC and thus the conflict of interests requirement does not apply in this situation.⁵⁰⁸

There are no additional requirements for the children's lawyer.

⁴⁹⁹ Fédération nationale des administrateurs ad hoc 2009, p. 15; Gouttenoire 2006, p. 63; Ligue des droits de l'Homme 2012; and Rongé 2008.

⁵⁰⁰ Fédération nationale des administrateurs ad hoc 2009, p. 14.

⁵⁰¹ Rongé 2008, p. 45.

⁵⁰² Attias 2012.

⁵⁰³ Fédération nationale des administrateurs ad hoc 2009, p. 22; and Gouttenoire 2006, p. 60.

⁵⁰⁴ Favre-Lanfray 2008b, p. 13; and Fédération nationale des administrateurs ad hoc 2009, p. 23.

⁵⁰⁵ Fédération nationale des administrateurs ad hoc 2009, p. 24.

⁵⁰⁶ Fédération nationale des administrateurs ad hoc 2009, p. 23-24.

⁵⁰⁷ Fédération nationale des administrateurs ad hoc 2009, p. 23-24.

⁵⁰⁸ Fédération nationale des administrateurs ad hoc 2009, p. 43.

5.2.1.6. Who decides whether the child will be represented?

For the **ad hoc administrator**, the person who decides whether the child will be represented depends on the legal ground for the appointment. On the ground of Article 383 FCC, the *juge des tutelles* (guardianship judge) has the competence to appoint an ad hoc administrator. The judge must do so automatically at the request of the legal representative (parent(s) or guardian) and has the discretion to do so if the legal representative has not requested it at the request of the public prosecutor or the child or *ex officio*.⁵⁰⁹ Article 388-2 FCC grants the *juge des tutelles* the competence to appoint an ad hoc administrator under the same conditions as included in Article 383 FCC and also grants the judge of the seized instance the competence to appoint an ad hoc administrator *ex officio*.⁵¹⁰ The latter judge can be any of judges involved in the proceedings, e.g. the family judge.⁵¹¹ The legal representatives of the child may appeal against the appointment of an ad hoc administrator (Art. 1210-2 FCCP) within 15 days of the appointment. The function of an ad hoc administrator is not obligatory, so the person who the judge has appointed can decide to decline, the judge will then appoint another ad hoc administrator.⁵¹²

With regards to the **children's lawyer** it is up to the child with discernment to decide whether to have a lawyer. The child can freely choose who to have as a lawyer.⁵¹³ In child protection proceedings, this is also the case according to Article 1186 FCCP, but the child may also request the *juge des enfants* (children's judge) to have the president of the court appoint a children's lawyer for them.

5.2.1.7. How is the child's representative financed?

The **ad hoc administrator** used to work on a gratuitous basis, however since 1999 the law provides for the compensation of an ad hoc administrator, albeit in the form of a fixed lump sum until 2008.⁵¹⁴ Since 2008, Article 1210-3 FCCP in conjunction with Article R. 93(I)(3) French Code of Criminal Procedure (*Code de procédure pénale*; hereafter FCPC) allows for the reimbursement of the ad hoc administrator's travel costs and the compensation for their work if it is an ad hoc administrator appointed from the list in R. 53 FCPC. The amount of compensation is linked to the work done, but remains a – low – fixed sum (Art. 1210-3 FCCP).⁵¹⁵ Although the State Treasury advances the compensation, the Treasury recovers these costs from the party ordered to pay them by the judge (Art. 1210-3 FCCP).⁵¹⁶

⁵⁰⁹ Fédération nationale des administrateurs ad hoc 2009, p. 42 and 44; and Massip 1995.

⁵¹⁰ Favre-Lanfray 2008b, p. 13; and Fédération nationale des administrateurs ad hoc 2009, p. 44.

⁵¹¹ Fédération nationale des administrateurs ad hoc 2009, p. 44.

⁵¹² Massip 1995.

⁵¹³ Chartre nationale de la defense des mineurs 2008.

⁵¹⁴ Fossier 2006, p. 17; and Grevot 2010, p. 16.

⁵¹⁵ Fédération nationale des administrateurs ad hoc 2009, p. 80; Grevot 2010, p. 17; and Guide Méthodologique Administrateur ad hoc 2014, p. 45.

⁵¹⁶ Fédération nationale des administrateurs ad hoc 2009, p. 83.

The costs for the **children's lawyer** should, in principle, be borne by their client. This means that the costs should be borne by the child, and will in some cases be borne by their parents.⁵¹⁷ However, when the children's lawyer functions within the framework of Article 388-1 FCC (the child's right to be heard), then Article 9-1 of the Legal Aid Act (Loi n° 91-647 du 10 juillet 1991 relative à l'aide *juridique*) automatically provides for the right to legal aid from the State. In other cases, the child can also request legal aid from the State according to the standard requirement of insufficient resources (Art. 2 Legal Aid Act).

5.2.2. What is the task of the child's representative in family law proceedings in France?

The task of the **ad hoc administrator** is not defined by the legislator, but their task is to represent the child, protect the child's interests and to assist and accompany the child in the legal proceedings according to legal doctrine.⁵¹⁸ The ad hoc administrator substitutes the child's place, which is normally substituted by the parent, as a party in the proceedings, thereby ensuring the child's access to justice.⁵¹⁹ The task of an ad hoc administrator is limited to the scope of the mission for which they are appointed by the judge, who cannot give the ad hoc administrator a 'blank cheque'.⁵²⁰

The task of the children's lawyer is dependent on the child's position in the legal proceedings. In proceedings to which the child is a party, e.g. child protection proceedings, the children's lawyer represents the child in the court proceedings as a normal lawyer would.⁵²¹ In family law matters where the child is not a party to the proceedings, the children's lawyer does not have a traditional representative function, but instead assists the child in exercising their right to be heard (Art. 388-1 FCC) and in understanding the proceedings and decision.⁵²²

5.2.2.1. What are the function requirements for the child's representative?

The function requirements of an ad hoc administrator are relatively broad, as diversity exists between who functions as an ad hoc administrator or, to put it more positively, the profile of an ad hoc administrator is 'multi-faceted'.⁵²³ According to Article 1210-1 FCCP, the judge will generally appoint a family member of or someone close to the child as the ad hoc administrator unless this is not in the interests of the child.⁵²⁴ In that case, the judge will appoint an ad hoc administrator from the

⁵¹⁷ Rongé 2008, p. 46.

⁵¹⁸ Favre-Lanfray 2008b, p. 11; Fédération nationale des administrateurs ad hoc 2009, p. 11 and 60; and Grevot 2010, p. 15-16. ⁵¹⁹ Fédération nationale des administrateurs ad hoc 2009, p. 11; and Grevot 2010, p. 15.

⁵²⁰ Fossier 2006, p. 17; and Massip 1995.

⁵²¹ Ligue des droits de l'Homme 2012.

⁵²² Avenard 2015, p. 34; Demarchi 2010; and Ligue des droits de l'Homme 2012.

⁵²³ Fédération nationale des administrateurs ad hoc 2009, p. 52 and 59. See also, Fossier 2006, p. 18; and Grevot 2010, p. 13.

⁵²⁴ Fédération nationale des administrateurs ad hoc 2009, p. 53; Fossier 2006, p. 18; and Grevot 2010, p. 13.

list created by the Court of Appeal according to Article R. 53 FCPC.⁵²⁵ This is a very diverse list consisting both of natural as well as legal persons, including private and public legal persons.⁵²⁶ Natural persons often do the work independently in combination with their other jobs.⁵²⁷ To be added to the list, they must: (1) be between the age of 30 to 70 years old, (2) have a proven interest in children and their capabilities, (3) reside in the jurisdiction of the Court of Appeal, (4) never have had a criminal, disciplinary or administrative punishment for acts contrary to good morals, (5) never have been bankrupt.⁵²⁸ As professionalism has made the ad hoc administrator more into a public office, the list contains many private and public legal persons.⁵²⁹ To be added to the list the legal persons must fulfill two requirements: (1) the directors of the legal person must fulfill requirements (4) and (5) applicable to natural persons, and (2) all the persons who will act on behalf of the legal person as an ad hoc administrator must fulfill all five of the requirements applicable to natural persons.⁵³⁰ There are no training requirements for the natural or legal persons on the list.⁵³¹

There are no specific function requirements for the **children's lawyer** as it is not specific institute, in principle any French *avocat* with the general qualifications can act as a children's lawyer.⁵³² However, a national charter on the role and functioning of the children's lawyer was composed and signed at the Conference of Court Presidents (*Conference des bâtonniers*) in 2008 which outlines best practices and creates a group of children's lawyers at the level of the regional bar, who on a voluntary basis bind themselves to these best practice guidelines.⁵³³ The child can freely opt for one of these children's lawyers.⁵³⁴

5.2.2.2. How should the child's representative complete their task?

The **ad hoc administrator** represents and is the child in the proceedings. This means that when the child is not a party to the proceedings, the ad hoc administrator cannot function as a party in the proceedings, but when the child is a party, then the ad hoc administrator has all the same procedural powers as the child would have had.⁵³⁵ If the ad hoc administrator functions as a party to the proceeding on behalf of the child, then they must represent and defend the child's interests.⁵³⁶ To do

⁵³⁴ Chartre nationale de la defense des mineurs 2008.

⁵²⁵ Fédération nationale des administrateurs ad hoc 2009, p. 53.

⁵²⁶ Fédération nationale des administrateurs ad hoc 2009, p. 54; and Grevot 2010, p. 14.

⁵²⁷ Grevot 2010, p. 14.

⁵²⁸ Art. 2 Décret n° 2003-841 du 2 septembre 2003 relatif aux modalités de désignation et d'indemnisation des administrateurs ad hoc institués par l'article 17 de la loi n° 2002-305 du 4 mars 2002.See also, Fédération nationale des administrateurs ad hoc 2009, p. 54.

⁵²⁹ Fossier 2010, p. 18.

⁵³⁰ Art. 3 Décret n° 2003-841 du 2 septembre 2003 relatif aux modalités de désignation et d'indemnisation des administrateurs ad hoc institués par l'article 17 de la loi n° 2002-305 du 4 mars 2002. See also, Fédération nationale des administrateurs ad hoc 2009, p. 55.

⁵³¹ Fédération nationale des administrateurs ad hoc 2009, p. 54-55.

⁵³² Duval 2008, p. 44.

⁵³³ Chartre nationale de la defense des mineurs 2008. See also, Attias 2012.

⁵³⁵ Fédération nationale des administrateurs ad hoc 2009, p. 70; and Fossier 2006, p. 18-19.

⁵³⁶ Fossier 2006, p. 18.

so they will work in collaboration with a lawyer, have contact with all parties involved, including the child, study the dossier and listen to the child and answer all their questions.⁵³⁷ With regards to what exactly the ad hoc administrator must represent, the child's views or the child's interests, it appears to be accepted in legal doctrine that an ad hoc administrator must act according to what they themselves think is right and thus ought to represent what is in the child's best interests.⁵³⁸ While the ad hoc administrator ought to transmit the child's views, the ad hoc administrator not only physically substitutes the non-emancipated child in legal proceedings but also substitutes the child's will because the child does not issue their own choice.⁵³⁹ However, if the ad hoc administrator's view is different from that of the child, then that should be disclosed to the court.⁵⁴⁰

How the **children's lawyer** should complete their task depends in part on the type of procedure, as that defines their task. When the task is the same as that of a regular lawyer, then they should execute their task as they would in normal cases. With regards to the child protection procedure, a few specific rules apply. According to Article 1187 FCCP the children's lawyer receives all the court documents and relevant files, but these may not be shared with the child. The children's lawyer is also obliged to visit the child.⁵⁴¹ When the children's lawyer functions in family matters where the child is not a party but has the right to be heard, then in light of the assisting function, the avocat cannot speak at the hearing.⁵⁴² Instead the children's lawyer informs the judge if and when the child wishes to be heard, without informing the judge what the child has disclosed to them due to the confidential nature of the relationship between the avocat and the child.⁵⁴³

More generally speaking, the children's lawyer must meet with the child, alone if the child is capable de discernement to prevent the risk of a conflict of interests with any third parties (e.g. parents).⁵⁴⁴ When meeting the child, the children's lawyer should explain their role, the confidential nature of their relationship, and the proceedings to the child.⁵⁴⁵ The children's lawyer can only be the child's lawyer, not the lawyer of (one of) the parents, and must inform the parents hereof.⁵⁴⁶ At the conclusion of the procedure, the child who is at least 6 years of age (according to the best practices charter) should be informed of the decision by the children's lawyer.⁵⁴⁷

⁵³⁷ Fédération nationale des administrateurs ad hoc 2009, p. 14-15 and 62; and Grevot 2010, p. 15-16.

⁵³⁸ Fédération nationale des administrateurs ad hoc 2009, p. 60-62; and Fossier 2006, p. 19.

⁵³⁹ Fédération nationale des administrateurs ad hoc 2009, p. 61; and Fossier 2006, p. 21.

⁵⁴⁰ Fédération nationale des administrateurs ad hoc 2009, p. 62.

⁵⁴¹ Ligue des droits de l'Homme 2012.

⁵⁴² Ligue des droits de l'Homme 2012.
⁵⁴³ Chartre nationale de la defense des mineurs 2008.

⁵⁴⁴ Chartre nationale de la defense des mineurs 2008.

⁵⁴⁵ Chartre nationale de la defense des mineurs 2008.

⁵⁴⁶ Avenard 2015, p. 34; Ligue des droits de l'Homme 2012; and Rongé 2008, p. 46.

⁵⁴⁷ Attias 2012; and Chartre nationale de la defense des mineurs 2008.

5.2.2.3. Is the child representative the only option to be heard, are other options available and can they be complementary?

In principle, the ad hoc administrator and children's lawyer function in different types of cases and complement each other in the sense that the one functions when the child is *capable de discernement* while the other functions when the child is not, it is possible that the child has double representation strictu sensu, because the ad hoc administrator will need a lawyer to act legally in proceedings where the child is a party.⁵⁴⁸

Besides the ad hoc administrator and the children's lawyer, the child also has the more general option to be heard (Art. 388-1 FCC) in all procedures which concern the child.⁵⁴⁹ The child is can be assisted by a children's lawyer when making use of this option. The child can either be heard directly by the judge or can be heard indirectly by a third person appointed by the judge.⁵⁵⁰ According to Article 338-9 FCCP, this third person must be an expert from the social, psychological or medical-psychological field. The judge has great discretion in choosing who this third person can be, although it is important that it is someone who can decipher the child's views.⁵⁵¹ Important in light of the right to be heard is that Article 388-1 FCC also includes the child's right to be informed about the possibility to be heard, albeit under the limitation of two conditions: (1) that the child is *capable de discernement*, and (2) that the proceeding must concern the child.

 ⁵⁴⁸ Fossier 2006, p. 20.
 ⁵⁴⁹ Demarchi 2010; and Justen 2014, p. 38.

⁵⁵⁰ Demarchi 2010.

⁵⁵¹ Demarchi 2010.

5.3. The Netherlands

5.3.1. What forms of representation are available for children in family law proceedings in the Netherlands and how are they regulated?

The Netherlands has three forms of representation available for children in family law proceedings. The first two being the bijzondere curator (hereafter: guardian ad litem) in the general form provided by Article 1:250 Dutch Civil Code (Burgerlijk Wetboek; hereafter DCC) and in the filiation form of the *lex specialis* of Article 1:212 DCC.⁵⁵² The third form of representation is separate legal representation. In certain cases, the child either is obliged to, or can choose to, have separate legal representation – a lawyer – to act in the proceedings. Having separate legal representation is the same for children as for adults, in the sense that it is not a separate 'institute' as the guardian ad litem is.

5.3.1.1. When were the forms of representation introduced or amended?

On the 5th of February 1995, the Netherlands ratified the UNCRC by which the Convention took immediate domestic effect.⁵⁵³ Although the ratification was an important step in recognizing children's rights, the UNCRC did not have an immense impact on Dutch legislation, as great reforms were not considered necessary by the government, the Constitution was not amended and no special children's laws or acts were drawn up.⁵⁵⁴

The guardian ad litem has a longstanding history with regards to matters concerning the property of children and the denial of legitimacy. As discussed with regards to South Africa, the existence of curators to represent minor's in litigation originates from Roman Law and continued its existence in Old Dutch Law.⁵⁵⁵ More specifically, in the codified Dutch civil codes of the 19th and 20th century, the vague outline of two current forms of guardian ad litem can already be found.

The precursor to the general guardian ad litem was the *bijzonderen curator* who could be appointed to represent the child in matters concerning the property of children when the interests of the child conflicted with that of the parent entrusted with the administration of the child's property.⁵⁵⁶ In 1995, a major amendment of the law on custody and access in the civil code introduced the general guardian ad litem as we now know it.⁵⁵⁷ The amendment extended the role of the guardian ad litem to include cases of an 'immaterial' nature,⁵⁵⁸ thus also forming a type of 'custody curator'.⁵⁵⁹ This enhanced role of the guardian ad litem was introduced for two reasons. On the one hand, to formalize

⁵⁵² Pieters 2008; Van Teeffelen 2008; and Vlaardingerbroek 2001.

⁵⁵³ Limbeek & Bruning 2015, p. 89-90.

⁵⁵⁴ Liefaard & Vonk 2016, p. 314; and Limbeek & Bruning 2015, p. 89.

 ⁵⁵⁵ Helmholz 1978, p. 229-230 and 232.
 ⁵⁵⁶ Asser 1957, p. 597; and Veegens 1923, p. 256.

⁵⁵⁷ Wet van 6 april 1995 tot nadere regeling van het gezag over en van de omgang met minderjarige kinderen, *Stb.* 1995, 240

⁵⁵⁸ Van Teeffelen 2008.

⁵⁵⁹ Van Wamelen 1995.

the existing practice of courts at that time appointing a guardian ad litem, not as provided for in property matters, but also in matters of an 'immaterial' nature'.⁵⁶⁰ On the other hand, to compensate for not introducing a formal access to proceedings for children.⁵⁶¹ An amendment in 1997 had little impact on Article 1:250 DCC, only adding the phrase 'or both guardians' to comply with the introduction of joint guardianship.⁵⁶² The amendment in 2009 was of more importance.⁵⁶³ The Continued Parenthood and Well-Planned Divorce Act 2009 had the general aim to improve the position of children and the relationship with their parents following divorce and amended Article 1:250 DCC to specifically improve the position of children in the legal proceedings.⁵⁶⁴ The amendment facilitates the appointment of the guardian ad litem, by allowing every judge to make the appointment in cases directly concerning a child, instead of solely the subdistrict court judge (kantonrechter).⁵⁶⁵ The amendment also extended the role of the guardian ad litem, who can now assist the child in the discussions concerning the parenting plan.⁵⁶⁶

The precursor of the **filiation guardian ad litem**, the *bijzonderen voogd*, already in the early 20th century recognized as more of a curator than a *voogd* (guardian), had to be appointed as a representative for the child when proceedings were started to deny the child's legitimacy.⁵⁶⁷ The *bijzonderen voogd* had to defend the interests of the child in these procedures.⁵⁶⁸ The modern day version of the filiation guardian ad litem was introduced in 1998 with the revised law of descent.⁵⁶⁹ The mandatory guardian ad litem was introduced with the aim of protecting the best interests of the child in matters of filiation.⁵⁷⁰ Looking to the future, the role of the filiation guardian ad litem might be further extended in light of the recommendations made by the Staatscommissie Herijking Ouderschap to involve a guardian ad litem on behalf of the unborn child in multiple parent families when they are making a parenting plan.⁵⁷¹

It is difficult to determine the introduction of the separate legal representative, because it is not a separate institution. Legal representatives as such have existed for centuries in legal proceedings for adults. Looking more specifically at the option for children in the Netherland to have a separate legal representative in family law proceedings, two important amendments can be mentioned. Compulsory

⁵⁶⁰ Kamerstukken II 1992/93, 23012, 3, p. 11.

⁵⁶¹ Vlaardingerbroek 2001, p. 103.

⁵⁶² Wet van 30 oktober 1997 tot wijziging van, onder meer, Boek 1 van het Burgerlijk Wetboek in verband met invoering van gezamenlijk gezag voor een ouder en zijn partner en van gezamenlijke voogdij, Stb. 1997, 506. See also, Jansen 2016b. ⁵⁶³ Wet van 27 november 2008 tot wijziging van Boek 1 van het Burgerlijk Wetboek en het Wetboek van Burgerlijke Rechtsvordering in verband met het bevorderen van voortgezet ouderschap na scheiding en het afschaffen van de mogelijkheid tot het omzetten van een huwelijk in een geregistreerd partnerschap (Wet bevordering voortgezet ouderschap en zorgvuldige scheiding), *Stb.* 2008, 500.

Kamerstukken II 2004/05, 301245, 3, p. 13. See also, Van Teeffelen 2007.

⁵⁶⁵ Kamerstukken II 2004/05, 301245, 3, p. 7.

⁵⁶⁶ Kamerstukken I 2008/09, 30145, E, p. 1. See also, Van Teeffelen 2008.

⁵⁶⁷ Asser 1957, p. 472-473; and Veegens 1923, p. 223.

⁵⁶⁸ Asser 1957, p. 473.

⁵⁶⁹ Wet van 24 december 1997 tot herziening van het afstammingsrecht alsmede van de regeling van adoptie, *Stb.* 1997, 772.

⁵⁷⁰ Rapport Staatscommissie Herijking ouderschap 2016, p. 180.

⁵⁷¹ Rapport Staatscommissie Herijking ouderschap 2016, p. 432.

legal representation for children in proceedings concerning the authorization for secure youth care was introduced on the 1st of January 2008 with the Secure Youth Care Act, to offer extra protection to the child considering the drastic nature of the decision.⁵⁷² The compulsory legal representation for children in disputes on the execution of a care and supervision order was introduced with the Implementation Act of the Youth Act 2014.

5.3.1.2. In which types of cases can they be represented?

The general guardian ad litem represents children in matters concerning the child's care and upbringing or concerning the child's property. These are proceedings surrounding the separation of parents, e.g. the (change of) custody, principal residence and access, as well as care proceedings, e.g. family supervision and placement in care orders.⁵⁷³ It is also includes matters of education, labor, and other specific cases concerning children and care measures, e.g. conflicts regarding foster care or an injunction to request treatment.⁵⁷⁴ Theoretically, the guardian ad litem can also be appointed in international child abduction proceedings, however in practice this has never been done as it might impede the strict 6 week timeframe of Hague Convention proceedings.⁵⁷⁵ The general guardian ad litem can represent children in these types of cases under the condition that the conflict must be sufficiently serious.⁵⁷⁶ The Dutch Supreme Court has emphasized that the role of the general guardian ad litem is not meant for general parenting issues, but instead for substantial conflicts.⁵⁷⁷ The guardian ad litem is meant for concrete problems which require legal proceedings if they cannot be resolved amicably.578

The filiation guardian ad litem represents children in matters of parentage. These are proceedings concerning the denial of paternity (Art. 1:200 DCC), denial of maternity (Art. 1:202a DCC), substitute consent to recognize parentage by the court (Art. 1:204(3) and (4) DCC), nullification of recognition of parentage (Art. 1:205 and 1:205a DCC), judicial determination of parentage (Art. 1:207 DCC), and proceedings concerning a claim or dispute of civil status (Art. 1:211 DCC).⁵⁷⁹ Although not expressly determined in the law, adoption proceedings appear to also qualify as (parentage) proceedings in which the filiation guardian ad litem can be appointed, especially because children aged 12 or older must explicitly consent to the adoption (Art. 1:228(1)(a) DCC).⁵⁸⁰

⁵⁷² Kamerstukken II 2005/06, 30644, 3, p. 23. See also, Van Teeffelen 2008.

⁵⁷³ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 5; Kamerstukken II 2004/05, 301245, 3, p. 7; and Ter Haar 2015. ⁵⁷⁴ Ter Haar 2016; and Jansen 2016b. See also, Klaas 2009, for proceedings in which a bijzondere curator may perhaps be

employed.

⁵⁷⁵ Jonker et al. 2015, p. 45. ⁵⁷⁶ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 3. See also, Jansen 2016b; and Ter Haar 2016.

⁵⁷⁷ Hoge Raad 4 February 2005, ECLI:NL:HR:2005:AR4850, para. 3.4.2.

⁵⁷⁸ Hoge Raad 4 February 2005, ECLI:NL:HR:2005:AR4850, para. 3.4.2. See also, Kinderombudsman 2016, p. 11.

⁵⁷⁹ Lok & Vonk 2016; and Schrama 2015.

⁵⁸⁰ See Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 4 and 6.

Children can have **separate legal representation** in a few specific proceedings concerning care and protection orders. In proceedings concerning the authorization for secure youth care the child is obliged to have a lawyer (Art. 6.1.10(4) Youth Act). If children wish to act in disputes on the execution of a care and supervision order, they are also obliged to do so with a separate legal representative according to Article 1:262b DCC jo. Article 1:265k(1) DCC. In multiple other cases children have a formal right to bring proceedings and can do so with a lawyer, although this is not obligatory. These cases are those concerning the termination of a care and supervision order (Art. 1:261 DCC), the revocation of a written instruction on the supervision order (Art. 1:264 DCC), the request to end or shorten the placement in care (Art. 1:265d DCC), and the adjustment of a decision concerning the care orders due to changed circumstances (Art. 1:265g(2) DCC).

5.3.1.3. When can children be represented in family law proceedings in the Netherlands?

The **general guardian ad litem** can be appointed at any phase of the family law proceedings, including during the preliminary injunction or at the stage of appeal.⁵⁸¹ In theory the guardian ad litem can also be involved prior to the court proceeding, however as a judge must appoint the guardian ad litem, the judge must be aware of the potential conflicts between the parent(s) and the child, thus proceedings must have already been instituted.⁵⁸² Once a general guardian ad litem has been appointed they must represent the child at law and otherwise.⁵⁸³ The duration of the role of the guardian ad litem depends on the assignment description given by the judge, can be a short or a longer term.⁵⁸⁴ In any case, the appointment of the guardian ad litem ends when an amicable settlement has been made between the parent(s) or guardian(s) and the child or if the court has made a final judgment.⁵⁸⁵

The **filiation guardian ad litem** will be involved from the very start of the proceedings. Either the court will appoint a guardian ad litem immediately after receiving the application or the guardian ad litem was the initiator of the proceedings by submitting an application on behalf of the child (and need not be re-appointed by the court).⁵⁸⁶ The latter is possible, in part because the filiation guardian ad litem also represents the child at law and otherwise.⁵⁸⁷

Children are free to decide when to hire a **separate legal representative**. Generally, it would make sense for children to do so prior to submitting their application. In the case of secure youth care, the child will be assigned a lawyer at the start of the proceeding.

⁵⁸¹ Kamerstukken II 2004/05, 301245, 3, p. 13; Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 5.

⁵⁸² Kinderombudsman 2016, p. 11.

⁵⁸³ Kamerstukken I 2008/09, 30145, E, p. 1.

⁵⁸⁴ De Graaf & Limbeek 2011.

⁵⁸⁵ Jansen 2016b.

⁵⁸⁶ Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 3.

⁵⁸⁷ Pieters 2008.

5.3.1.4. What requirements are set for the children, e.g. age, level of maturity?

The text of both articles concerning the **guardian ad litem** do not mention any requirements with regards to the child's age or maturity,⁵⁸⁸ however it is a factor taken into account. The courts have great discretion in deciding whether to appoint a **general guardian ad litem** and the age or maturity of the child is only one of the factors the court will consider.⁵⁸⁹

It is a whole other story for the **filiation guardian ad litem**, who must be appointed once a filiation proceeding is pending, the courts have no discretion.⁵⁹⁰ Thus, when the filiation guardian ad litem is appointed following the commencement of proceedings by another party there is no age or maturity requirement for the child. This is different when the guardian ad litem initiates the proceedings on behalf of the child. In those cases, children should be of a certainty maturity, they should be sufficiently able to foresee the consequences of the application.⁵⁹¹ Generally, children from the age of 12 onwards will be considered sufficiently mature as this is also the age limit applied for the child's right to veto a change of their legal status (see e.g. Art. 1:204(1)(d) DCC).⁵⁹² Only in exceptional situations will children under the age of 12 years old be able to initiate a proceeding through their guardian ad litem. Considering the existing examples in the case law, it will exceptionally even be possible for very young children to be represented through a guardian ad litem initiating filiation proceedings if this is in their best interests.⁵⁹³

With regards to the **separate legal representative**, there are two situations. In secure youth care proceedings, there is no requirement placed with regards to the age or maturity of the child. In all the cases concerning care and supervision orders and the placement in care, the child is required to have at least reached the age of 12.⁵⁹⁴

5.3.1.5. What other requirements are applied, e.g. conflict of views between child and parents?

As mentioned above, the appointment of a **filiation guardian ad litem** is mandatory in filiation proceedings and therefore no other requirements are applied. Other requirements are applied by the courts in determining whether to appoint a **general guardian ad litem** as the courts have discretion. Article 1:250 DCC grants courts the competence to appoint a guardian ad litem when 'the interests of the parent(s) with parental authority or of the guardian(s) are in conflict with those of the child [...] if the court considers this necessary in the child's best interests, taking into account in particular the

⁵⁸⁸ See also, Schrama 2015.

⁵⁸⁹ Jansen 2016b.

⁵⁹⁰ De Graaf & Limbeek 2011; and Lok & Vonk 2016.

⁵⁹¹ Schrama 2015.

⁵⁹² Schrama 2015.

⁵⁹³ Schrama 2015; and Van Teeffelen 2008.

⁵⁹⁴ See Art. 1:261, 1:262b, 1:264, 1:265d and 1:265g DCC.

nature of this conflict of interests'.⁵⁹⁵ Two main criteria are applied, the conflict of interests and the necessity in the interests of the child. The first should be understood broadly according to the guidelines of the Landelijk Overleg Vakinhoud Familie- en Jeugdrecht (the National Consultations on Family and Child Law matters, hereafter: LOVF).⁵⁹⁶ It need not be a direct conflict between the child and (one of) the parents, but there can already be a conflict of interests if there is a conflict between the two parents or guardians.⁵⁹⁷ This is so, because if the parents' views are diametrically opposite each other in the conflict, they are both not capable of representing the child's views as they normally should.⁵⁹⁸ The best interests of the child, as included in the second criterion of necessity, should form a primary consideration of the court.⁵⁹⁹

Although some guidance is provided, the criteria remain rather vague which, in combination with the discretion afforded to courts, leads to uncertainty in practice.⁶⁰⁰

There are no additional requirements applied with regards to separate legal representation. In most situations, it is for the child to decide whether to take a lawyer, with regards to secure youth care the judge will always ex officio assign a lawyer to the child (Art. 6.1.10(4) Youth Care Act).

5.3.1.6. Who decides whether the child will be represented?

The general guardian ad litem is appointed by the judge of the court in which the proceeding is pending, according to Article 1:250 DCC this can be the subdistrict court judge, the judge at the court first instance, at the court of appeal or at the Supreme Court.⁶⁰¹ The judge can decide to appoint a guardian ad litem at the request of an interested party (belanghebbende) or ex officio.⁶⁰² Who is an interested party? The term 'belanghebbende' refers to Article 798 Dutch Code of Civil Procedure (Wetboek van Burgerlijke Rechtsvorming; hereafter DCCP), interested parties are those whose rights and duties are directly affected by the case. This includes the child, the parent(s) or guardian(s), persons who have 'family life' with the child, but also the guardianship agency or the association which operates a shelter.⁶⁰³ In deciding whether to appoint a general guardian ad litem, the judge can opt to hear the child about the request or the intention to appoint ex officio.⁶⁰⁴ If the judge decides to appoint, then it appears that a letter of intent is required from the appointed guardian ad litem.⁶⁰⁵

⁵⁹⁵ Original text: 'de belangen van de met het gezag belaste ouders of een van hen dan wel van de voogd of de beide voogden in strijd zijn met die van de minderjarige [...] indien hij dit in het belang van de minderjarige noodzakelijk acht, daarbij in het bijzonder de aard van deze belangenstrijd in aanmerking genomen'.

⁵⁹⁶ Werkproces benoeming bijzondere curator o.g.v. Art. 1:250 BW 2014. See also, Jansen 2016a, p. 2179; and Ter Haar 2015. ⁵⁹⁷ Werkproces benoeming bijzondere curator o.g.v. Art. 1:250 BW 2014, p. 3.

⁵⁹⁸ Ter Haar 2015.

⁵⁹⁹ Ter Haar 2016.

⁶⁰⁰ Limbeek & Bruning 2015, p. 92; and Strutz & Verhagen 2015, p. 308.

⁶⁰¹ Kamerstukken I 2008/09, 30145, E, p. 1.

⁶⁰² Kamerstukken I 2008/09, 30145, E, p. 1. See also Kinderombudsman 2016, p. 11; and Jansen 2016a, p. 2179.

⁶⁰³ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 3; and Ter Haar 2015.

⁶⁰⁴ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 4.

⁶⁰⁵ Pieters 2008.

The **filiation guardian ad litem** is appointed by the court before which the procedure is pending according to the article.⁶⁰⁶ If there is no procedure pending, the appointment of a guardian ad litem to initiate proceedings on behalf of the child can be requested by the legal representative of the child (e.g. parent(s) or guardian(s) following Art. 1:245(4) DCC), the legal parent without parental authority as well as the child's natural father who has 'family life' with the child (as an interested party following Art. 798(1) DCCP).⁶⁰⁷ If the guardian ad litem is to initiate proceedings at the child's request, it will probably occur directly without prior appointment by the court.

In proceedings concerning the authorization of secure youth care, the court orders ex officio the Legal Aid Board to assign a separate legal representative to the child (Art. 6.1.10(4) Youth Care Act). In all other proceedings where a child can act with or without a separate legal representative it is for children themselves to hire a lawyer. The courts cannot appoint a separate legal representative for the child.

5.3.1.7. How is the child's representative financed?

There is no special legal aid regime applicable to both types of guardian ad litem in the Netherlands or to the separate legal representative. The general Legal Aid Act (Wet op de rechtsbijstand) is applicable.⁶⁰⁸ The costs of the guardian ad litem or the representative can be reimbursed through statesubsidized legal aid if the guardian ad litem or representative is registered at the Legal Aid Board (Raad voor Rechtsbijstand).⁶⁰⁹ The guardian ad litem will be reimbursed with a fixed amount calculated at the hand of the specific tasks and the type of case for which the guardian ad litem has been appointed.⁶¹⁰ While legal aid is normally awarded following a means test, according to Article 8[©] of the Legal Aid (Personal Contributions) Decree (*Besluit eigenbijdrage rechtsbijstand*) the means of the child and the parents are disregarded when the child requests legal aid in a conflict with their parent(s).⁶¹¹ Furthermore, Article 6(1)(d) of the Decree determines that the child is not obliged to pay a personal contribution when a general guardian ad litem has been appointed on their behalf.⁶¹² The filiation guardian ad litem is not named in that article. Case law of the Dutch Supreme Court has shown that the filiation guardian ad litem can be ordered to pay the costs of the proceedings, which must be borne by the child, if the guardian ad litem is found to be in the wrong and has litigated

⁶⁰⁶ See also, Vlaardingerbroek 2001.

⁶⁰⁷ Lok & Vonk 2016. ⁶⁰⁸ *Kamerstukken I* 2008/09, 30145, E, p. 2; Pieters 2012; and Ter Haar 2016. ⁶⁰⁹ Kamerstukken I 2008/09, 30145, E, p. 2; Pieters 2012; Strutz & Verhagen 2015, p. 310; and Ter Haar 2016.

⁶¹⁰ *Kamerstukken I* 2008/09, 30145, E, p. 2; Strutz & Verhagen 2015, p. 311.

⁶¹¹ Kamerstukken I 2008/09, 30145, E, p. 2; Jansen 2016b.

⁶¹² Kamerstukken I 2008/09, 30145, E, p. 2; Pieters 2012; and Ter Haar 2016.

unnecessarily.⁶¹³ The separate legal representative in cases of secure youth care is appointed through the Legal Aid Board and thus financed through the State (Art. 6.1.10(4) Youth Care Act).

5.3.2. What is the task of the child's representative in family law proceedings in the Netherlands?

The task of the general guardian ad litem is to represent the best interests of the child in a concrete conflict.⁶¹⁴ The primary role, according to the guidelines of the LOVF, is to present the child's views to the parents in order to mediate an amicable settlement.⁶¹⁵ However, if an amicable settlement is not possible, then the guardian ad litem must represent the child's views and especially the child's objective interests in court.⁶¹⁶ In a written report to the court, the guardian ad litem must include a description of the actions taken, their findings, and an advice in light of the child's interests on the conflict.⁶¹⁷ The guardian ad litem also is expected to – but not legally obliged to – inform and explain the course and potential outcome of the proceedings, as well as the final decision. in an ageappropriate manner to the child.⁶¹⁸

The task of the filiation guardian ad litem is to exclusively represent the best interests of the child in the filiation proceedings.⁶¹⁹ The guardian ad litem is the representative, but also legal counsel of the child.⁶²⁰ However, the guardian ad litem is not simply a 'mouthpiece' for the child,⁶²¹ their task is to independently and objectively determine what is in the short and long term interests of the child, through speaking to the involved parties and aiming to determine the factual and legal truth.⁶²² It does not matter whether the person who requested the representation of the child disagrees with the guardian ad litem's point of view, as the guardian ad litem ought only to represent the child's best interests.623

The task of the **separate legal representative** of the child is the same as that of a legal representative on behalf of adults: to represent the child in the legal proceedings. This means that the legal representative ought to act on the directions of the child, as the child is their client.

⁶¹³ Hoge Raad 6 October 2006, ECLI:NL:HR:2006:AV9444. See also, Schrama 2015.

⁶¹⁴ Pieters 2008.

⁶¹⁵ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 11; Kentie & Hendriks 2013; Kinderombudsman 2016, p. 10; and Strutz & Verhagen 2015, p. 311.

⁶¹⁶ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 11; De Gaaf & Limbeek 2011; and Strutz & Verhagen 2015, p. 311.

⁶¹⁷ De Gaaf & Limbeek 2011; and Strutz & Verhagen 2015, p. 313.

⁶¹⁸ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 12; and Strutz & Verhagen 2015, p. 311 and ^{313.}
 ⁶¹⁹ Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 6; Schrama 2015; and Vlaardingerbroek 2001, p.

^{104.} ⁶²⁰ Vlaardingerbroek 2001, p. 107.

⁶²¹ Lok & Vonk 2016.

⁶²² Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 6; Schrama 2015; and Vlaardingerbroek 2001, p.

⁶²³ Lok & Vonk 2016.

5.3.2.1. What are the function requirements for the child's representative?

In theory, the **general guardian ad litem** can be anyone, Dutch law does not impose any requirements, it is up to the judge to decide.⁶²⁴ However, the guidelines of the LOVF and the legal aid rules, do include some requirements. The list of general guardians ad litem at the Legal Aid Board are lawyers and mediators (some with other professional backgrounds, e.g. psychologists and orthopedagogues) who have registered with the Board and can therefore represent children at State expense.⁶²⁵ These neutral professional guardians ad litem are most often appointed, as they are considered to have mediation skills, child communication skills, and experience with parental conflicts.⁶²⁶ A parent or guardian of the child should not be appointed as guardian ad litem,⁶²⁷ and in principle, neither should someone who has previously been in contact with the child, unless this is specifically opted for as they are well-informed of the existing conflict.⁶²⁸ Generally, someone from the Child Care and Protection Board is also not suited, as it will impede the Child Care and Protection Board is also not suited as it will impede the Child Care and Protection Board at the guardian ad litem, the judge should consider the nature of the conflict, the professional background of the guardian ad litem and the place of residence of the child and other involved parties.⁶³⁰ In exceptional cases, it is possible for the judge to appoint two guardians ad litem.⁶³¹

The courts have composed a limited list of potential **filiation guardians ad litem**, which is now provided for by the Legal Aid Board.⁶³² The guardians ad litem on this list are lawyers who have 'demonstrable and recent experience' in filiation proceedings, have followed courses or training programs in that area, and preferably have mediation skills according to the LOVF guidelines.⁶³³ If the guardians ad litem on the list prove to be unsatisfactory, they can be removed from the list.⁶³⁴

The **separate legal representative** of the child can only be a lawyer who is a member of the Netherlands Bar Association (*NovA*) and a member of the District Bar Association. The representative must comply with all the standard requirements for lawyers as provided for by the NovA. With regards to the representative in secure youth care proceedings, the lawyer must comply with the

⁶²⁴ Kamerstukken I 2008/09, 30145, E, p. 1; and Strutz & Verhagen 2015, p. 310.

⁶²⁵ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 6; and Van Leuven et al. 2015.

⁶²⁶ Kentie & Hendriks 2013.

⁶²⁷ Pieters 2008.

⁶²⁸ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 6.

⁶²⁹ Jansen 2016b.

⁶³⁰ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 6.

⁶³¹ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 6.

⁶³² Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 3.

⁶³³ Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 2.

⁶³⁴ Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 3.

requirements of the specialization for youth cases, the representative in the other cases must comply with the specialization for family law provided for by the Legal Aid Board.⁶³⁵

5.3.2.2. How should the child's representative complete their task?

There are no special guidelines as to how a **separate legal representative** should fulfill their task when representing a child in family law disputes. The standard guidelines for lawyers as provided for by NovA should be followed. For both types of **guardian ad litem** the LOVF guidelines explain how their tasks should be completed.

A general guardian ad litem is appointed through an interlocutory order in which the court formulates a concrete task description.⁶³⁶ For example, the court may require the guardian ad litem to speak to certain third parties (although a guardian ad litem can also do this at their own initiative).⁶³⁷ The guardian ad litem then gets four weeks time to provide the court with a written report which includes the guardian ad litem's position on the conflict.⁶³⁸ During these four weeks the guardian ad litem should personally speak to the child, if the child has reached the age of 12 years old or is younger, but can be considered sufficiently capable of a reasonable appreciation of their interests to be spoken to.⁶³⁹ If a psychologist or other behavioural expert has been appointed as guardian ad litem, then the guidelines also advise contact with much younger children.⁶⁴⁰ The guardian ad litem should not only speak to children to determine what will be in their best interests and to present the child's views to the court, but also to explain the proceedings and the outcome to the child.⁶⁴¹

The **filiation guardian ad litem** also gets a task description and instructions from the court once appointed. If appointed to represent the child when another party initiated the filiation proceedings, the guardian ad litem must submit what is in the best interests of the child in a written report after 4 weeks (unless an extension is requested).⁶⁴² If appointed by the court or approached as the potential applicant on behalf of the child, then the guardian ad litem must determine whether initiating proceedings is in the best interests of the child.⁶⁴³ It remains unclear what happens when the guardian ad litem ad litem decides not to initiate proceedings. Schrama is of the opinion that a guardian ad litem appointed with the task to intiate proceedings on behalf of the child must do so, albeit with a supplementary report stating that in their view it is contrary to the interests of the child to grant the

⁶³⁵ Inschrijvingsvoorwaarden advocatuur 2016.

⁶³⁶ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 7; Kentie & Hendriks 2013; and Ter Haar 2015.

⁶³⁷ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 7.

⁶³⁸ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 8.

⁶³⁹ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 10.

⁶⁴⁰ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 10.

⁶⁴¹ Werkproces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 14.

⁶⁴² Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 5; and Pieters 2008.

⁶⁴³ Broekhuijsen-Molenaar 2016.

application, in order to allow the court to consider the case.⁶⁴⁴ Both Pieters and Lok and Vonk are of the opinion that the guardian ad litem can opt not to submit an application on behalf of the child, although Lok and Vonk doubt whether the guardian ad litem must then still submit a written report to the court.⁶⁴⁵

In any case, the filiation guardian ad litem may do various things to determine the best interests of the child. The LOVF guideline clearly expects the guardian ad litem to personally speak to the child who has reached the age of 12 and those who are younger but considered sufficiently mature, even if the court itself also hears the child.⁶⁴⁶ The guardian ad litem should also speak separately to the other involved parties, e.g. the child's mother, legal father, the acknowledger, the man who wants to acknowledge the child, or any of their heirs.⁶⁴⁷ In addition, the parties should be informed that a DNA test may be required and the child should be explained what the proceedings are about.⁶⁴⁸ Determining what is in the best interests of the child in filiation proceedings cannot be done through a checklist, but various factors have been drafted by Schrama: whether the parents are in agreement, the child's interests in having two legal parents, the child's interests in having legal, social and biological parentage match-up, and potentially the child's own views.⁶⁴⁹

5.3.2.3. <u>Is the child representative the only option to be heard, are other options available and</u> can they be complementary?

There are other options for the child's views to be taken into account in Dutch family law proceedings. The first being the judicial meetings in which the child is heard in person by the judge(s). According to Article 809 DCCP, the judge may only decide in family-law matters concerning children who are 12 years or older (or 16 years or older in child maintenance cases) after having given the child the opportunity to share their views in person or in writing.⁶⁵⁰ The judge may also give children under the age of 12 (or 16) the opportunity to share their views on the matter. The second option, reports from the Child Care and Protection Board, is less direct than the judicial meetings and does not have hearing the child as its core aim. The judge in family-law matters (except in child maintenance cases) can request such an advice report (Art. 810 DCCP), and must do so in certain care proceedings according to Article 810(4) DCCP. The Child Care and Protection Board will often speak to the child, but need not inform the court of the child's views.

Can the forms of representation and the other options function complementarily? The **general guardian ad litem** and the **filiation guardian ad litem**, can generally not function complementarily

⁶⁴⁴ Schrama 2015.

⁶⁴⁵ Lok & Vonk 2016; and Pieters 2008.

⁶⁴⁶ Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 4; and Lok & Vonk 2016.

⁶⁴⁷ Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 4.

⁶⁴⁸ Richtlijn benoeming bijzondere curator o.g.v. Art. 1:212 BW 2014, p. 4; and Lok & Vonk 2016.

⁶⁴⁹ Schrama 2015.

⁶⁵⁰ See also, Van Triest 2004.

as the types of cases in which they can be appointed differ. However, as mentioned previously, it is possible, in exceptional cases, to have two general guardians ad litem.⁶⁵¹ A separate legal representative cannot function at the same time as a filiation guardian ad litem, as they function in different types of cases. It is possible to have a separate legal representative at the same time as a general guardian ad litem, although it is exceptional.⁶⁵²

The guardians ad litem and separate legal representatives can also be complementary to the other options for the child to be heard. Even though a guardian ad litem has been appointed, or the child is represented by a lawyer, it is expected that the judge will hear the child in person.⁶⁵³ It is also possible for the judge to request an advice from the Child Care and Protection Board in addition to the appointment of a guardian ad litem or when the child has a separate legal representative.⁶⁵⁴

 $^{^{651}}$ Werk proces benoeming bijzondere curator o.g.v. art. 1:250 BW 2014, p. 6. 652 Klaas 2009.

⁶⁵³ Lok & Vonk 2016.

⁶⁵⁴ See e.g., Vlaardingerbroek 2001.

5.4. South Africa

5.4.1. <u>What forms of representation are available for children in family law</u> proceedings in South Africa and how are they regulated?

Two forms of representation are available in South Africa for children in family law proceedings: the **legal representative** and the **curator ad litem**. The legal representative is provided for in the Bill of Rights, Section 28(1)(h) of the Constitution of the Republic of South Africa, 1996. This section grants children 'the right to have a legal practitioner assigned to him or her, by the state at state expense, in civil proceedings affecting the child, if substantial injustice would otherwise result'. The curator ad litem has a common law origin⁶⁵⁵ and is a person who 'conduct[s] litigation in the name and in the interests of the minor'.⁶⁵⁶

5.4.1.1. When were the forms of representation introduced or amended?

The right to separate legal representation for children in civil proceedings was introduced in 1996 with the Constitution of the Republic of South Africa. The interim Constitution of 1993 and its section on the fundamental rights of the child (s. 30), was much barer and did not include any right to be heard or to be represented.⁶⁵⁷ This changed following the ratification of the UNCRC. The UNCRC was the first international human rights treaty to be ratified by South Africa's first universally elected democratic government on the 16th of June 1995.⁶⁵⁸ Through the swift ratification of the UNCRC, children's rights were placed on a pedestal and, according to Sloth-Nielsen, given a 'prominent role in the reconstruction of South African society'.⁶⁵⁹ The main rights afforded by the UNCRC were imbedded into the Constitution of 1996 by the extensive section 28.⁶⁶⁰ According to the South African Constitutional Court, 'section 28 must be seen as responding in an expansive way to our international obligations as a State party to the United Nations Convention on the Rights of the Child'.⁶⁶¹ Amongst others, the child's right to representation of Article 12(2) of the UNCRC has been incorporated into s. 28(1)(h) of the South African Constitution.⁶⁶²

While an incidental form of representation was already provided for in the Divorce Act of 1979,⁶⁶³ the new Constitutional protection of children's rights drove the government to take further measures to revise legislation, policy and practice regarding children and lead to 'substantial and measurable gains'.⁶⁶⁴ One of the revisions was the new Section 8A inserted into the Child Care Act of

⁶⁵⁵ Boezaart 2013b, p. 708.

⁶⁵⁶ Legal Aid Board in re Four Children (512/10) [2011] ZASCA 39 (29 March 2011), para. 12.

⁶⁵⁷ Kassan 2003, p. 167; Sloth-Nielsen 1996, p. 323.

⁶⁵⁸ Sloth-Nielsen 1996, p. 323.

⁶⁵⁹ Boezaart 2013a, p. 357; Sloth-Nielsen 1996, p. 324.

⁶⁶⁰ UNCRC Initial Periodic Report of South Africa 1998, para. 1; and Sloth-Nielsen 2002, p. 139.

⁶⁶¹ S v M (Centre for Child Law as Amicus Curiae) (CCT 53/06) [2007] ZACC 18 (26 September 2007), para. 16. Referred to in: Liefaard & Doek 2015.

⁶⁶² Zaal & Skelton 1998, p. 540.

 $^{^{663}}$ See S. 6(4) which provides for the appointment of a legal practitioner to represent a child in divorce proceedings at the cost of the parties.

⁶⁶⁴ Boezaart 2013a, p. 357; Kassan 2003, p. 167; and Sloth-Nielsen 1996, p. 324.

1983 through the Child Care Amendment Act 96 of 1996, to bring the Child Care Act in line with s. 28(1)(h) of the Constitution.⁶⁶⁵ Section 8A provided for the appointment of a legal representative at state expense to represent a child in a hearing of the children's court where this is in the child's best interest,⁶⁶⁶ but was repealed by the Children's Act 38 of 2005.⁶⁶⁷ The adoption of the latter Act was even more important for the child's right to participation and representation in South Africa.⁶⁶⁸ The aim of the Children's Act includes giving effect to the constitutional rights of the child (s. 2(b)). According to Boezaart and de Bruin, the Act 'pioneered a new era in child participation in legal proceedings', ⁶⁶⁹ as it offers consistency that was previously lacking with regard to the child's right to be heard.⁶⁷⁰

While some authors are of the opinion that s. 28(1)(h) of the Constitution also includes the right of the child to a curator ad litem,⁶⁷¹ the curator ad litem has a much more extensive history originating from Roman law. In Roman law, the distinction was made between the *tutor* and the *curator*, the latter having the duty to assist the minor in litigation, next to protecting the minor's property.⁶⁷² The child's right to participation in litigation as provided for in Roman law remained similar in the Roman-Dutch law of the 17th and 18th Centuries: children were represented by their fathers or their guardians.⁶⁷³ This included representation by a curator ad litem in certain cases.⁶⁷⁴ Through the transplant of Roman-Dutch law to South Africa, the curator ad litem was introduced and remained in the common law of South Africa.⁶⁷⁵

5.4.1.2. In which types of cases can they be represented?

Legal representative

Children can be represented by a legal representative in a wide variety of cases as laid down in the Children's Act of 2005 (hereafter SACA 2005) and the Divorce Act of 1979.

In the Children's Act of 2005 there are several specific types of cases for which a legal representative is provided, further elaborating the general principle of participation in s. 10 SACA 2005.⁶⁷⁶ In court proceedings concerning parental responsibilities and rights agreements (s. 22(4)(b)),⁶⁷⁷ the assignment of contact and care to an interested person (s. 23), the assignment of

⁶⁶⁵ Zaal & Skelton 1998, p. 541. ⁶⁶⁶ Zaal & Skelton 1998, p. 552.

⁶⁶⁷ S. 313 of the Children's Act 38 of 2005.

⁶⁶⁸ Heaton 2012, p. 403.

⁶⁶⁹ Boezaart & de Bruin 2011, p. 417.

⁶⁷⁰ Cleophas & Assim 2015, p. 294.

⁶⁷¹ Boezaart 2013b, p. 712.

⁶⁷² Helmholz 1978, p. 229-230 and 232.

⁶⁷³ De Bruin 2010, p. 65.
⁶⁷⁴ Bisschop 1904, p. 43; and De Bruin 2010, p. 507.
⁶⁷⁵ De Bruin 2010, p. 50 and 371. Roman-Dutch law is the common law of South Africa, see also Heaton 2012, p. 398.

⁶⁷⁶ Skelton et al. 2010, p. 264.

⁶⁷⁷ Before registering a parental responsibilities and rights agreement the court must be satisfied that the agreement is in the best interests of the child, therefore the court can appoint a legal practitioner (see s. 22(5) SACA 2005).

guardianship (s. 24), the confirmation of paternity (s. 26(b)), or the termination, extension, suspension or restriction of parental responsibilities and rights (s. 28), section 29(6)(a) SACA 2005 provides that the court may appoint a legal practitioner to represent the child. With regards to international child abduction cases there is also a specific right for the child to have legal representation. According to s. 279 SACA 2005 the child must have a legal representative in all applications regarding the Hague Convention on International Child Abduction. This is to ensure that the child is afforded a serious opportunity to object to being returned, s. 278(3).⁶⁷⁸

More generally, s. 55 SACA 2005 provides for the legal representation of children who are not yet being represented but are involved in any matter before the children's court. Looking at s. 45, this concerns a wide range of matters besides those already discussed above, including: the protection and well-being of a child, the support of a child, the provision of development or intervention services, civil proceedings concerning maltreatment, neglect and abuse, the temporary safe care or alternative care of a child, and the adoption of a child. This means that in child protection cases, as further discussed in sections 150 to 160 SACA 2005, the child also has a right to legal representation. Also in adoption cases the child has the right to legal representation. Children over the age of 10 and children under the age of 10 who are of an age, maturity and stage of development to understand the implications have to consent to their adoption (s. 233(1)(c) SACA 2005). As the child cannot be exempted from giving this consent, it is important the children are sufficiently counseled and assisted.⁶⁷⁹

Section 6(4) of the Divorce Act of 1979 provides specifically for the appointment of a legal representative for a child in divorce proceedings for the purposes of safeguarding the child's interests in orders regarding maintenance, custody, guardianship, or access to the child.⁶⁸⁰

Curator ad litem

There is no clear list of types of proceedings in which children can or cannot be represented by a curator ad litem. In part, because the curator ad litem is grounded in common law. Rule 57 of the Uniform Rules of Court, which applies to the High Courts, does not delineate in which types of cases a curator ad litem can or cannot be appointed. The same applies to s. 33 of the Magistrates' Court Act of 1944, which simply provides for the curator ad litem at the level of proceedings in magistrates' courts in 'any case in which such a curator is required or allowed by law'. It has been alleged by some that s. 6 of the Divorce Act of 1979 not only provides for the representation by a legal representative but also by a curator ad litem. However, Boezaart argues that this is clearly not correct, s. 6(4) explicitly refers only to a legal practitioner for divorce proceedings.⁶⁸¹ Potentially the same can be said about s. 29(6)(a) SACA 2005, which refers to a 'legal practitioner', while s. 55 SACA 2005

⁶⁷⁸ Boezaart 2013a, p. 365.

⁶⁷⁹ Ferreira 2013, p. 379.

⁶⁸⁰ See s. 6(3) Divorce Act.

⁶⁸¹ Boezaart 2013b, p. 711.

refers more broadly to a 'legal representative'. According to the Supreme Court of Appeal the court has a wide discretion to appoint a curator ad litem.⁶⁸² In previous case law, curators ad litem have been appointed in a wide variety of types of proceedings, amongst others, in (inter-country) adoption proceedings⁶⁸³ and parental responsibilities and rights disputes.⁶⁸⁴

5.4.1.3. When can children be represented in family law proceedings in South Africa?

It is not clear from what moment in, or leading up to, the proceedings that children can be represented. Based on a textual interpretation of the various sections in the Constitution, Children's Act, and Divorce Act it appears that the legal representative is mostly appointed from the moment that proceedings have been instituted, e.g. 'in' or 'at' proceedings. However, it is likely that both the legal representative and the curator ad litem can be appointed at an earlier stage as their role can be to assist the child in getting access to the court, s. 14 SACA 2005.

5.4.1.4. What requirements are set for the children, e.g. age, level of maturity?

The above sections show that there are various sections and rules applicable to the forms of representation for children in South Africa. This section will briefly consider whether there are any requirements set as to when children can make use of the forms of representation. Firstly, s. 10 SACA 2005 which contains the general principle of the child's right to participate in any matter concerning them, does limit the child's right to participate to 'every child that is of such an age, maturity and stage of development as to be able to participate'. Thus, this section provides for an open norm as to when children should participate. It does not provide for a norm regarding when children have access to representation. Section 14 SACA 2005 which provides for the child's right to bring, and to be assisted in bringing, a matter to the court does not contain any requirements regarding the child's age or maturity. Although Heaton has argued that the child's access to court, e.g. with assistance of a curator ad litem, in this section is limited to children over the age of $7^{,685}$ there is no distinction made and thus one can conclude no age limit is imposed.686

Legal representative

More specifically, s. 29(6)(a), s. 55 and s. 279 SACA 2005 and s. 6(4) of the Divorce Act all do not impose an age limit or a requirement of maturity for the child to be represented by a legal representative. Although the consent for adoption is required based on either the age limit or the level

⁶⁸² Legal Aid Board in re Four Children (512/10) [2011] ZASCA 39 (29 March 2011), para. 12.

⁶⁸³ Du Toit v Minister of Welfare and Population Development (Lesbian and Gay Equality Project as Amicus Curiae) (CCT 40/01) [2002] ZACC 20 (10 September 2002); AD v DW (Centre for Child Law as Amicus Curiae; Department for Social Development as Intervening Party) [2007] ZACC 27 (7 December 2007); Ex parte Centre for Child Law in re: six minor *children* [2012] Unreported case no. 11762/2012. See also: Boezaart 2013b, p. 717-719. ⁶⁸⁴ S v J (695/10) [2010] ZASCA 139 (19 November 2010).

⁶⁸⁵ The age limit of 7 years and older comes from the distinction in Roman-Dutch law between an *infans* (\leq 7 years old) and *minor* (> 7 years old) with regards to the legal capacity of the child, see also Boezaart 2013 2013b, p. 708. 686 Boezaart & de Bruin 2011, p. 420.

of maturity and stage of development, there are no requirements imposed with regards to the legal representative.

Curator ad litem

With regards to the curator ad litem, both rule 57 of the Uniform Rules of Court and s. 33 of the Magistrates' Court Act do not impose any requirements with regards to the child's age, maturity or stage of development.

5.4.1.5. What other requirements are applied, e.g. conflict of views between child and parents?

The constitutional right to legal representation for children provided in s. 28(1)(h) seems to be dependent on the requirement 'if substantial injustice would otherwise result'.⁶⁸⁷ This would mean that the legal representation for children in South Africa would always be dependent on the ambiguous 'substantial injustice' test.⁶⁸⁸ However, a different interpretation is also possible. Namely, that the 'substantial injustice' test refers solely to the child's right to have legal representation assigned by the state at state expense.⁶⁸⁹ This also appears to be how the Legal Aid Board of South Africa understands the 'substantial injustice' test.⁶⁹⁰ In that regard the Legal Aid Guide of 2014 distinguishes six criteria to decide if the child passes the test: 1. seriousness of the issue for the child, 2. complexity of the law and procedure, 3. ability of the child to represent him or herself effectively without a lawyer, 4. financial situation of child, their parents or guardians, 5. child's chances of success in the proceeding (but this criteria only applies if the child brings the proceedings), and 6. whether the child has a substantial disadvantage compared to the other parties.⁶⁹¹ According to Cleophas and Assim, in various cases the South African courts also consider the 'substantial injustice' test to determine whether to appoint a legal representative for the child.⁶⁹²

Legal representative

With regards to the appointment of a legal practitioner for children in family law proceedings, two situations must be considered. Section 6(4) of the Divorce Act poses no further requirements for the appointment of a legal practitioner. It is however located in the section 'safeguarding of interests of dependent and minor children' and the appointment is possible 'for the purposes of this section', so one could argue that the appointment is dependent on it being in the interests of the child. For all other appointments of legal practitioners in family law proceedings through the Children's Act, s. 55 makes

 ⁶⁸⁷ Zaal & Skelton 1998, p. 541.
 ⁶⁸⁸ Kassan 2003, p. 168.

⁶⁸⁹ Cleophas & Assim 2015, p. 298.

⁶⁹⁰ Legal Aid Guide 2014, para. 4.18.1, p. 65.

⁶⁹¹ Legal Aid Guide 2014, para. 4.18.1, p. 65.

⁶⁹² Cleophas & Assim 2015, p. 298.

the appointment of a legal representative (if the child is not yet represented) conditional upon 'if the court is of the opinion that it would be in the best interests of the child to have legal representation'.⁶⁹³

Curator ad litem

There are four instances when a curator ad litem can be appointed by a court: 1. when the child has no parent or guardian; 2. where the interests of the child clash with those of the parent or guardian or if a possibility of such a clash exists; 3. where the parent or guardian of the child cannot be found; 4. where the child's parent or guardian unreasonably refuses to assist the minor in legal proceedings.⁶⁹⁴ For the appointment of a curator ad litem in family law proceedings, the second instance is the most common. As clarified by the Supreme Court of Appeal, the discretion exercised by the court in determining whether to appoint a curator ad litem for the child in one of these four instances is solely guided by the best interests of the child.⁶⁹⁵

5.4.1.6. Who decides whether the child will be represented?

Legal representative

It is generally the court that 'may appoint' a legal representative, e.g. in s. 6(4) of the Divorce Act and s. 29(6)(a) SACA 2005. However, it is unclear who makes the final decision in cases concerning the Children's Act. Section 55 states that the court 'must refer the matter to the Legal Aid Board', who must then deal with the matter 'in accordance with section 3B' of the Legal Aid Act. This seemingly places the decision regarding legal representation at the Legal Aid Board, not the court.⁶⁹⁶ However, s. 3B of the Legal Aid Act of 1969 requires the Legal Aid Board to evaluate and report to the court whether the child should be provided with legal representation at State expense. Once the court has received the report, then the court makes the final decision and can order the Legal Aid Board to provide a legal representative at State expense for the child.⁶⁹⁷

The child can also apply for the appointment of a legal representative directly to the Legal Aid Board. In *Legal Aid Board v R and another* it was confirmed that this can be done on grounds of s. 28(1)(h) of the Constitution without a prior order from the courts.⁶⁹⁸ This decision also clarified that when appointing a legal representative on the application of a child, the Legal Aid Board also does not need to obtain the consent from the child's parent or guardian.⁶⁹⁹ According to the Legal Aid Guide of 2014, if the child requests a legal representative to be able to intervene in divorce, care or maintenance proceedings, then the Regional Operations Executive (the head of the regional office)

⁶⁹³ This also applies for legal representation in international child abduction cases, which according to s. 279 SACA 2005 is subject to s. 55, as well as to representation ex. s. 29(6) which is also subject to s. 55 SACA 2005.

⁶⁹⁴ Boezaart 2013b, p. 709-710; and Boezaart & de Bruin 2011, p. 422-423.

⁶⁹⁵ Legal Aid Board in re Four Children (512/10) [2011] ZASCA 39 (29 March 2011), para. 13.

⁶⁹⁶ Sloth-Nielsen 2008, p. 499.

⁶⁹⁷ Legal Aid Guide 2014, para. 4.18.5, p. 67.

⁶⁹⁸ Legal Aid Board v R and another [2009] (2) SA 262 (D).

⁶⁹⁹ Boezaart & de Bruin 2011, p. 432; and Skelton et al. 2010, p. 265.

must give prior written consent.⁷⁰⁰ The consent will be given if a legal representative is needed to protect the best interests of the child and if substantial injustice would otherwise result.⁷⁰¹

Curator ad litem

A curator ad litem is usually appointed by the court based on an *ex parte* application by either the child, the relative of a child or some other person with a reasonable interest in the child.⁷⁰² This application can be made to the High Court following the requirements of rule 57 Uniform Rules of Court or to the Magistrate's Court (s. 33 Magistrates' Court Act). The courts have a wide discretion in appointing curators ad litem and thus can also do so *ex officio*.⁷⁰³ In fact, if the court is of the opinion that it is to the benefit of the child and in his best interests to appoint a curator ad litem, then this can even be done against the child's will or without his knowledge.⁷⁰⁴

5.4.1.7. How is the child's representative financed?

Legal representative

The costs of the child's legal representative can be financed in two different ways. On the one hand, the parties to the proceedings (e.g. the parents), or either one of them, can be ordered by the court to pay the costs of the legal representative. This is the case in s. 29(6)(b) SACA 2005 as well as in s. 6(4) of the Divorce Act. In the latter, the court is only empowered to order the appointment of a legal representative for the child on the costs of the parties, not at State expense. Thus, in divorce proceedings a legal representative through a court order is only reserved for children of wealthy parents.⁷⁰⁵ However, as mentioned above children themselves can request a legal representative through the Legal Aid Board at State expense.⁷⁰⁶ In contrast, s. 29(6)(b) SACA 2005 provides for the court to order the state to pay the costs of the legal representative instead of ordering the parents, if a substantial injustice would otherwise occur.

In all other matters of the Children's Act, s. 55 provides for the opportunity of representation at State expense. Following the Constitutional right afforded to children under s. 28(1)(h) the Legal Aid Board applies the 'substantial injustice' test previously discussed. If the criteria of the 'substantial injustice' test are met the child has the right to legal aid.⁷⁰⁷ The application for legal aid can be made either by the child, an adult acting on behalf of the child, or if there is a court order for legal representation at State expense there need not be an application.⁷⁰⁸ In most cases, the Legal Aid Board

⁷⁰⁰ Legal Aid Guide 2014, para. 4.18.7, p. 70.

⁷⁰¹ Legal Aid Guide 2014, para. 4.18.7, p. 71.

⁷⁰² Boezaart 2013b, p. 713; *Legal Aid Board in re Four Children* (512/10) [2011] ZASCA 39 (29 March 2011), para. 12.

 ⁷⁰³ Boezaart & de Bruin 2011, p. 422; *Legal Aid Board in re Four Children* (512/10) [2011] ZASCA 39 (29 March 2011), para. 12.
 ⁷⁰⁴ Legal Aid Board in re Four Children (512/10) [2011] ZASCA 39 (29 March 2011), para. 12. See also, Boezaart 2013b, p.

¹⁰⁴ Legal Aid Board in re Four Children (512/10) [2011] ZASCA 39 (29 March 2011), para. 12. See also, Boezaart 2013b, p. 717.

⁷⁰⁵ Heaton 2012, p. 402.

⁷⁰⁶ Legal Aid Guide 2014, para. 4.18.7, p. 70.

⁷⁰⁷ Legal Aid Guide 2014, para. 4.18.1, p. 65.

⁷⁰⁸ Legal Aid Guide 2014, para. 4.18.2, p. 65.

still applies the 'normal' means test either to the child's means or the means of the parents if they are assisting the child.⁷⁰⁹ If the parents have sufficient means but fail, refuse or neglect to pay the legal representative, then the Legal Aid Board will still provide the child with legal aid if 'substantial injustice' would otherwise occur and can then claim these costs through proceedings against the parents.710

Curator ad litem

The curator ad litem of the child could either be paid for by the parties involved, so either through the means of the child or the parents, but it is also possible to request legal aid. If legal aid is required and if the curator ad litem is an employee of Legal Aid South Africa, then the court must refer the case to the Regional Operations Executive who will make a decision in that regard. If legal aid is required but the curator ad litem will not be an employee of Legal Aid South Africa, then the curator must complete and sign a form which must be submitted to the Legal Aid Board.⁷¹¹

5.4.2. What is the task of the child's representative in family law proceedings in South Africa?

The main difference between the legal representative and the curator ad litem in South Africa lies in their respective tasks. In principle, the legal representative is a client-directed advocate who represents the child's views while the curator ad litem is a best-interests advocate who advances the best interests of the child.⁷¹² However, there are some nuances.

Legal representative

The role of the legal representative is to be an advocate of the child's views in the proceedings. The legal representative has the same status in the proceedings as the other parties and can therefore make sure that the child's views are represented in court.⁷¹³ However, as emphasized in *Soller NO* v G the legal representative is 'not a mere mouthpiece'.⁷¹⁴ Depending on the age, maturity and level of development of the child, the nature of the proceedings, as well as the extent to which the child wishes to participate the role of the legal representative can differ.⁷¹⁵ The legal representative should clarify what his or her role will be at the outset of the proceedings by making a determination on all these factors, with the capacity and wishes of the child as the primary factor.⁷¹⁶ If the child is sufficiently mature, developed and wishes to participate directly, then the legal representative should

⁷⁰⁹ Legal Aid Guide 2014, para. 4.18.3, p. 66.

⁷¹⁰ Legal Aid Guide 2014, para. 4.18.3, p. 66.

⁷¹¹ Legal Aid Guide 2014, para. 4.18.8(b), p. 72. ⁷¹² Boezaart 2013b, p. 716.

⁷¹³ Cleophas & Assim 2015, p. 300.

⁷¹⁴ Soller NO v G [2003] (5) SA 430 (W). See also Sloth-Nielsen 2008, p. 504.

⁷¹⁵ Boezaart 2013a, p. 369; and Cleophas & Assim 2015, p. 300.

⁷¹⁶ Cleophas & Assim 2015, p. 301.

take instructions from the child and follow them.⁷¹⁷ If the child is very young and unable to give instructions, then the role of the legal representative should be more like that of the curator ad litem: a best-interests advocate.⁷¹⁸ This has also been acknowledged by the court in B v G, where a legal representative was appointed in a role 'akin to that of a curator ad litem' for a four year old boy.⁷¹⁹

Curator ad litem

The curator ad litem is an advocate who conducts legal proceedings on behalf of the child with the child's best interests at heart.⁷²⁰ Therefore, the task of the curator ad litem is to advance all the arguments in favor of the child's objective interests in the specific case. The curator ad litem's own opinion with regards to the child's situation is, in principle, irrelevant.⁷²¹

5.4.2.1. What are the function requirements for the child's representative?

Legal representative

All the sections which determine that a legal representative should be or can be provided for the child refer to a 'legal' representative or practitioner, even s. 28(1)(h) of the Constitution. Thus, the legal representative ought to be a lawyer. This was also reaffirmed in Soller NO v G, where the court found that a social worker, psychologist or counsellor cannot be appointed as a legal representative, it must be a lawyer due to the particular skills and expertise required.⁷²² More specifically the lawyer appointed should be 'an individual with knowledge of and experience with the law but also the ability to ascertain the views of a client, present them with logical eloquence and argue the standpoint of the client in the face of doubt or opposition from an opposing party or a court'.⁷²³

Curator ad litem

Generally, a curator ad litem is an advocate.⁷²⁴ In proceedings at the High Court, the court usually directs the Bar Council to appoint a suitable advocate of the High Court with no interests in the matter as a curator ad litem.⁷²⁵ If this is not possible or feasible, an attorney can also be appointed, see rule 57(5) of the Uniform Rules of Court. A Family Advocate (to be discussed below) cannot be appointed as a curator ad litem, because the Family Advocate has a neutral and very distinct role compared to a curator ad litem.726

⁷²⁵ Boezaart 2013b, p. 713-714.

⁷¹⁷ Boezaart 2013a, p. 369; and Cleophas & Assim 2015, p. 300.

⁷¹⁸ Cleophas & Assim 2015, p. 301.

⁷¹⁹ B v G [2012] 2 SA 329 (SGJ HC). See also Boezaart 2013a, p. 370.

⁷²⁰ Legal Aid Board in re Four Children (512/10) [2011] ZASCA 39 (29 March 2011), para. 12. See also, Boezaart & de Bruin 2011, p. 423; and Cleophas & Assim 2015, p. 301.

 ⁷²¹ Boezaart 2013b, p. 716.
 ⁷²² Soller NO v G [2003] (5) SA 430 (W). See also Sloth-Nielsen 2008, p. 503.

⁷²³ Soller NO v G [2003] (5) SA 430 (W). See also Sloth-Nielsen 2008, p. 503.

⁷²⁴ Legal Aid Board in re Four Children (512/10) [2011] ZASCA 39 (29 March 2011), para. 12.

⁷²⁶ Boezaart 2013b, p. 714-715.

5.4.2.2. How should the child's representative complete their task?

Legal representative

The child's legal representative should first determine which role he or she should take. If the child is sufficiently mature and developed and wants to direct the legal representative, then the legal representative should establish the views of the child and convey them to court.⁷²⁷ In doing so, the legal representative should apply his or her legal knowledge and expertise to translate the child's views to the court when appearing on behalf of the child.⁷²⁸ The legal representative may also apply for the child to be joined as a party in the proceedings.⁷²⁹

Curator ad litem

According to the long-standing common law, the curator ad litem must represent the minor in the pending proceedings and should 'watch and protect his interest in the case as a good and prudent father would have done'.⁷³⁰ In more modern terms, the curator ad litem must investigate the child's circumstances by informing and interviewing the child and by making any other necessary enquiries.⁷³¹ On the basis thereof, the curator ad litem reports recommendations to the court.⁷³² Furthermore, if necessary the curator ad litem can also apply for the appointment of a legal representative for the child.⁷³³

5.4.2.3. Is the child representative the only option to be heard, are other options available and can they be complementary?

The representation of the child by a **curator ad litem** and a **legal representative** can be complementary. This derives in part from the fact that the curator ad litem can apply for the appointment of a legal representative for the child (see above).

The other option for the child to be heard in family law proceedings in South Africa is the office of the Family Advocate. A Family Advocate is a neutral advocate who makes enquiries to collect information and report to the court about the child's welfare in divorce cases.⁷³⁴ The enquiry is done together with qualified family counsellors or social workers following a divorce application at

⁷²⁷ Boezaart 2013a, p. 369.

⁷²⁸ Sloth-Nielsen 2008, p. 504.

⁷²⁹ Boezaart 2013a, p. 369.

⁷³⁰ Legal Aid Board in re Four Children (512/10) [2011] ZASCA 39 (29 March 2011), para. 12.

⁷³¹ Rule 57(5) Uniform Rules of Court and *Centre for Child Law and Another v Minister of Home Affairs and Others* [2005]
(6) SA 50 (T) (13 September 2004), para. 6. See also, Boezaart 2013b, p. 716; Boezaart & de Bruin 2011, p. 436; and Cleophas & Assim 2015, p. 301.

⁷³² Rule 57(5) Uniform Rules of Court and *Centre for Child Law and Another v Minister of Home Affairs and Others* [2005]
(6) SA 50 (T) (13 September 2004), para. 6. See also, Boezaart 2013b, p. 716; and Boezaart & de Bruin 2011, p. 436.
⁷³³ Boezaart & de Bruin 2011, p. 436; and Cleophas & Assim 2015, p. 301. See for an example: *Centre for Child Law and*

 ⁷³³ Boezaart & de Bruin 2011, p. 436; and Cleophas & Assim 2015, p. 301. See for an example: *Centre for Child Law and Another v Minister of Home Affairs and Others* [2005] (6) SA 50 (T) (13 September 2004).
 ⁷³⁴ S. 4 of the Mediation in Certain Divorce Matters Act of 1987. See also, Boezaart 2013b, p. 715; and Cleophas & Assim

⁷³⁴ S. 4 of the Mediation in Certain Divorce Matters Act of 1987. See also, Boezaart 2013b, p. 715; and Cleophas & Assim 2015, p. 302.

the request of the parties or the court.⁷³⁵ The Family Advocate will generally interview the parents and can talk to the child. However, the Family Advocate is not obliged to hear the child or include his or her views in the report.⁷³⁶ Thus, it is not specifically an opportunity for the child to be heard. The Family Advocate can function simultaneously with either a curator ad litem and/or a legal representative of the child, because all three have very different roles.⁷³⁷ While the latter two, as advocates of the child, do not remain neutral, that is the function of the Family Advocate.⁷³⁸

 ⁷³⁵ Cleophas & Assim 2015, p. 302.
 ⁷³⁶ Cleophas & Assim 2015, p. 303.
 ⁷³⁷ Boezaart 2013b, p. 715; Sloth-Nielsen 2008, p. 503; and Skelton et al. 2010, p. 265.
 ⁷³⁸ Soller NO v G [2003] (5) SA 430 (W) and Legal Aid Board in re Four Children (512/10) [2011] ZASCA 39 (29 March 2011), para. 21. See also, Boezaart 2013b, p. 715.